# HEAT AND MASS TRANSFER ANALYSIS IN A SPOUTED BED DRYER COUPLED WITH A CYCLONE FOR PEPPER DRYING

Gayani Kaushalya Jayatunga

(118075F)

Degree of Doctor of Philosophy

Department of Chemical & Process Engineering

University of Moratuwa

Sri Lanka

May 2020

# HEAT AND MASS TRANSFER ANALYSIS IN A SPOUTED BED DRYER COUPLED WITH A CYCLONE FOR PEPPER DRYING

Gayani Kaushalya Jayatunga

# (118075F)

Thesis submitted in partial fulfilment of the requirements for the degree Doctor of Philosophy in Chemical and Process Engineering

Department of Chemical & Process Engineering

University of Moratuwa

Sri Lanka

May 2020

# **DECLARATION OF THE CANDIDATE AND SUPERVISOR**

I declare that this is my own work and this dissertation does not incorporate without acknowledgement any material previously submitted for a Degree or Diploma in any other University or institute of higher learning and to the best of my knowledge and belief it does not contain any material previously published or written by another person except where the acknowledgement is made in the text.

Also, I hereby grant to University of Moratuwa the non-exclusive right to reproduce and distribute my thesis/dissertation, in whole or in part in print, electronic or other medium. I retain the right to use this content in whole or part in future works (such as articles or books).

Signature: .....

Date:....

The above candidate has carried out research for the PhD thesis/ Dissertation under my supervision.

Signature of the supervisor: ..... Date :....

Name of the supervisor: Prof(Mrs.) B.M.W.P.K Amarasinghe

#### Abstract

This thesis outlines the analysis of heat and mass transfer process of drying of black pepper in a spouted bed coupled with a cyclone separator. Black pepper is a widely used spice and is well-known for its pungency and aroma. Experiments conducted with wetted black pepper reveals that conventional spouted bed has the higher drying rate, higher moisture diffusivity, lower drying time and lower activation energy compared to those of spouted beds equipped with porous draft tube and non-porous draft tube. Therefore, the conventional spouted bed was used to conduct the experiments to achieve the objectives of the study since it was identified as the optimum configuration of the spouted bed.

Spouting behaviour of raw black pepper was studied in the conventional spouted bed. The minimum spouting velocities of particular bed heights at ambient temperature were determined. Minimum spouting velocity of raw black pepper in the conventional spouted bed versus static bed height was correlated to a power form of function and the exponent of static bed height was 0.67. The effect of operational conditions; drying air temperature, air velocity and static bed height on drying kinetics of raw black pepper was analysed. The higher values of moisture removal rates were achieved when the dryer was operated with moderately high air temperatures, high air flow rates and shallow beds. Drying kinetic data of raw black pepper in the conventional spouted bed shows only falling rate periods. Effective moisture diffusivity values increased with increasing drying air temperature of 75 °C. Activation energy for drying of raw black pepper in conventional spouted bed was 38.59 kJ/kmol. Specific energy consumption was calculated and the specific energy consumption values increased with increasing air velocity and the static bed height.

Drying kinetic data obtained from experiments were fitted to five thin layer drying models. Results show that the Logarithmic model gives the best fit. In addition, four models were developed for black pepper drying in the conventional spouted bed by correlating drying constants and coefficients of Logarithmic model to stagnant bed height at specified temperatures; 45 °C, 55 °C, 65 °C, and 75 °C. Developed models can be used to estimate the drying time of black pepper in the conventional spouted bed dryer for given moisture reduction in the ranges of 0.14-0.22 m stagnant bed heights and in 2.37 m/s air velocity at specified temperatures.

Essential oil was extracted using hydro distillation from black pepper dried at five different drying conditions. The analysis of components by Gas Chromatography Mass Spectrometry technique shows that the black pepper essential oil comprised mainly monoterpenes and sesquiterpenes. Analysis of variance was conducted and the results show that variation of sesquiterpenes concentration in black pepper essential oil was significant while variations of monoterpenes, oxygenated terpenes and caryophyllene concentration were non-significant in black pepper essential oil with drying air temperature. In addition, the variation of essential oil yield is statistically significant with drying air temperatures. 65 °C drying air temperature provides consistent quality essential oil with high percentage of caryophyllene and higher oil yield.

Heat transfer coefficients for black pepper drying in the conventional spouted dryer were estimated for different drying conditions of unsteady state drying of black pepper dried from initial moisture content to final moisture content of 15% dry basis. Heat transfer coefficient varied from 35-68 W/m<sup>2</sup>K for the conditions under the investigation. Dimensional analysis

was carried out and important dimensionless numbers were identified. A correlation was developed for heat transfer process as a function of dimensionless groups namely Reynolds number, Nusselt number, Gukhman number and static bed height to particle diameter with 0.791 of coefficient of determination. Heat transfer coefficients predicted from the developed correlation show a good agreement with the experimentally determined heat transfer coefficients.

Keywords: Black pepper, spouted bed, drying kinetics, heat and mass transfer, essential oil

## ACKNOWLEDGEMENT

It is my pleasure to extend sincere appreciation to all who rendered support in numerous ways to make my research a success.

First and foremost, I wish to express my sincere gratitude and respect to my supervisor, Prof. (Mrs) B.M.W.P.K Amarasinghe, Department of Chemical & Process Engineering, University of Moratuwa, for the excellent guidance throughout my postgraduate study period in many ways. Her dedication, patience, knowledge and experience inspired me to become an independent researcher and helped me realise the power of critical thinking.

I am also grateful to Prof. A.D.U.S Amarasinghe for all the advice and encouragement that he gave me throughout the time as one of progress review committee member. Then I must offer my sincere thanks to Prof. P.G Rathnasiri, Dr.(Ms). M. Y Gunesekara and Dr. Marliya Ismail for the valuable comments given, as my progress review committee members.

Prof. S. Walpalage, Prof. P.G Rathnasiri and Dr. S Gunawardhena who were the Head of the Chemical & Process Engineering Department throughout the period and all the lecturers of the Department of Chemical & Process Engineering, University of Moratuwa, who helped me in various ways to complete this research successfully, must be specially mentioned.

I am also grateful to the HETC project of Ministry of higher education for providing me the research funding and Post Graduate Studies Division, University of Moratuwa for approving my research project. I am grateful for the broad range of service provided by library of University of Moratuwa. I would like to extend my sincere gratitude to Institute of Technology University of Moratuwa for granting study leave and vacation leave to complete the research work and nominating me for HETC/HRD scholarship. Also my sincere tanks due to Ms. Nilanthi Fonseka, former Head; Division of Polymer, Textile and Chemical Engineering Technology Institute of Technology University of Moratuwa.

I am also very much grateful to Ms. Dinusha Martino; analytical chemist, Mr. Chinathka Narangoda; system analyst and Ms. K L Chandrasiri for the valuable help and support given for my research. Mrs. H.B.R Sajeewani, Mrs.I.K Athukorala, Mr. J.Wijesinghe, Mr. D. Senarath Epa, Mrs. W.K.I. Gayani , Mrs. W.S.M De Silva and all the staff members of Particulate Engineering Laboratory, Energy Engineering, Industrial Chemistry, Polymer Engineering, Food & Microbiology laboratories of Department of Chemical & Process Engineering, University of Moratuwa are reminded with heartfelt of thanks for their support given me in various occasions.

I would like to extend my thank to my sister and Dr. Duleeka Gunarathna and Anushka Perera for providing valuable literature during my research period. Furthermore, special thanks to Mr. Upali Jayasekara Managing Director of Mega Heaters (Pvt) Ltd and his staff members for carefully and patiently constructing the experiment set up.

I would be thankful to all my colleagues Ms. Thamali Wijewickrama, Ms. Bagya Herath, Ms. Sachini Thilakerathna, Ms. Sureshini, Ms.Wathsala Wickramaarchchi, Mr. Janith Bandara, Mr. Chamila Wickramasinghe, Mr. Kaun Senarathna, Mr. Nuditha Dilnayana, Ms. Imesha Samarathunga, Ms. Iroshini Kumarage and Ms. Daham Shyamalee who were with me and gave the best support by making the research period pleasant and enjoyable.

Finally to my husband, my parents in low, my parents and my sisters who have been my side through this journey. My PhD dream became a reality through their commitment and sacrifice. Also, without my house-maid Rita whom with us for four years, this dissertation would never been possible with my little baby Nethuli and my son Senuja.

# TABLE OF CONTENT

DECLARATION OF THE CANDIDATE AND SUPERVISOR	i
LIST OF FIGURES	X
LIST OF TABLES	xvi
LIST OF FIGURES IN APPENDICES	xviii
LIST OF TABLES IN APPENDICES	xix
LIST OF ABBREVIATIONS	XX
1. INTRODUCTION	1
1.1 Background to the research problem	1
1.2 iResearch objectives	2
1.3 Significant of the research	2
1.4 Structure of the dissertation	3
Abbreviations	3
2. LITERATURE REVIEW	4
2.1 Principles & theory of drying	4
2.1.1 Moisture content	4
2.1.2 Drying	4
2.1.3 Moisture diffusivity	7
2.1.4 Specific energy consumption (	9
2.1.5 Classification and selection of dryers' farming crops	10
2.2 Drying of agricultural crops	11
2.2.1 Black pepper drying	12
2.3 Spouted beds	13
2.3.1 Introduction to spouted beds	13
2.3.2." iMinimum spouting velocity""	17
2.3.3. Maximum spoutable bed height	20
2.3.4. Pressure drop	20
2.4 Dimensionless models for analysis of heat transfer in spouted bed dryers	22
2.5 Drying models	25

2.6 Scale up	30
2.7 Research on black pepper essential oil	30
2.8 Quality of black pepper	31
Chapter specific nomenclature	31
3. METHODOLOGY	35
3.1 Experimental methods	35
3.1.1 Handling of raw black pepper for experiments	36
3.1.2 Preparation of wetted black pepper	37
3.1.3 Determination of moisture content of black pepper particles	38
3.1.4 Experiment set up and equipment	38
3.1.5 Preliminary experiments	45
3.1.6 Major experiments with raw black pepper	47
3.1.7 Summary of experimental conditions5	54
3.2 Model for heat transfer	55
3.2.1 Determination of heat transfer coefficient	55
3.2.2 Dimensional analysis of heat transfer process	59
3.3 Drying model development	50
3.3.1 Mathematical modeling	50
Chapter specific nomenclature	52
4. RESULTS AND DISCUSSION ON PRELIMINARY EXPERIMENTS	56
4.1 Minimum spouting velocity	56
4.2 Effect of draft tube configurations on drying	57
4.2.1 Effect of diameter of draft tube	58
4.2.2 Effect of entrainment height of draft tube	71
4.2.3 Effect of height of draft tube	73
4.3 Effect of air velocity on drying kinetics for spouted bed configurations7	76
4.4 Effect of spouted bed configurations on drying kinetics	77
4.5 Diffusivity of wetted black pepper for three spouted bed configurations	83
4.6 Decision on selecting the optimum configuration for experiment with raw black pepper	88
4.7 Mathematical modeling of drying kinetics of wetted black pepper in three spouted bed configurations	89

Chapter specific nomenclature96
5. HYDRODYNAMICS OF SPOUTING, DRYING KINETICS AND MATHEMATICAL MODELING OF DRYING OF RAW BLACK PERPER IN
CONVENTIONAL SPOUTED BED
5.1 Minimum spouting velocity97
5.2 Drying kinetics
5.2.1 Effect of drying air temperature
5.2.2 Effect of bed height
5.2.3 Effect of air flow rate111
5.2.4 Summary on effect of operational variables
5.3 Specific energy consumption
5.4 Effective moisture diffusivity of raw pepper drying in conventional spouted
bed117
5.6 Mathematical modeling of drying kinetics
Chapter specific nomenclature
6. ANALYSIS OF QUALITY OF BLACK PEPPER
6.1 Analysis of black pepper essential oil
6.2 Analysis of variance
6.3 Effect of drying condition on Caryophyllene concentration144
6.4 Effect of drying condition on essential oil yield145
Chapter specific nomenclature147
7. NON DIMENSIONAL MODEL
7.1 Dimensional analysis for heat transfer process
7.2 Experimental results
7.3 Correlation for heat transfer process
7.4 Effect of operational conditions on heat transfer
Chapter specific nomenclature
8. CONCLUSION & RECOMMENDATIONS FOR FUTURE WORK167
8.1 Conclusion
8.2 Recommendations for future work170
REFERENCES171
APPENDIX A

APPENDIX C	197
APPENDIX D	
APPENDIX E	
APPENDIX F	
APPENDIX G	
APPENDIX H	
List of Publications	

# LIST OF FIGURES

Figure 2.1: Moisture-content vs. time of a material being dried in a dryer (Typical
drying curve)
Figure 2.2: Typical drying rate curve including constant rate and falling rate drying
conditions [14]7
Figure 2.3 : Schematic diagram of a conventional spouted-bed [37]14
Figure 2.4-: Spouted bed Configurations: (a) conventional, (b) conical (c) two
dimensional with draft plate and (d) triangular spouted bed [27]16
Figure 2.5: a) open sided draft-tube b) non porous draft-tube [46]17
Figure 2.6: "Photographic sequence of the evolution of a spouting process of PET
chips performed in a squared-based half sectional 0.2 m side unit" [56]21
Figure 2.7: "Pressure drop versus. flow rate hysteresis between an increasing and the
decreasing flow in a spouted bed of PET chips" [56]21
Figure 3.1: Raw black pepper preparation for experiments (a) black pepper spikes
(b) raw black pepper seeds
Figure 3.2: Schematic diagram of the spouted bed drying experiment Set up;39
Figure 3.3: Image of spouted bed drying experiment set up40
Figure 3.4: Arrangement of temperature sensors inside the spouted bed contactor41
Figure 3. 5: a) non porous draft tube (b) porous draft tube [101] and (c) Image of
both type od draft tubes43
Figure 3. 6: A porous draft tube is ready to fit into the spouted bed contactor43
Figure 3.7: Oil extraction apparatus
Figure 3.8: Clevenger type oil trap
Figure 3.9: (a) Clevenger type oil trap in experiment set up (b) interphase of oil and
water in oil trap
Figure 3. 10: Schematic diagram for unsteady state spouted bed drying process56
Figure 4.1: Effect of diameter of non-porous draft tube on drying of wetted black
pepper under minimum spouting conditions in SBNPDT;69
Figure 4.2: Rate of change of moisture content vs. moisture content;69
Figure 4.3: Effect of diameter of porous draft tube on drying of wetted black pepper
under minimum spouting conditions in SBPDT; $T_d$ = 55 $^\circ C,$ H =0.16 m, $D_i$ =0.05 m, ,
$L_{H} = 0.03 \text{ m}, H_{T} = 0.16 \text{ m} \dots 70$
Figure 4.4: Rate of change of moisture content vs. moisture content;70
Figure 4.5: Effect of entrainment height of non- porous draft tube on drying of
wetted black pepper under minimum spouting conditions in SBNPDT; $T_d = 75$ °C,
$H = 0.16 \text{ m}, D_i = 0.035 \text{ m}, H_T = 0.16 \text{ m}, D_T = 0.035 \text{ m} \dots 71$
Figure 4.6: Rate of change of moisture content vs. moisture content72

Figure 4.7: Effect of entrainment height of porous draft tube on drying of black
pepper under minimum spouting conditions in SBPDT; $T_d = 75$ °C, H =0.16 m, $D_i$
=0.05m, $H_T$ = 0.16 m, $D_T$ = 0.035m72
Figure 4.8: Rate of change of moisture content vs. moisture content73
Figure 4.9: Effect of tube height of non- porous draft tube o drying of wetted black
pepper under minimum spouting conditions in SBNPDT;74
Figure 4.10: Rate of change of moisture content vs. moisture content74
Figure 4.11: Effect of tube height of porous draft tube on drying of black pepper
under minimum spouting conditions in; $T_d\!\!=75^\circ\!C,H\!=\!\!0.16$ m, $D_i\!=\!\!0.05$ m, $L_H\!\!=0.06$
m, $D_T = 0.035$ m75
Figure 4.12: Rate of change of moisture content vs moisture content75
Figure 4.13: Effect of air velocity on drying kinetics of black pepper when dried in
CSB, SBPDT and SBNPDT. Experiment conditions: $T_{d}$ = 75 $^{\circ}C,$ $D_{o}$ =0.035 m, H =
$0.16 \text{ m}, D_T = 0.035 \text{ m}, L_H = 0.06 \text{ m}, L_T = 0.26 \text{ m}$
Figure 4.14: Effect of air velocity on drying kinetics of black pepper when dried in
CSB, SBPDT and SBNPDT. Experiment conditions: $T_d$ = 75 $^\circ C,  D_o$ =0.035 m, H =
$0.16 \text{ m}, D_{T} = 0.035 \text{ m}, L_{H} = 0.06 \text{ m}, L_{T} = 0.26 \text{ m}80$
Figure 4.15: Effect of spouted bed configuration on drying kinetics of black pepper
when dried in CSB, SBPDT and SBNPDT at minimum spouting conditions.
Experiment conditions: $T_d$ = 75 °C, $D_o$ =0.035 m, H = 0.16 m, $D_T{=}$ 0.035 m, $L_H$ =
$0.06 \text{ m}, L_T = 0.26 \text{ m}$
Figure 4. 16: Rate change of moisture content vs. moisture; Effect of spouted bed
configuration on drying kinetics of black pepper when dried in CSB,SBPDT and
SBNPDT at minimum spouting conditions; Experiment conditions: $T_d = 75$ °C, $D_o$
=0.035 m, H = 0.16 m, $D_T$ = 0.035 m, $L_H$ = 0.06 m, $L_T$ = 0.26 m
Figure 4.17: Ln (Moisture ratio) vs. drying time of wetted pepper dried in
conventional spouted bed dryer Experiment conditions: $H = 0.16m$ , $D_i = 0.035 m \dots 83$
Figure 4.18: Ln (Moisture ratio) vs. drying time of raw pepper dried in porous draft
tube fitted spouted bed dryer; Experiment conditions: H =0.16 m, $D_i$ =0.035 m, $L_H$ =
$0.06 \text{ m}, D_{T} = 0.035 \text{m}, H_{T} = 0.02 \text{ m}$
Figure 4.19: Ln (Moisture ratio) vs. drying time of wetted pepper dried in non-
porous draft tube fitted spouted bed dryer; Experiment condition: H =0.16 m, $D_i$
=0.035 m, $L_{H}$ = 0.06 m, $D_{T}$ = 0.035 m, $H_{T}$ = 0.02 m
Figure 4.20: Ln (D <sub>eff</sub> ) vs. $1/T_d$ of wetted pepper dried in three spouted bed
configurations; Experiment conditions: H =0.16 m, $D_i$ =0.035 m, $L_H$ = 0.06 m, $D_T$ =
$0.035 \text{ m}, \text{H}_{\text{T}} = 0.02 \text{ m}86$
Figure 4. 21: A comparison between the observed moisture ratio and predicted
moisture ratio of black pepper dried at different drying air temperatures in
conventional spouted bed using Logarithmic model

Figure 4. 22: A comparison between the observed moisture ratio and predicted
moisture ratio of black pepper dried at different drying air temperatures in
conventional spouted bed using Henderson and Pabis model
Figure 4. 23: A comparison between the observed moisture ratio and predicted
moisture ratio of black pepper dried at different drying air temperatures in
conventional spouted bed using Two Compartment model94
Figure 4. 24: A comparison between the observed moisture ratio and predicted
moisture ratio of black pepper dried at different drying air temperatures in
conventional spouted bed using Newton model94
Figure 4. 25 : A comparison between the observed moisture ratio and predicted
moisture ratio of black pepper dried at different drying air temperatures in
conventional spouted bed using Page model95
Figure 5.1: Pressure drop vs. superficial air velocity for 0.07 m static bed of raw
black pepper at ambient temperature
Figure 5.2: Pressure drop vs. superficial air velocity for 0.14 m static bed of raw
black pepper at ambient temperature99
Figure 5.3:Pressure drop vs. superficial air velocity for 0.18 m static bed of raw
black pepper at ambient temperature99
Figure 5.4: Pressure drop vs. superficial air velocity for 0.22 m static bed of raw
black pepper at ambient temperature100
Figure 5. 5: Pressure drop vs. superficial air velocity for different static bed heights
of raw black pepper at ambient temperature: (a) 0.07 m, (b) 0.14 m, (c) 0.18 m, (d) 0.
22 m101
Figure 5.6: Minimum spouting velocity vs. static bed height102
Figure 5. 7: Moisture content- dry basis vs. time of black pepper when dried in
conventional spouted bed at different drying air temperatures. Experiment
conditions: $D_i{=}0.05$ m, H =0.14 m, and air flow of 2.37 m/s104
Figure 5.8: Rate of change of moisture content vs moisture content- dry basis of
black pepper when dried in conventional spouted bed at different drying air
temperatures; experiment conditions: $D_i{=}0.05$ m, H =0.14 m and air flow of 2.37 m/s
Figure 5.9: Volume of air required for raw black pepper when dried in conventional
spouted bed at different drying air temperatures. Experiment conditions: $D_i\!\!=\!\!0.05$ m,
H =0.14 m, and air flow of 2.37 m/s107
Figure 5.10: Drying time for black pepper when dried in conventional spouted bed at
different drying air temperatures. Experiment conditions: $D_i\!\!=\!\!0.05$ m, H =\!\!0.14 m, and
air flow of 2.37 m/s107
Figure 5.11: Moisture content (% dry basis) versus time of black pepper when dried
in conventional spouted bed at different static bed heights: Experiment conditions: D <sub>i</sub>

Figure 5.12: Rate of change of moisture content, dx/dt versus moisture content (% dry basis) of black pepper when dried in conventional spouted bed at different static bed heights; experiment conditions:  $D_i = 0.05$  m,  $T_d = 65$  °C and air flow of 2.37 m/s

Figure 5.13 :Volume of air required for raw black pepper when dried in conventional spouted bed at different static bed heights; experiment conditions:  $D_i = 0.05$  m,  $T_d = 65$ °C and air flow of 2.37 m/s ......110 Figure 5.14: Drying time and Volume of air required for raw black pepper when dried in conventional spouted bed at different static bed heights; experiment Figure 5.15: Moisture content (% dry basis) vs. time of black pepper when dried in conventional spouted bed dryer at different superficial air velocities; Experiment Figure 5.16: Rate of change of moisture content vs. moisture content- dry basis of black pepper when dried in conventional spouted bed at different superificial air velocities; Experiment conditions:  $D_i=0.05$  m,  $T_d=65$  °C and H=0.14 m ......112 Figure 5.17: Drying time and volume of air required for raw black pepper when dried in conventional spouted bed dryer at different superficial air velocities; Experiment conditions:  $D_i=0.05$  m,  $T_d=65$  °C and H=0.14 m......113 Figure 5.18 : Volume of air required for raw black pepper when dried in conventional spouted bed dryer at different superficial air velocities; Experiment Figure 5.19: Specific energy consumption vs. air velocity for black pepper dried in conventional spouted bed at different superficial air velocities; Experiment Figure 5.20: Ln (MR) vs time of raw black pepper dried in conventional spouted bed Figure 5.21: Ln (effective diffusivity) vs 1/temperature for raw black pepper dried in Figure 5.22: A comparison between the experimental moisture ratio and predicted moisture ratio (using Logarithmic model) of black pepper dried in various Figure 5.23: A comparison between the experimental moisture ratio and predicted moisture ratio (using Logarithmic model) of black pepper dried in various Figure 5.24: A comparison between the experimental moisture ratio and predicted moisture ratio (using Logarithmic model) of black pepper dried in various Figure 5.25: A comparison between the experimental moisture ratio and predicted moisture ratio of black pepper dried in 45 °C......132

Figure 5.26: A comparison between the experimental moisture ratio and predicted Figure 5.27: A comparison between the experimental moisture ratio and predicted Figure 5.28: A comparison between the experimental moisture ratio and predicted moisture ratio of black pepper dried in 75 °C......134 Figure 5.29: A comparison between the experimental moisture ratio and predicted moisture ratio of black pepper dried in various temperatures at static 0.14 m bed Figure 5.30: A comparison between the experimental moisture ratio and predicted moisture ratio of black pepper dried in various temperatures at static 0.18 m bed Figure 5.31: A comparison between the experimental moisture ratio and predicted moisture ratio of black pepper dried in various temperatures at static 0.20 m bed Figure 5.32: A comparison between the experimental moisture ratio and predicted moisture ratio of black pepper dried in various temperatures at static 0.22 m bed Figure 6.1: Percentage of mean sesquiterpenes, monoterpenes and oxygenated compounds in Black pepper essential oil at different drying conditions ......141 Figure 6.2: Percentage of Caryophyllene and percentage of caryophyllen oxide vs. Figure 6.3: Mean oil yield vs. drying conditions ......146 Figure 6.4:-Black pepper essential oil yield vs. drying conditions......146 Figure 7.1: Comparison of experimented heat transfer coefficient with predicted heat Figure 7.2: Comparison of experimented Nu with response surfaces of predicted Nu Figure 7.3: Comparison of experimental Nu with predicted Nu using existing Figure 7.4: Effect of drying temperature on heat transfer coefficient for black pepper dried in conventional spouted bed at different drying temperatures; Experiment Figure 7.5: Nusselt number as a function of drying temperature for black pepper dried in conventional spouted bed at different drying temperatures; Experiment Figure 7. 6: Response surfaces for predicted Nu number as a function of Re number Figure 7. 7: Response surfaces for predicted heat transfer coefficient as a function of 

Figure 7.8: Effect of static bed height on heat transfer coefficient for black pepper dried in conventional spouted bed of different bed heights; Experiment conditions:  $D_i=0.05 \text{ m}, T_{gi}=65 \text{ °C} \text{ and } U=2.37 \text{ m/s}....162$ Figure 7.9: Nusselt number as a function of static bed height for black pepper dried in conventional spouted bed of different bed heights; Experiment conditions: Figure 7.10: Response surfaces for predicted Nu number as a function of Gu number Figure 7.11- Effect of air velocity and Re number on heat transfer coefficient for black pepper dried in conventional spouted bed at different superficial air velocities; Experiment conditions:  $D_i=0.05$  m,  $T_{gi}=65$  °C and H=0.14 m......164 Figure 7.12-: Nusselt number as a function of Reynolds number and air velocity for black pepper dried in conventional spouted bed at different superficial air velocities; Figure 7.13: Response surfaces for predicted Nu number as a function of Gu number 

# LIST OF TABLES

Table 2.1:Correlations for minimum spouting velocity 19
Table 2.2 : Correlation associated with heat transfer in spouted beds    24
Table 2.3: Thin layer drying models 28
Table 3.1: Geometric factors of the spouted bed41
Table 3.2: Dimensions of non-porous draft tubes 44
Table 3.3: Dimensions of porous daft tubes 44
Table 3.4: Operating conditions of preliminary experiments conducted to analyze the
effect of spouted bed configuration at minimum spouting air velocity
Table 3.5: Operating conditions of preliminary experiments conducted to analyze the
effect of spouted bed configuration at superficial velocity of 0.89 m/s46
Table 3.6: Operating conditions of preliminary experiments conducted to analyze the
effective moisture diffusivity of black pepper in three spouted bed configuration47
Table 3.7 : The conditions used in drying experiments with conventional spouted bed
Table 3.8: Important variables for heat transfer of black pepper in the spouted bed
dryer
Table 3.9: Selected thin layer drying models used for analysis
Table 4.1: Minimum spouting velocity
Table 4.2: Drying time of black pepper at different spouted bed configurations at
different air velocities
Table 4.3: Drying time of spouted bed configurations at different drying
temperatures
Table 4.4: Effective moisture diffusivity of wetted black pepper dried in
conventional spouted bed at different temperatures85
Table 4.5: Effective moisture diffusivity of wetted black pepper dried in porous draft
tube fitted spouted bed at different temperatures85
Table 4. 6: Effective moisture diffusivity of wetted black pepper dried in non-porous
draft tube fitted spouted bed at different temperatures
Table 4.7: Pre -exponential factor and activation energy of wetted black pepper dried
in different spouted bed configurations
Table 4.8: Results of statistical analysis for three spouted bed configurations at
55_°C drying air temperature90
Table 4.9: Results of statistical analysis for three spouted bed configurations at
65_°C drying air temperature91
Table 4.10: Results of statistical analysis for three spouted bed configurations at
75_°C drying air temperature
Table 5.1: Experimental and predicted minimum spouting velocity for selected
stagnant bed heights of raw black pepper103
Table 5. 2- Experimental and predicted maximum pressure drop for selected stagnant
bed heights of raw black pepper103

Table 5.3: Drying time and volume of air required for raw black pepper when dried
in conventional spouted bed at different drying air temperatures. Experiment
conditions: $D_i=0.05$ m, H =0.14 m (bed weight = 1 kg), and air flow of 2.37 m/s106
Table 5.4: Volume of air required for raw black pepper when dried in conventional
spouted bed at different static bed heights; experiment conditions: $D_i = 0.05$ m, $T_d = 65$
°C and air flow of 2.37 m/s
Table 5.5 : Drying time and volume of air required for raw black pepper when dried
in conventional spouted bed dryer at different superficial air velocities; Experiment
conditions: $D_i=0.05$ m, $T_d=65$ °C and H=0.14 m
Table 5.6: Specific energy consumption for drying of black pepper at different
drying temperaures (air velocity 2.37 m/s)116
Table 5.7 : Effective moisture diffusivity of raw black pepper dried in conventional
spouted bed dryer at different temperatures119
Table 5.8: Newton model developed for black pepper drying in conventional spouted
bed for various stagnant bed heights and drying air temperatures at 2.37 m/s of air
velocity121
Table 5.9: Page's model developed for black pepper drying in conventional spouted
bed for various stagnant bed heights and drying air temperatures at 2.37 m/s of air
velocity122
Table 5.10: Henderson and Pabis model developed for black pepper drying in
conventional spouted bed for various stagnant bed heights and drying air
temperatures at 2.37 m/s of air velocity
Table 5. 11: Two compartment model developed for black pepper drying in
conventional spouted bed for various stagnant bed heights and drying air
temperatures at 2.37 m/s of air velocity
Table 5. 12: Logarithmic model developed for black pepper drying in conventional
spouted bed for various stagnant bed heights and drying air temperatures at 2.37 m/s
of air velocity
Table 5. 13: The results of the multiple regressions on the coefficients and constants
of the logarithmic model for the effects of the stagnant bed height130
Table 5. 14: Modified Logarithmic model for black pepper drying for 45 $^\circ\!C$ - 75 $^\circ\!C$
Table 6.1: Constituents of essential oil extracted from black pepper dried under
different drying conditions140
Table 6.2: Ten Most Abundant Components of Black Pepper Essential Oil
Table 7.1: Dimensional matrix of the fundamental dimensions related to heat transfer
process
Table 7.2: Operating conditions, heat transfer coefficient for black pepper dried in
conventional spouted bed

# LIST OF FIGURES IN APPENDICES

Appendix A Figure 1: Powder classification diagram for fluidization	by air at
ambient conditions [12]	186
Appendix B Figure 1: Precision balance; VWRECN 611-2300	195
Appendix B Figure 2: Protable digital balance; KERN PCB 350-3	195
Appendix B Figure 3: Drying Oven; A lab Tech LDO-060E	195
Appendix B Figure 4: EXTECH CFM Thermo anemometer model 40	)7113; (b)-
Thermo-Hygrometer Model GMK-920HT	196
Appendix B Figure 5: GC-MS)-7890A gas chromatograph (Agilent,	American)
equipped with a 5975C plus mass spectrometer (Agilent, American)	196
Appendix C Figure 1: Spouted bed column	197
Appendix C Figure 2: non porous draft tube	198
Appendix C Figure 3: Reverse flow cyclone separator	198
Appendix C Figure 4: Arrangement of temperature sensors	199
Appendix G Figure 1 : Model for heat transfer	
Appendix G Figure 2: Model for heat transfer	

# LIST OF TABLES IN APPENDICES

Appendix A Table 1 : Brief description of widely used dryers
Appendix A Table 2: Categorization of powders in relation of fluidization
characteristics [12]
Appendix A Table 3: Some of the key characteristics of gas spouted and gas
fluidized bed dryers
Appendix A Table 4: Characteristics of the conventional spouted bed versus spouted
bed with draft tubes
Appendix A Table 5: Correlation for Minimum Spouting Velocity189
Appendix A Table 6: Correlations available for prediction of maximum spoutable
height190
Appendix A Table 7: Correlation available for prediction of maximum pressure drop
Appendix A Table 8: Options for scaling up of conventional spouted bed192
Appendix A Table 9: Specifications of black pepper defined by some institutes 193
Appendix A Table 10: Standard quality specifications of black pepper approved by
Sri Lanka Standard Institute193
Appendix C Table 1: dimensions of cyclone separator198
Appendix E Table 1: Mean composition of essential oil derived from black pepper
dried at different conditions
Appendix E Table 2: Percentage of Constituents of essential oil extracted from black
pepper dried at different drying conditions
Appendix E Table 3: Information of five most abundant sesquiterpene in black
pepper essential oil under investigation [120]293
Appendix E Table 4: Information of six most abundant monoterpene in black pepper
essential oil under investigation [120]
Appendix E Table 5: ANOVA table for comparative study of sesquiterpenes,
monoterpenes and oxygenated terpenes with drying temperature
Appendix E Table 6: ANOVA table for Caryophyllene with drying temperature295
Appendix E Table 7: ANOVA table for oil yield vs. drying temperature296
Appendix F Table 1: Operating conditions, heat transfer coefficient, important non-
dimensional numbers of heat transfer process of black pepper drying in conventional
spouted bed
Appendix G Table 1: Summary of model for heat transfer
Appendix G Table 2: ANOVA Table for model for heat transfer
Appendix G Table 3: Coefficients for model for heat transfer

## LIST OF ABBREVIATIONS

ANOVA - Analysis of Variance

ASTA - American spice trade association

CFD - computational fluid dynamics

CSA - Canadian spice association

CSB - conventional spouted bed

ESA - Europe spice association

GC-MS - Gas Chromatography Mass Spectrometry

MR – Moisture ratio

NCPHT & VA - National Committee on Post-Harvest Technology and Value Addition

 $R^2$  - Coefficient of determination

RMSE - Root mean square error

SBNPDT - Spouted bed equipped with non-porous draft tube

SBPDT - Spouted bed equipped with porous draft tube

SEC - Specific energy consumption

SLCARP -"Sri Lanka Council for Agricultural Research Policy"

SSE - Sum of square error

VSD - variable speed drive

## Chapter 1

### **1. INTRODUCTION**

#### **1.1 Background to the research problem**

Black pepper is the berries of a tropical vine botanically known as '*Piper nigrum L*.'. Dried black pepper is one of the largest commodity in the international spice trade. Global annual demand for black pepper is 250,000-300,000 MT in year 2017. Of the total world trade in spices, one third of total volume is black pepper. Brazil, India, Indonesia, Malaysia, Sri Lanka and Vietnam are the major pepper producing countries. In Sri Lanka during year 2006 to 2016, the average annual black pepper production is approximately 17,000 MT while average export is approximately 10,000 MT [1].

Black pepper is a widely used spice in food preparation all over the world. Moreover, black pepper is also used in pharmaceutical industry and cosmetic industry due to its aroma, pungency and other medicinal features such as antioxidant, anti-inflammatory, anti-bacterial and anti-asthmatic actions [2, 3, 4]. Black pepper is commonly available as dry seeds or in its powder form in the market.

The main pungent component in black pepper is piperine;  $C_{17}H_{19}O_3N$  which is 2 trans, 4-trans piperidine amide of piperic acid. Piperine is non-volatile and can be isolated from solvent extraction (alcohol extraction) as oleoresins [5]. The essential oil of black pepper, which consists of volatile components obtained by the steam distillation or other methods, is a valuable additive in the food manufacturing industry such as sausages, canned meats, soups, beverages and liquors and also in perfumery and cosmetics industry. The quality of black pepper may be affected depending on factors such as maturity level, type of cultivars, season and post-harvest processing method [6].

Considering the post-harvest methods, drying is the most important unit operation as it affects the quality of the final product. After harvesting, pepper berries are removed from the spikes by either hand or using a mechanical thresher. These berries are then spread as a thin layer on mats and dried in the sun for five to six days depending upon the climate conditions. Delays in drying of black pepper after harvesting will cause growth of moulds on the berries and lowers the product quality. The best moisture content of black pepper to prevent the growth of moulds during storage is found to be 12-14% dry basis [7, 8].

Sun drying is the commonly used method for drying of agricultural crops in tropical countries. However, the method has several limitations such as non-uniform sun light during the day and problems caused during the rainy seasons. Therefore, the use of alternative drying techniques becomes essential. Among the techniques used for drying of agricultural products, packed beds, fluidized beds, spouted beds and cross circulation drying using tray dryers are commonly used. Many researchers promote these drying methods due to their lower capital cost and maintenance cost, higher heat and mass transfer rates and their ability to produce quality product compared to high-end drying technologies such as freeze drying, drum drying and steam drying [9].

Spouted bed dryers are particularly suitable for heat sensitive materials such as agricultural products or polymeric granules. Regular circulation of particles in three hydrodynamic regions namely spout, fountain and annulus regions permit the use of higher temperature gas without the risk of thermal damage to the particles than those of the non-agitated driers. More information on spouted beds will be available in Chapter 2.

#### **1.2 Research objectives**

The overall objective of this research was to analyse the heat and mass transfer process of black pepper drying in the spouted bed dryer. Specific objectives are as follows:

- 1. Evaluation of the spouted bed dryer for black pepper drying process
  - 1.1. Study hydrodynamics of black pepper spouting.
  - 1.2. Analysing the effects of initial bed height, air temperature, air velocity and draft tube on drying.
  - 1.3. Examine optimum drying air temperature to achieve black pepper with maximum essential oil yield with consistent quality.
  - 1.4. Analyse the effect of drying air temperature on mass transfer characteristics namely effective moisture diffusivity.
- 2. Model heat transfer characteristics in the spouted bed dryer and validate the model using the experimental data
- 3. Develop suitable drying model to describe drying behaviour (drying kinetics).

### **1.3 Significant of the research**

Black pepper has been identified as a high priority crop by National Committee on Post-Harvest Technology and Value Addition (NCPHT & VA) of Sri Lanka Council for Agricultural Research Policy (SLCARP). Furthermore, postharvest technologies for black pepper such as drying, processing, packaging and storage are also focused as research priority areas to be attended [10, 11].

Feasibility of using spouted bed dryer as a drying technique for black pepper drying is explored from this research. It is required to identify the optimum operational parameters such as air velocity, drying temperature and bed height to improve the performance of the drying process. The drying kinetics data, heat and mass transfer characteristics identified for black pepper drying in the spouted bed, thin layer modelling and the non-dimensional model developed in this research can be utilized in agriculture industry or other researchers to optimize the drying operations in conjunctions with the various types of drying models such as computational fluid dynamics (CFD) models or other studies.

#### 1.4 Structure of the dissertation

This thesis consists of nine chapters. Chapter 1 provides general overview of the study and the scope of the study. Chapter 2 provides the literature review of the study. Chapter 3 describes the methodology used to accomplish the objective of the study. The results and discussion of the study are discussed in Chapter 4 to 7. Chapter 4 provides the results of preliminary experiments conducted with wetted black pepper. Chapters 5 to 7 present the results and discussion of main experiments conducted with raw black pepper to accomplish the objectives. Chapter specific nomenclatures and abbreviations are given at the end of each chapter from Chapter 1 to 7. Chapter 8 presents the overall conclusion and the recommendations for future works. Finally, References and Appendices are presented.

#### Abbreviations

NCPHT & VA - National Committee on Post-Harvest Technology and Value Addition

SLCARP - Sri Lanka Council for Agricultural Research Policy

CFD - computational fluid dynamics

### Chapter 2

### **2. LITERATURE REVIEW**

Many researchers have investigated on drying of agricultural crops. Various techniques have been experimented. Drying kinetics, heat and mass transfer have been studied extensively. This section covers a comprehensive review on drying theory, research conducted with spouted bed and drying models. In addition, conventional methods of black pepper drying, research conducted on black pepper essential oil and quality of black pepper are reviewed.

#### 2.1 Principles & theory of drying

#### 2.1.1 Moisture content

Moisture content of a material is described as either moisture content of dry basis (X) or moisture content of wet basis. They are expressed as shown in Equation 2.1 and 2.2.

Moisture content (dry basis) = 
$$\frac{\text{kg of moisture}}{\text{kg of dry solid}}$$
 (2. 1)

Moisture content (wet basis) =  $\frac{\text{kg of moisture}}{\text{kg of wet solid}}$  (2. 2)

### 2.1.2 Drying

Drying is the removal of moisture adhering to a material by vaporization. Moisture is located on the surface and in the interior of the material such as capillaries and interstices.

When a sample of wet material is dried in a stream of air from initial moisture content to equilibrium moisture content, typical variation of moisture content of the sample with time can be shown in Figure 2.1. Equilibrium moisture content ( $X_e$ ) is the moisture content of the material in equilibrium with the air at given humidity and temperature. Measuring the slope of tangent drawn to the curve of Figure 2.1 or by

determining rate of change moisture content (dX/dt), drying rate (N) can be determined.



Figure 2.1: Moisture content vs. time of a material being dried in a dryer (Typical drying curve)

Drying rate, expressed in Equation (2.3) as mass of moisture evaporated per unit area per time depends on the way of moisture present in the material to be dried and the type of the material [12].

$$N = -\frac{W_d}{A}\frac{dX}{dt} \qquad (2.3)$$

Where

W<sub>d</sub> - mass of dried solid

A - cross section of the bed measured at right angles to the air flow

More information on drying can be obtained when the drying rate was plotted against moisture content of the material as in Figure 2.2. If the material is very wet, unbound moisture is available as a thin film of liquid on the surface of the solid. During the drying, the surface moisture is vaporized first. Even the moisture on surface is depleting, moisture concentration in depths is high. Due to high moisture diffusivity, liquid moisture in capillaries and interstices move to the surface rapidly where rate of drying remains constant at the value of  $N_c$  as shown between points Q and R in Figure 2.2. Surface is saturated with moisture or thin layer of moisture remains on the surface of the material during the constant rate-drying period. Constant rate drying processes are called as externally controlled processes since externally controlled parameters govern the drying process [13]. Externally controlled parameters are operational conditions of drying process such as air temperature, air humidity, air velocity, etc.

The induction period or initial adjustment period shown in the period P'Q and PQ of Figure 2.1 and Figure 2.2 is usually ignored in many works due its rapidity. If the material to be dried is colder than the ultimate surface temperature, initial adjustment occurs by increasing drying rate as shown in curve PQ. If the material temperature is higher than the ultimate surface temperature, drying rate decreases as in curve P'Q.

When the moisture content reaches to the critical moisture content  $(X_c)$ , material is unable to keep its thin layer of moisture on the surface and hot spots are appeared on the surface. This gives rise to the unsaturated surface drying, first part of the falling rate-drying period, between the points R and S. Eventually moisture available on surface is entirely evaporated at an average moisture content corresponding to point S.

On further drying, the moisture is expelled, because of concentration gradients exists between deeper part of the material and the surface. Rate of internal movement of moisture decreases as a result of reducing moisture concentration in the deeper parts of the material. In the later part of the falling rate drying period, from points S to T, rate of drying falls more rapidly than that of the first part of the falling rate-drying period. It shows that internal movement of moisture governs the drying rate in falling rate-drying period. Therefore drying processes having falling rate periods are called as internally controlled processes. Movement of moisture from depth to the surface is occurred due to many principles such as liquid diffusion, capillary movement, vapor diffusion, pressure, etc. [14].



Figure 2.2: Typical drying rate curve including constant rate and falling rate drying conditions [14]

#### 2.1.3 Moisture diffusivity

When a sample of wet material is dried, the drying rate can be constant or vary with time. In constant rate drying period, the surface of the solid is saturated with moisture and the rate of drying is controlled by the rate of water evaporation. In falling rate drying period, movement of moisture from interior to surface of the drying material is occurred mainly by diffusion and controls the rate of drying.

Effective diffusivity describes the potential of moisture diffusing from material inside to particle surface. Many researchers have estimated the effective moisture diffusivity ( $D_{eff}$ ) of many materials for numerous type of material, dryer types and numerous operating conditions, but a few researchers have estimated the convective mass transfer coefficient of particle air interface. In most of the cases, moisture or mass transfer coefficient has been estimated for steady state or constant rate drying operations. To the best of the authors' knowledge, experimental techniques have not been used for estimating mass transfer coefficient of drying processes in unsteady state operations. If a reliable technology exists to measure the humidity of the

particle air interface continuously with time, experimental mass transfer coefficients in unsteady state conditions can be easily determined.

Moisture diffusivity in a material depends on the type of material, porosity, shape, air temperature, initial moisture content, etc [15]. The general solution of Fick's equation can be derived for spherical particles as depicted in Equation 2.4 and it has been used by several researchers in order to obtain the effective moisture diffusivity of agricultural products [16, 17, 18].

$$MR = \frac{X_t - X_e}{X_0 - X_e} = \frac{6}{\pi^2} \sum_{i=1}^n \frac{1}{n^2} exp(-n^2 \pi^2 F_0) \quad (2.4)$$

Where

$$\label{eq:F0-Fourier number} \begin{split} F_0 \mbox{-Fourier number} \\ MR \mbox{-moisture ratio} \\ t \mbox{-time (s)} \\ X_e \mbox{-equilibrium moisture content (kg of moisture /kg dry solids)} \\ X_o \mbox{-initial moisture content (kg of moisture /kg dry solids)} \\ X_t \mbox{-moisture content at time t (kg of moisture /kg dry solids)} \end{split}$$

Then Equation (2.4) is evaluated numerically for Fourier number  $F_0$  as in Equation (2.5).

$$F_0 = \frac{D_{eff}t}{R_p^2}$$
(2.5)

Where

 $D_{eff}$  -effective moisture diffusivity (m<sup>2</sup>/s)

 $R_{p}$ - radius of the spherical particle (m)

The moisture ratio is simplified to  $X_t/X_0$  neglecting  $X_e$  term by some researchers, since  $X_e$  term is relatively small compared to  $X_t$  and  $X_0$  and the continuous fluctuation of the relative humidity of drying air leading to change of equilibrium moisture content of the particle being dried [19]. By neglecting the terms beyond the first term of summations and replacing the Fourier number with proper variables, the Equation (2.6) can be obtained.

$$\ln MR = \ln \frac{X_t}{X_0} = \ln \frac{6}{\pi^2} - \pi^2 \frac{D_{eff}}{R_p^2} t$$
(2.6)

Plotting  $ln \frac{x_t}{x_0}$  vs time at a given air temperature will result in a straight line of slope  $\pi^2 \frac{D_{eff}}{R_p^2}$  which can be used to find the effective diffusivity. In literature, it is found that  $D_{eff}$  has been expressed as a function of temperature in an Arrhenius type equation as given in Equation (2.7) [20, 17]:

$$D_{eff} = D_0 exp\left(-\frac{E_a}{RT}\right) \tag{2.7}$$

Where

D<sub>0</sub> -pre exponential factor of Arrhenius equation

E<sub>a</sub> -activation energy (kJ/mol)

R -universal gas constant (kJ/mol K)

T -absolute drying air temperature (K)

The plot of  $ln(D_{eff})$  versus l/T is used to determine the D<sub>0</sub> and E<sub>a</sub>.

### 2.1.4 Specific energy consumption

Specific energy consumption (SEC), the energy supplied to evaporate one kg of moisture is defined according to Equation (2.8) [21]

$$SEC = \left[\frac{Q(c_{pa}+c_{pv}Y_a)(T_d-T_{am})}{V_h}\right]\frac{t_{drying}}{m_v}$$
(2.8)

Where

 $C_{pa}$ -specific heat capacity of air (J/kg°C)

 $C_{pv}$ -specific heat capacity of water vapour (J/kg°C)

m<sub>v</sub> - mass of water removal (kg)

Q - Air flow rate  $(m^3/s)$ 

 $T_{am}$  - ambient temperature (°C)

 $T_d$  - drying air temperature at the inlet of the drying chamber (°C)

t<sub>drying</sub> - Total drying time (s)

 $V_h$  - humid volume or specific volume of air (m<sup>3</sup>/kg of dry air)

Y<sub>a</sub> - humidity of air (kg of moisture/ kg of dry air)

#### 2.1.5 Classification and selection of dryers

Dryers are normally categorized based on the method of heat transfer namely direct contact type and indirect contact type. Usually direct contact type dryers transfer heat by convection while indirect contact type transfers heat by conduction and radiation. In addition, dryers can be classified as either batch dryers or continuous dryers considering the method of operation. Furthermore, some researchers have classified either on the physical form of the feed or residence time of the product being dried in the dryer. Moreover, that some of the newer and novel dryers cannot be fit to the classification suggested in literature even they are not mentioned in most textbooks. About 50 types of dryers are widely used in practice to dry wide range of materials but more than 400 types of dryers have been cited in the literature [22]. Examples for some of the commonly used in the industry are rotary dryer, tray dryer, fluidized bed dryer, spray dryer, impingement dryer and flash dryer. Brief description of some of widely used dryers is provided in Table 1 of Appendix A.

There is no specified theory or principle for dryer selection. Many factors have to be considered to select a suitable dryer compromising among cost, efficiency, product quality and the clean environment. In addition, selection of suitable dryer for particular material is depending on the prior experience. Capital cost and operational cost, quality of the product, safety considerations, convenience of handling and installations have a direct impact on selecting a dryer for particular material. Quality of final product such as flavour retention of most farming crops depend on the method of drying and type of the operation such as batch or continuous mode. Usually drying processes cause air pollution by emissions of dust, gases and water vapor. Therefore, dryers are coupled with dust collection equipment such as cyclone separators, bag filters, scrubbers and electrostatic precipitator. In addition, absorption, adsorption and incineration units can be used for treatment of noxious gaseous pollutants exhausting from dryers.

Despite of lack of data availability for design and applications of spouted bed, many researchers have investigated the use of spouted bed dryers for agricultural products. Spouted bed gives many advantages such as small space requirement and etc. Section 2.3 elaborates more information on spouted beds.

### 2.2 Drying of agricultural crops

Drying of agricultural crops, as a post-harvest method is a requisite for safe storage in order to prevent product contamination due to microbial attacks. Preservation of agricultural crops through drying, dates back centuries is based upon solar power. Wide varieties of crops such as grains, oil seeds, spices and some perishable crops such as fruits and vegetables are subjected to drying. Development of alternate drying technologies has begun due to limitation in sun drying in particular uncontrolled weather conditions and mass production.

Many investigations have been conducted on conventional drying methods as well as new drying techniques. Conventional drying methods such as packed bed, cabinet or tray drying, fluidized bed drying are convective air-drying techniques. Most of new drying methods consist of state of art equipments with high capital and operational cost namely freeze drying, vacuum, osmotic, microwave and combinations of some of the above.

However, drying contributes to the loss of aroma, flavour and nutritive quality of product and hence careful selection of the best-suited drying technique and condition for a given product is extremely important. Many researchers have compared the drying methods by analysing traditional quality parameters such as: colour, taste, bulk density and rehydration ratio of product while some researchers use standardization technique such as measuring levels of vitamins, bioactive material and minerals [23].

#### 2.2.1 Black pepper drying

In case of black pepper drying, most widely used method of drying is solar drying. First pepper spikes are picked off from the fruiting branches of the vines at the correct stage and they are kept as such for a day for de-spiking. The most common pre-treatment for black pepper before drying is hot water blanching [24]. Blanching reduces the mold and other microbial contamination from the surface of the berries. In addition uniform glossy black color can be obtained to the product. After blanching, pepper is spread as a thin layer on mats or pavements for sun drying. Depending on the intensity of sun light, black pepper is dried in the sun for five to six days. Green color of the skin of the peppercorn is changed to black as a consequence of enzymatic reaction.

Solar dryers, cabinet dryers and flat-bed dryers are the commercially available artificial dyers in Sri Lanka. Renewable energy sources such as fuel wood, rice husk and saw dust as well as commercially available fossil fuels either kerosene or diesel is used as the fuel for hot air generation in the cabinet and flat-bed dryer [25]. Department of Export Agriculture of Sri Lanka recommends drying air temperature of black pepper in the artificial dryers should be in range of 55 °C to 60 °C. They emphasized that the drying air temperature should not exceed the 60 °C since volatile compound in the black pepper can be evaporated with reducing the quality. In addition, they recommend not to conduct the drying operation continuously but intermittently [26].

Solar thermal energy is used as the heating medium in the solar dryers. Different sizes of solar dryers are available in the market ranging from 10 to 60 kg of raw black pepper capacity. Chambers with green houses also used to dry large quantities of black pepper. Black pepper can be dried using solar dryers at low cost or no cost [26].

The cabinet dryer consists of two main components namely the tray drying chamber and the hot air generating unit. Hot air is entered into the drying cabinet supplied by an air blower. Maximum capacity of the cabinet dryer with bio mass driven air heater is approximately 300 kg of raw black pepper while total drying time would be round 18 to 20 hours. The cabinet dryer with kerosene driven air heater has maximum capacity of 450 kg of raw black pepper with around 12 hours of drying time. Positions of the trays during the drying operation have to be changed periodically to overcome the non-uniform drying [26].

The flat bed dryer also consists of two main components namely hot air generating unit and the drying chamber with a base of perforated metal plate. The raw black pepper is packed in the drying chamber and hot air passes through the perforated plate at the base by means of an air blower. Particles are mixed periodically to achieve uniform drying. Flat bed dryer with bio mass driven air heater has capacity of 1000 kg to 1500 kg while drying time would be 18 hrs to 20 hrs. Usually flat bed dryers are widely used since they have higher capacity and availability in market at low cost. Also the flat dryer is popular due to ease of operation in loading, unloading and mixing compared to the cabinet dryer [26].

#### 2.3 Spouted beds

#### **2.3.1 Introduction to spouted beds**

Spouted beds are gas-particle contactors which has been applied not only for drying of granular materials, paste with inert particles, solutions and suspensions but also in a wide range of operations and chemical processes, such as blending of solids, coating, granulation, cooling, combustion and pyrolysis, [27, 28]. Numerous experimental and theoretical investigations on spouted beds during the past five decades have been carried out with various types of particles such as bio mass particles namely barley, millet [29],carrot cubes [30], paddy [31], soy bean [32] corn [33], saw dust, etc. [34] and non-bio mass particles such as glass beads [35], sand [36].


Figure 2.3 : Schematic diagram of a conventional spouted bed [37]

A typical conventional spouted bed dryer is illustrated in Figure 2.3. This representation shows the fluid inlet, the upward movement of solids in the spout region and subsequent descending in the annulus region and the fountain region. The particles can be loaded either batch wise or continuous mode.

Heat carrying gas is introduced vertically upward in to the bed through the centrally located opening at the bottom of the vessel. Hot air jet causes a stream of particle to rise rapidly through the hollowed central core or spout within the bed of solid. Particles after rising to a height above the surface of the surrounding packed bed or annulus, rain back as a fountain on to the annulus [37].

Particles in the annulus region slowly move downwards and fluid from the spout leaks into the annulus and percolates through the moving packed bed and, to some extent inward as a loosely packed bed. The gaseous streams flowing away through annulus and spout region enter the fountain region, and both streams are mixed with each other and flow away the dryer through the fluid outlet of the dryer [36]. Pneumatic transport in the spout-region and the moving bed in the downcomer are the two distinct hydrodynamic regions depicted in a spouted bed. Below mentioned two hydrodynamic conditions have to be fulfilled to achieve a stable spouting regime in a spouted bed.

- i. The bed depth must be lower than the maximum spoutable bed depth
- ii. The gas flow rate has to be exceeding the minimum spouting velocity

Spouted bed dryer is suitable to handle coarse particles such as Group -D particles in the Geldart classification of particles [38]. Figure 1 and Table 2 of Appendix A gives more information on Geldart classification. Major difference between the spouted bed dryer and the fluidized bed dryer is the particle flow pattern. In a spouted bed the fluid is moving through the spout region using a nozzle rather than a perforated distributor as in a fluidized bed [39].

Ordinary fluidized bed experiences an oscillatory and more random particle flow pattern while particles in the spouted bed move upwards through the spout and come down through the annulus. Some of the key characteristics of gas spouted and gas fluidized beds dryers are compared in Table 3 in Appendix A.

The conventional spouted bed dryer is characterized by a cylindrical vessel with a conical base. It has undergone various modifications to improve its performance [27] such as spouted beds of different geometries [40], spout- fluid beds [41], insertion of either draft tubes [42] or draft plates [33], multiple spouts, etc. [43]. Spouted bed vessels may have several geometries such as cylindrical, conical-cylindrical, two dimensional (slot rectangular), triangular, paraboloid based and etc. Figure 2.4 shows different geometric configurations of spouted beds including conical-cylindrical, conical, two dimensional and triangular spouted bed.



Figure 2.4-: Spouted bed Configurations: (a) conventional, (b) conical (c) two dimensional with draft plate and (d) triangular spouted bed [27]

While particles moving down through the annulus region, they can enter into the spout region resulting a random behaviour of particles. Insertion of an axially positioned tube can reduce the random behaviour of the particles and improves the solid circulation and stability of spouting [45]. Different types of tubes namely non porous draft tubes, porous draft tubes and open sided draft tubes have been used for different processes. Diameter of the draft tube ( $D_T$ ), entrainment height ( $L_H$ ), length of the draft tube ( $L_T$ ) and opening ratio of porous draft tube are key design parameters for draft tube. Solid circulation, particle cycle time, gas distribution are governed by entrainment height. In addition, the hydrodynamic parameters such as minimum spouting velocity and operational pressure drop are also functions of the type of draft-tube used [46]. Usually draft tube diameter is selected as similar to the diameter of the inlet gas nozzle [37]. Figure 2.5 (a) and (b) show non porous draft tube and open sided draft tubes used by Altzibar et al. Figure 2.5 (c) shows a non porous draft tube fitted conical spouted bed also. [46].



(a) (b) (c) Figure 2.5: (a) open sided draft tube (b) non porous draft-tube (c) draft tube fitted conical spouted bed [46]

The characteristics of the conventional spouted bed versus spouted bed with nonporous draft tube and spouted bed with porous draft tube are listed in Table 4 of Appendix A for more information.

Spouted bed has good solid mixing and relatively large air velocity helps to maintain the constant product temperature. As a consequence of it spouted beds can eliminate the possibility of overheating when the moisture content of material is low at the end of drying period [47].

## 2.3.2. Minimum spouting velocity

Knowledge of minimum spouting velocity of a specific type of material is mandatory to design and scale up of a spouted bed. In conventional spouted beds, minimum spouting velocity can be determined by using the plot of pressure drop versus superficial gas velocity with increasing and decreasing the superficial gas velocities. In contrast to the conventional spouted beds, the hydrodynamics of the conical spouted beds are quite different. Minimum spouting velocity of conical bed is the gas velocity at the onset of internal spouting, which corresponds to the pressure peak point on the curve of the pressure drop versus superficial gas velocity obtained by increasing and decreasing the superficial gas velocity as shown in Figure 2.7.

Numerous correlations are available for determination of minimum spouting velocity of spouted bed systems as shown in Equation 2.9 to Equation 2.14 of Table 2.1. Among them, Equation 2.9; Mathur- Gishler correlation is the simplest predictor of the minimum spouting velocity for wide variety of solid materials, bed dimensions and nozzle diameters [37]. Moreover, Table 5 of Appendix A shows more details about existing correlations.

Further majority of spouted bed studies are also limited to experimental columns of diameter having 300 mm or less and studies on column of 1 m or larger in diameter are rare [48]. Consequently, Lim and Grace (1987) studied the hydrodynamics of a spouted bed column of diameter 0.91 m and found that the minimum spouting velocity is not well predicted by the correlations developed for smaller vessels.

The minimum spouting velocity of a spouted bed without draft tube is higher than that of a system with a draft tube. A spouted bed with open sided draft tube requires much higher values of minimum spouting velocity comparing to a spouted bed with non-porous daft tube [46, 45, 49].

Author	correlation	Equati
Aution	Correlation	on no
Mathur and	$(d) (D_i)^{1/3} \overline{2aH(\rho_n - \rho)}$	
Gishler	$U_{ms} = \left(\frac{\pi}{D}\right) \left(\frac{\pi}{D}\right) = \sqrt{\frac{-2\pi}{\rho}}$	2.9
1955	N T	
[50]		
Fane and	$(d_{m}) (D_{i})^{1/3} \overline{2aH(a_{m}-a)}$	
Mitchell	$U_{ms} = 2.0 D^n \left(\frac{np}{D}\right) \left(\frac{-1}{D}\right) = \sqrt{\frac{-2.0 (p-p)}{\rho}}$	2.10
(1984)	N	
[48]		
Wu et al.	$U_{ms} = 10.6 \left  \frac{d_p}{d_p} \right ^{1.05} \left  \frac{D_o}{D_o} \right ^{0.266} \left  \frac{H_o}{d_b} \right ^{-0.095} \left  \frac{\rho_s - \rho}{\rho_s - \rho} \right ^{0.256}$	2.11
(1987)	$\frac{1}{\sqrt{2gH_o}} = 10.0 \left[ \frac{1}{D_c} \right] \qquad \left[ \frac{1}{D_c} \right] \qquad \left[ \frac{1}{D_c} \right] \qquad \left[ \frac{1}{D_c} \right]$	
[51]		
Olazar et al.	$(Be)_{max} = 0.126 A r^{0.5} \left(\frac{D_b}{D_b}\right)^{1.68} \left(\tan\frac{\gamma}{D_b}\right)^{-0.57}$	
(1992)	$(10_0)_{ms} = 0.12011  (D_0)  (112)$	2.12
[52]		
San José et	$(Re_o)_{ms}$	2.13
al., (2007)	$= 0.126 4m^{0.5} (D_b)^{1.68} [\tan \gamma]^{-0.57} (H_o - l_d)^{0.45} (D_i)^{0.17}$	
[53]	$= 0.126AT^{-1} \left( \frac{D_o}{D_o} \right)   \tan \frac{1}{2}   \left( \frac{1}{H_o} \right)  \left( \frac{D_i - d_d}{D_i - d_d} \right)$	
Altzibar et	$(R_{e}) = 0.2044r^{0.475} \left(\frac{H_{o}}{L_{H}}\right)^{1.240} \left(\frac{L_{H}}{L_{H}}\right)^{0.168} \left(\tan\frac{\gamma}{L_{H}}\right)^{-0.135}$	2.14
al. (2009)	$(D_o) = (D_o) + (D_T) + (U_T) + (U_T$	
[54]		

Table 2.1:Correlations for minimum spouting velocity

#### 2.3.3. Maximum spoutable bed height

There are three mechanisms, which causes spouting to become unstable for a given limit of height. They are fluidization of solids in the upper surface of the annular zone, choking of spout and propagation of surface instability created at the base of the bed [55]. Therefore, maximum spoutable bed height is an important parameter when improving the volumetric processing capacity and as well as the scale up operation. Several researchers have studied maximum spoutable height in spouted beds and various correlations were developed for the prediction as shown in Table 6 in Appendix A [37, 56].

## 2.3.4. Pressure drop

Numerous researches have been carried out to investigate the pressure drop in spouted beds of different geometries with different types of particles and fluids. Various correlations are available in the literature for determination of maximum pressure drop of spouted bed operation as shown in Table 7 in Appendix A.

When the gas flow rate is increased, certain pressure drop is built up through the bed of particles. The change in pressure drop with increasing and decreasing flow is caused by the different packing conditions of the bed particles. Figure 2.6 shows the photographic sequence of an evolution of spouting process of polyethylene terephthalate (PET) chips performed in a squared-based half sectional 0.2 m side unit by group of researchers [56]. Figure 2.7 describes the consequent hydrodynamic evolution of spouting process, which express the pressure drop versus flow rate hysteresis between an increasing and the decreasing flow of spouted bed.



Figure 2.6: Photographic sequence of the evolution of a spouting process of PET chips performed in a squared-based half sectional 0.2 m side unit [56]



Figure 2.7: Pressure drop versus. flow rate hysteresis between an increasing and the decreasing flow in a spouted bed of PET chips [56]

#### 2.4 Dimensionless models for analysis of heat transfer in spouted bed dryers

Designing of a dryer and evaluation of drying time and drying rate require the knowledge of heat and mass transfer. Mathematical models which used to simulate the temperature and moisture content of the material and the drying gas during a drying process requires the input of heat and mass transfer coefficients.

Many studies have been carried out on analysis of heat and mass transfer in spouted bed dryers. However, the consistency of the heat transfer coefficients estimated from the existing correlations is very poor. Some shortcomings are appeared on them such as assuming the whole drying period is a constant rate drying period. When heat transfer coefficients are estimated from existing correlations, difference of one or more orders of magnitude occurs for same value of input parameters. Many existing correlations are not concerned about the flow pattern of the gas and particles and the quality of spouting [57].

Generally, Nusselt number (Nu) is used to express the heat and mass transfer coefficient as a function of one or more dimensionless groups such as Reynolds number (Re), Prandlt number (Pr), the ratio of dryer vessel diameter to particle diameter, the ratio of vessel diameter to static bed height, etc. The knowledge of interfacial surface area, heat flow and the driving force; temperature difference is required to calculate the heat transfer coefficients between the particle and fluid stream [57].

Many authors have interpreted their drying processes have only constant rate drying period without analysing the surface behaviour of the material being dried. Very limited literature is available to determine the drying process is in whether constant rate or falling rate drying period by direct measurement of surface behaviour. However many investigations have been conducted by analysing surface temperature and drying kinetics [58].

In many recent research works of computational fluid dynamics (CFD) modelling of spouted bed drying, existing correlations have been used to evaluate the heat transfer coefficient [59]. However, most of the existing correlations were developed based on

either the external control conditions or in the steady state conditions. Therefore, it is questionable to achieve good agreement with experimental value and the simulated value because some drying processes have falling rate drying periods as well.

A number of investigations on heat transfer between particle and fluid in spouted bed dryers are those of among the works of Prachayawarakorn et al. [39], Kmiec [60], Kudra et al. [61], El-nass et al. [62], Rocha et al. [63], etc. In case of spouted beds which supplies additional heat or remove heat out of the bed, wall to bed heat transfer coefficient were evaluated by some workers [64, 65].

Fluid to particle heat transfer coefficient was evaluated and several correlations were developed by many researchers either applying dimensional analysis or empirically. Equation 2.15 to Equation 2.21 in Table 2.2 show the correlations associated with heat transfer of spouted beds [60, 63].

Equation 2.15 was developed after analysis of simultaneous heat and mass transfer during batch drying of silica gel and activated carbon in spouted bed by Kmiec assuming the overall process had the constant rate dying period [60]. Prachayawarakorn et al. developed two correlations; Equation 2.16 and Equation 2.17 for heat transfer coefficient for downcomer (annulus) and spout regions separately for continuous drying of agricultural materials in spouted bed with draft plates [39]. Oliveira and Feire studied continuous drying of liquid materials in conical spouted bed [66]. One of the hypotheses used by them was that drying took place at the external control condition that is called as constant drying rate period. Owing to great deviation of experimentally obtained heat transfer coefficient with values obtained from existing correlations, they developed a correlations using non-linear regression as shown in Equation 2.19. Dimensionless equation for heat transfer coefficient shown in Equation 2.20 in Table 2.2 was developed by Kudra et al., for particulate drying in two dimensional spouted bed in the region of constant rate drying period [61].

		Equa
Author	Correlation	tion
		no
Kmiec (1975)	$N_{\rm H} = 0.807 P_0 e^{0.464} P_r^{0.333} \Lambda r^{0.116} (t_{20} \theta)^{-0.813} (H_0)^{-1.19} \phi^{2.261}$	2.15
[60]	$\operatorname{Hu} = 0.097 \operatorname{He} \operatorname{Hi} \operatorname{Hi} \left( \operatorname{tan} \frac{1}{2} \right) \left( \frac{1}{d_p} \right)  \emptyset$	
Prachayawara	$H_{\rm d} = 42.07 \text{ p}_{-0.454} \left( \text{H}_{\rm d} \right)^{-1.006}$	2.16
korn et al.	$Nu = 42.07 \operatorname{Re}_{p,d}^{-1} \left( \frac{d_p}{d_p} \right)$	
(2006) [39]		
Prachayawara	$N_{\rm H} = 0.4 {\rm P}_{\rm s} 0.779 \left( {\rm H}_{\rm s} \right)^{-0.81}$	2.17
korn et al.	$NU = 0.4 \operatorname{Re}_{p,s}^{0.7} \left( \frac{1}{d_p} \right)$	
(2006) [39]		
Englart et al	Nu	2.18
(2009) [67]	$= 0.0030 \text{Re}^{0.836} \text{Pr}^{0.333} \text{Ar}^{0.236} \left(\frac{\text{D}_0}{\text{d}_p}\right)^{3.35} \left(\frac{\text{D}_b}{\text{d}_p}\right)^{-4.121} \left(\frac{\text{m}_w}{\text{m}_g}\right)^{0.600} \emptyset^{-0.918}$	
Oliveira &	$N_{\rm W} = 0.000625 \left( \frac{W_{\rm l}}{W_{\rm l}} \right)^{1.189} p_{\rm c}^{0.991} C_{\rm W} = 1.855 C_{\rm l}^{10.264} A_{\rm r} = 0.004 P_{\rm r}^{-0.333} / A_{\rm r}$	
Feire	$Mu = 0.000825 \left(\frac{W_g}{W_g}\right)$ $Re_i^{a} Gu = Gu = G_i^{a} Gu = Gu = M_i^{a}$	2.19
(1996) [66]		
Kudra et al.	$N_{\rm H} = 1.075  {\rm Pe}_{0.64} \left( {\rm H} \right)^{-1.20} \left( {\rm H} \right)^{0.45} \left( {\rm s} \right)^{0.26}$	
(1989) [61]	$Nu = 1.975 Re^{-1} \left(\frac{1}{d_p}\right) = \left(\frac{1}{w}\right) = \left(\frac{1}{d_p}\right)$	2.20
Reger et al	$N_{\rm H} = 0.0597 \text{Re}^2 \text{Ar}^{-0.438} \text{Gu}^{0.61} \left(\frac{\text{H}}{\text{H}}\right)^{-1}$	
(2011) [27]	$du = (d_p)$	2. 21

Table 2.2 : Correlation associated with heat transfer in spouted beds

### 2.5 Drying models

Mathematical modelling of drying processes describes the behaviour of drying. They are used to estimate the drying time and temperature profile of the material being dried and the drying gas for selected moisture reduction without operating a real dryer [68]. Furthermore, mathematical modelling generates data to scale up, design and control of existing dryers to identify optimum dimensions of the dryer and operational parameters [69].

Early studies on spouted bed was focused on analysis of effect of operating conditions (drying temperature, bed height and particle diameter) and contactor geometry (air inlet diameter and cone angle) on drying behaviour and hydrodynamic behaviour of gas and particulates. However many number of recent studies on spouted bed drying have mainly focused on numerical simulation of drying behaviour using modelling [27].

According to the Passos et al., published models for drying of solids in a spouted bed dryer can be categorized into three levels namely zero level, one level and two level modelling [27]. Zero level models, which consist of simple algebraic equations, are based on the application of simple mass and energy balances for both gas and solid phases at the inlet and outlet of the dryer. In zero level models, once both inlet gas phase and solid phase conditions and outlet solid phase conditions are known, outlet air conditions can be determined.

One level models are composed of overall mass and energy balance for the both gas and solid phases at the inlet and outlet of the dryer and the sorption curve of solids [36].

Two level models include the conservation of mass and momentum equation and additional closure laws to describe the solid stress, interfacial forces and turbulence of two phases. Particle -particle and particle-fluid interactions are considered using these additional closure laws. Two level models are categorized into 2 types; A and B. In 2A models, gas and water vapour mixture assumed to be behaved as a

continuous mixture while particulate solids are to be considered as a discrete phase. 2B models are complex and they consider the gas and solid particulate flow in three directions while only statistically mean particle path and main gas flow direction are considered for 2 A model. Computational fluid dynamics (CFD) packages are being used for solving of 2B model equations as they are more complex.

Application of Computational fluid dynamics (CFD) improves the understanding of hydrodynamic behaviour of spouted beds. CFD has been applied to conventional, conical ,two dimensional spouted beds [70], spouted fluid beds [30] and spouted bed with either draft tubes or plates [71] to get an accurate prediction of drying behaviour. Computer packages 'Fluent', 'Comsol Multi Physics', etc have been used to facilitate the model [30, 72, 73].

So far, there have been many numbers of mathematical models in literature to describe the drying behaviour of many agricultural products. However another concise analysis of drying model by Jittanit et al. showed that models can be categorized as empirical models, semi empirical models and models based on heat and mass balances, etc. [69]. Among them, thin layer drying models which frequently having a semi empirical relationship are popular within many researchers. Although thin layer drying models are popular, many of them are limited to specific equipment, dryer configuration and specific material and operational conditions. Thin layer drying modes are widely applied by many researchers due to its simplicity and lack of required data [74].

Thin layer drying models are useful in designing new drying equipments, improving existing drying processes and for prediction of drying time accurately [75]. Furthermore, they contribute to get an understanding about the heat and mass transfer of the drying systems [15].

Thin layer drying models are categorized as theoretical, semi-theoretical and empirical models [74]. They describe the variation of moisture ratio (MR) of the material with the time during the drying process. Ficks' second law of diffusion is the most widely applied theoretical model for drying and it considers the internal resistance to transfer moisture between the drying material and the air. Semitheoretical and empirical models consider the external resistance to moisture transfer between material and the air. However, they can be applied within a range of operational parameters such as drying temperature, air flow rate, initial moisture content, and for a particular drying material that they developed. Numerous semi theoretical and empirical models are available in literature to fit the data of drying kinetics of many agricultural products. Some of them are listed as shown in Equation 2.22 to Equation 2.33 in Table 2.3. The Newton model, Page model, Henderson and Pabis model, Two component model and Logarithmic model are some of semi theoretical models while Wang and Sing model and Thompson model are empirical models available in literature. Coefficients of some thin layer models have been correlated to operational parameters of the process such as drying temperature, air velocity, etc. [76, 77, 78, 79].

Table	2.3:Thin	laver	drving	models

Model name	Model	Equati on no	Reference
Newton	$MR = \frac{X_t - X_e}{X_0 - X_e} = e^{-kt}$	2. 22	[80]
Page	$MR = \frac{X_t - X_e}{X_0 - X_e} = e^{-kt^n}$	2. 23	[81]
Modified Page	$MR = \frac{X_t - X_e}{X_0 - X_e} = e^{(-kt)^n}$	2.24	[81]
Henderson and Pabis	$MR = \frac{X_t - X_e}{X_0 - X_e} = ae^{-kt}$	2.25	[82]
Logarthmic	$MR = \frac{X_t - X_e}{X_0 - X_e} = ae^{-kt} + b$	2.26	[76]
Two compartment	$MR = \frac{X_t - X_e}{X_0 - X_e} = ae^{-k_1 t} + be^{-k_2 t}$	2. 27	[80]
Abbasi et al. ( Modified Middilli- Kucuk)	$MR = \frac{X_t - X_e}{X_0 - X_e} = ae^{-kt^n} + b$	2. 28	[81, 75]
Midillli - Kucuk	$MR = \frac{X_t - X_e}{X_0 - X_e} = ae^{-kt^n} + bt$	2. 29	[75]
Hii et al.	$MR = \frac{X_t - X_e}{X_0 - X_e} = ae^{-k_1 t^n} + be^{-k_2 t^n}$	2.30	[83]
Wang and Sing model	$MR = \frac{X_t - X_e}{X_0 - X_e} = 1 + at + bt^2$	2. 31	[84]
Verma et al. (modified two term exponential) Model	$MR = \frac{X_t - X_e}{X_0 - X_e}$ = $aexp(-k_1t) + (1 - a)exp(-k_2t)$	2. 32	[75,81]
Thompson model	$MR = \frac{X_t - X_e}{X_0 - X_e} = alnMR + b(lnMR)^2$	2. 33	[75]

a, b, k,  $k_1$ ,  $k_2$  and n are drying coefficients and constants

Thin layer drying models are applied when either thin layer of material is maintained or the air velocity is high enough to cause good solid mixing to have uniform drying. Moreover, the operating conditions of the drying process such as air temperature and humidity are kept constant throughout the process [85].

According to Kuck et al., 67 potential thin layer drying models are available in the literature from 2003 to 2013 [75]. Goodness of fit was evaluated under 28 performance assessment criteria. Coefficient of determination( $\mathbb{R}^2$ ), root mean square error (RMSE), reduced chi square ( $\chi^2$ ), mean relative percentage error, standard error of estimate are among the most used benchmarks. In order to evaluate the goodness of fit, many researchers used highest values of either coefficient of determination or modelling efficiency, and lowest values of one of these parameters such that reduced chi-square, the root mean square error and the mean relative percentage error [75].

Mathematical modelling on thin layer drying processes has been conducted on many agricultural products such as cocoa [83], mint, parsley [86], apple [87], green pepper [19], banana [88], eggplant [89], leek slices [90], jackfruit [91], etc. Most of the drying processes were conducted using an air ventilated oven [83], open sun drying [86], packed bed and fluidized bed dryers [92, 93].

According to Jittanit et al., thin layer drying models are applied to spouted bed drying process since regular circulation of particles in the spouted bed achieves uniform moisture content and uniform temperature of samples [80]

A limited published literature is available for analysis of thin layer models in spouted bed drying process [79, 77, 33, 32]. The thin layer drying models are evaluated for spouted bed dryers considering the product type, drying parameters such as temperature [80], air velocity, bed depth [79], etc. It was hardly found for considering the configuration of the draft tubes. Although many researches have been carried out on mathematical modelling of drying of agricultural crops, published data available on black pepper drying in spouted beds is very limited.

#### 2.6 Scale up

Scaling up of a spouted bed can be approached in two ways either changing the geometry of the conventional spouted bed or modifying the spouting operation [94]. The first approach implies increasing the size of a single unit or changing geometry of conventional spouted bed in to two dimensional spouted bed or conical spouted bed. Later is concerned to insertion of auxiliary device such as draft tube, draft plate, mechanical devices or introduce additional fluid in the annular region. Design and constructions of both options have to be conducted carefully to minimize heat loss, investment and operating cost since they have some advantages and drawbacks. Table 8 of Appendix A shows more information on scale up of spouted beds.

#### 2.7 Research on black pepper essential oil

Essential oil of black pepper contains the volatile organic compounds. They which imparts the flavor and perfumery to the black pepper have been studied by various researchers [95, 96]. The volatile oil component of black pepper contains terpene hydrocarbons and oxygenated terpenes. 90% of terpene hydrocarbon consists of monoterpene and sesquiterpenes. The chemical formula of monoterpene and sesquiterpene are  $C_{10}H_{16}$  and  $C_{15}H_{24}$  respectively. Terpene hydrocarbons possess main desirable attributes of black pepper flavor. Oxygenated compounds which present less than 4%, imparts the characteristic musty, moldy odor and off flavor of black pepper [96].

A research conducted by Mccarron *et. al.* found that  $\beta$  -Pinene and Carryophellene ( $\beta$  caryophyllene) and Sabinene were commonly available compounds in Sri Lankan and Indian black pepper derived essential oil [6]. Another study reported by Menon et. al. added three more compounds to the main components of Indian black pepper oil namely delta-3 Carene, Limonene and  $\alpha$ -Pinene. They analyzed four selected cultivars of black pepper and identified 55 components of volatile organic compounds [95]. Buckle et al. have recommended the local Sri Lankan cultivars for perfumery industry than Indian and Sarawak cultivars since Sri Lankan cultivars were rich in  $\beta$  caryophyllene which is the abundant component of sesquiterpene

hydrocarbons [97]. According to Mccarron et al. and Buckle et al., the composition of black pepper oil depends on the geographical conditions and varietal variations and post-harvest methods. In fact recent studies carried out with comprehensive gas chromatography–mass spectrometry (GC-MS) has identified more than 300 compounds in essential oil derived from black pepper [98]. In a recent research of Jelen and Gracka, 273 components have been identified [96].

## 2.8 Quality of black pepper

Depending on the country, specification for quality requirement of black pepper is varied. ASTA (American spice trade association), ESA (Europe spice association, CSA (Canadian spice association) are some of authorized associations pertaining to assure quality of spices and herbs in the global market. Table 9 of Appendix A consists of some specifications required for black pepper defined by several institutes related to spice and herb.

According to the European spice association (ESA) specifications of quality minima for black pepper should have 12% w/w of moisture content and minimum 2% v/w of volatile oil content for trading within the Europe [99]. Table 10 of Appendix A consists of the quality standards approved by the Sri Lanka Standard Institute.

## **Chapter specific nomenclature**

a, b, k, k<sub>1</sub>, k<sub>2</sub> & n - drying coefficients and constants

A -cross section of the bed measured at right angles to the air flow

Ar - Archimedes number

Cl - liquid concentration in the material (kg/kg)

 $C_{\text{pa}}$  and  $C_{\text{pv}}$  - specific heat capacity of air and vapour respectively (J/kg $^{\circ}C)$ 

D, D<sub>c</sub> - diameter of the column (m)

d, d<sub>p</sub> - particle diameter , horizontally projected diameter

D<sub>0</sub> -pre exponential factor of Arrhenius equation

 $D_{\text{b}}$  - upper diameter of the stagnant bed in conical spouted bed

 $D_{eff}$  -effective moisture diffusivity (m<sup>2</sup>/s)

D<sub>i</sub>- gas inlet diameter (m)

D<sub>T</sub>, d<sub>d</sub>-diameter of draft tube (m)

E<sub>a</sub> -activation energy (kJ/mol)

F<sub>0</sub> -Fourier number

Gu- Guckhman number

H<sub>,</sub> H<sub>o</sub> -static bed depth (m)

H<sub>d</sub> static bed height in downcomer region (m)

H<sub>s</sub> static bed height in spout region (m)

H<sub>T</sub>- tube height (m)

L<sub>H</sub> -entrainment height (m)

 $L_T$ ,  $l_d$  - total height of the draft tube

m<sub>v</sub>- mass of water removal (kg)

 $\dot{m}_q$ - mass flow rate of the gas (kg/s)

 $\dot{m}_{w}$ - mass flow rate of the liquid (kg/s)

N - drying rate or (mass moisture evaporated/(area)(time) (kg/m<sup>2</sup>s)

Nu- Nusselt number

Pr- Prandlt number

Q- air flow rate  $(m^3/s)$ 

R -universal gas constant (kJ/mol K)

Re - Reynolds number

(Re<sub>o</sub>)<sub>ms</sub> - Reynolds number of minimum spouting, referred to gas inlet

R<sub>p</sub>-radius of the spherical particle (m)

s - slot width in two dimensional draft tubes (m)

T -absolute temperature (K)

t -time (s)

 $T_{am}$  - ambient temperature (°C)

 $T_d$ - inlet air temperature to the drying chamber or drying temperature (°C)

t<sub>drying</sub> - Total drying time (s)

U<sub>ms</sub> - minimum spouting velocity (m/s)

 $V_h$  - humid volume or specific volume of air (m<sup>3</sup>/kg of dry air)

W - bed width (m)

W<sub>d</sub> - mass of dried solid

Wg - gas mass velocity (m/s)

W<sub>1</sub>-liquid mass velocity (m/s)

X - moisture content dry basis

X<sub>0</sub> - initial moisture content (kg/kg dry basis)

 $X_c$  - critical moisture content

Xe -equilibrium moisture content (kg/kg dry basis)

X<sub>t</sub> -moisture content at time t (kg/kg dry basis)

Y<sub>a</sub> - humidity of air (kg of moisture/ kg of dry air)

 $\Phi$  - particle shape factor,  $\Phi > 1$ 

 $\gamma$ ,  $\theta$  - angle of the conical base of the contactor, degrees

 $\rho$  - density of air or fluid

 $\rho_p$ ,  $\rho_s$ - density of particle, kg/m<sup>3</sup>

# Abbreviations

SEC - specific energy consumption

ASTA - American spice trade association

ESA - Europe spice association

CSA - Canadian spice association

MR - moisture ratio

# **Chapter 3**

# **3. METHODOLOGY**

This chapter describes the material preparation for spouted bed experiments, experiments for determination of spouting behavior of black pepper, black pepper drying experiments, determination of heat transfer coefficient, methodology of dimensional analysis of heat transfer process and mathematical model development of spouted bed drying process.

### **3.1 Experimental methods**

In this section, experiments conducted to accomplish the objectives of this study are described, including the materials, equipment, instruments and operational parameters used.

A number of experiments were carried out to study the spouting behavior and drying of black pepper to accomplish the objectives stated in Chapter 1. The experiments are categorized in to two sections as preliminary experiments and major experiments. Preliminary experiments were conducted to identify the best configuration of spouted bed among two types of draft tube fitted spouted beds and the conventional spouted bed. Major experiments were conducted to examine optimum spouted bed parameters to achieve black pepper with maximum essential oil yield, to model heat transfer characteristics and to develop mathematical model to describe drying behavior.

Preliminary experiments were conducted using wetted black pepper and section 3.1.2 describes the preparation of wetted black pepper. Preliminary experiments are categorized as:

- Finding minimum spouting velocity
  - In conventional spouted bed
  - In spouted bed with internal devices
    - Non-porous draft tube fitted spouted bed
    - Porous draft tube fitted spouted bed

- Drying experiments
  - In conventional spouted bed
  - In spouted bed with internal devices
    - Non-porous draft tube fitted spouted bed
    - Porous draft tube fitted spouted bed

Major experiments were conducted in the conventional spouted bed using raw black pepper and section 3.1.1 describes the methods used to handle raw black pepper prior to experiments. Major experiments are categorized as:

- Finding minimum spouting velocity
- Drying experiments
- Extraction of essential oil

## 3.1.1 Handling of raw black pepper for experiments

Raw black pepper (matured, unripe green colour) with their spikes collected from a farm located in Gampaha District of Sri Lanka was used for conducting major experiments. The initial moisture content of raw black pepper seeds ( $X_0$ ) was 3.50 kg moisture/kg of dry solids (350%). The raw black pepper were stored in a refrigerator at 2-5 °C to avoid spoilage. Required amount of raw black pepper was taken out of the refrigerator around one hour before the experiment. Then seeds were removed from their spikes manually and allowed to reach the room temperature.

The Precision balance; VWR ECN 611-2300 with reading accuracy of 0.01g was used to measure the required amount of seeds. Figure 3.1 shows a tray of raw black pepper with their spikes and prepared black pepper seeds for experiments.



Figure 3.1: Raw black pepper preparation for experiments (a) black pepper spikes (b) raw black pepper seeds

## 3.1.2 Preparation of wetted black pepper

Since black pepper is a seasonal crop and it is not available throughout the year, wetted black pepper was used to conduct the preliminary experiments. Preliminary experiments were conducted to determine the best configuration of the spouted bed among the draft tubes fitted spouted bed and the conventional spouted bed. Effect of configuration of draft tubes such as draft tube diameter ( $D_T$ ), entrainment height ( $L_H$ ) and draft tube height ( $H_T$ ) on drying kinetics were analysed by conducting drying experiments with the wetted black pepper.

Dried black pepper was purchased from Jayantha Agrochemicals of Galle, Sri Lanka. Dried black pepper usually has moisture content 8-15% dry basis. They were rewetted to moisture content around 80% dry basis, as described in next paragraph.

Dried black pepper seeds were wetted by adding distilled water. Dried black pepper was put in to a polypropylene container and predetermined amount of distilled water was added and mixed to wet the surface of the particles and absorb more water. Then they were kept in a refrigerator at 3-5 °C temperature for 5-6 days in the tightly closed container, until black pepper particles reached to the stable conditions. Samples of black pepper were taken out and moisture content was measured daily using the direct and indirect measuring methods.

### 3.1.3 Determination of moisture content of black pepper particles

Moisture content of the solid is the key parameter for investigations in drying processes. Moisture content is determined by direct methods or indirect methods [100].

The direct method consists of drying known sample of solid material in a drying oven at 102  $^{\circ}$ C -105  $^{\circ}$ C until the weight of the samples becomes constant and measurement of sample weight before and after the drying process. Then the moisture content, X (% dry basis) is given by Equation (3.1).

$$X(\% drybasis) = \frac{W_w W_d}{W_d} \times 100$$
(3.1)

Where

W<sub>w</sub>- initial weight of the sample (kg)

W<sub>d</sub>-weight of sample after drying in the oven (kg)

For the direct measurement of moisture content, a drying oven; A lab Tech LDO-060E with reading accuracy of 0.5 °C was used to dry black pepper samples. Weight of black pepper samples were measured by a portable digital balance; KERN PCB 350-3 in the range of 0-300g and accuracy of  $\pm$  0.001g.

Digital Moisture Analyzer (Citizen, MB 200X), with reading accuracy of  $\pm 0.01$  kg/kg was used for the indirect measurement of moisture of the black pepper seeds.

### 3.1.4 Experiment set up and equipment

A laboratory scale spouted bed dryer was designed and got fabricated by Mega Heaters (Pvt) ltd, Kottawa, Sri Lanka. The dryer consists of an air blower with variable speed drive (VSD) to control the airflow, air heater with control unit, spouted bed contactor, temperature measurement sensors, a U tube manometer and a cyclone separator as shown in Figure 3.2 and Figure 3.3.

The cyclone separator is reverse flow and used to capture entrained fine particles. The spouted bed contactor is a cylindrical vessel with conical bottom. The dimensions of the spouted bed column are: the diameter of the cylindrical column  $(D_c)$  or diameter of the upper section of the conical base, 0.15 m; angle of conical section, 60° and the height of the conical section 0.069 m. The gas inlet diameter  $(D_i)$  can be set for any of three values; 0.029 m, 0.035 m and 0.05 m. The total height of the vessel (conical bottom plus cylindrical column) is approximately 1.07 m. A perforated plate is fixed at the bottom of the spouted bed column which prevents black pepper seeds entering the gas inlet tubes. The dryer could be adjusted to any preferred drying air temperature between room temperature to 120 °C. The particles to be dried are packed in the spouted bed column. The desired drying air temperature is attained by electric heater and the temperature control unit.

The entire dryer is made of stainless steel. The bottom of the cone section of the spouted bed dryer is connected to the heater and blower while the reverse flow cyclone separator is channelled to top of spouted bed contactor to capture entrained fine particles. Tables 3.1 summarizes the geometric factors of spouted bed set up.



Figure 3.2: Schematic diagram of the spouted bed drying experiment Set up; 1-VSD, 2-Blower, 3,4-Air heater with temperature control system, 5,7- Temperature indicators, 6- Spouted bed column, 8- Cyclone separator

The blower is operated by a 2.2 kW motor and the air heater has a capacity of 12 kW. Pressure measurements were carried out by means of a U tube water manometer. Two Pt100 thermometers are located at the spouted bed inlet and outlet to measure the temperatures of the inlet and exit air. In all experiments, the fluctuations of temperature were within  $\pm 1^{\circ}$ C.

Five temperature sensors; DS18B20-PAR with  $\pm$  0.5 °C accuracy were placed inside the spouted bed. Temperature sensors are compatible with Arduino development boards and coding was done to communicate with sensors. Arrangement of temperature sensors are shown in Figure 3.4. Schematic diagram of arrangement of temperature sensors in spouted bed column is shown in Figure 4 in Appendix C. Measurements of temperature inside the spouted bed were used for the modelling of heat transfer to determine the heat transfer coefficients.



Figure 3.3: Image of spouted bed drying experiment set up



Figure 3.4: Arrangement of temperature sensors inside the spouted bed contactor

spouted bed -Stainless steel				
Column diameter	$D_{C}(m)$	0.15		
Cone angle	$\Gamma$ (deg)	60		
Height of the cylindrical sections	$H_L(m)$	1		
Height of the cone sections	$H_{c}(m)$	0.07		
Gas Inlet diameters	$D_i(m)$	0.049	0.035	0.024
Cone base diameter	$D_o(m)$	0.05		

Table 3.1: Geometric factors of the spouted bed

Air velocity at the inlet was measured by means of EXTECH CFM Thermo anemometer model 407113 with an accuracy of  $\pm 2$  %. Relative humidity of air at outlet of the dryer and the ambient air were measured by Thermo-Hygrometer Model GMK-920HT with an accuracy of  $\pm 2$ %.

The draft tubes are cylindrical tubes with three arms made of stainless steel and placed collinear with the axis of cylinder. Two types of draft tubes were used

separately. They are namely porous draft tube and nonporous draft tubes having similar configuration except to the perforations as explained in section 2.3. Figure 3.5 (a) and (b) illustrates non-porous and a porous draft tubes respectively.

The draft tube was fixed using its three arms at the bottom of the vessel collinear with the axis of spouted bed column. The conical base of the dryer allows fitting a draft tube at the inlet of conical section as shown in Figure 3.6. Distance between the base of the arm and the base of the tube is called as the entrainment height. As per the literature, dimensional parameters of a draft tube is categorized as diameter of draft tube (D<sub>T</sub>), entrainment height (L<sub>H</sub>), tube height (H<sub>T</sub>=L<sub>T</sub>-L<sub>H</sub>), width of the arm where L<sub>T</sub>- total height of the draft tube [46].

Width of arm was kept constant during this study while changing the other parameters. Sixteen draft tubes with different configuration have been used. Seven of them were non-porous and other nine were porous. Table 3.2 & 3.3 show the dimensions of both types of draft tubes used.

Figures of all the equipment and instruments used for experiments are listed in Appendix B. More information on spouted bed contactor, draft tube and the reverse flow cyclone separator are also available in Appendix C.



Figure 3. 5: a) non porous draft tube (b) porous draft tube [101] and (c) Image of both type od draft tubes



Figure 3. 6: A porous draft tube is ready to fit into the spouted bed contactor

Draft tube no	Draft-tube diameter	Drat-tube	Draft-tube
	$(D_T)(m)$	entrainment	height (H <sub>T</sub> )
		height (L <sub>H</sub> )	(m)
		(m)	
1	0.035	0.060	0.016
2	0.035	0.080	0.016
3	0.035	0.060	0.020
4	0.035	0.060	0.024
5	0.035	0.030	0.016
6	0.029	0.030	0.016
7	0.050	0.030	0.016

Table 3.2: Dimensions of non-porous draft tubes

Table 3.3: Dimensions of porous daft tubes

Draft tube no	Draft-tube	Drat-tube	Draft-tube height
	diameter $(D_T) (m)$	entrainment height	(H <sub>T</sub> )
		(L <sub>H</sub> ) (m)	(m)
1	0.050	0.030	0.016
2	0.060	0.030	0.016
3	0.070	0.030	0.016
4	0.035	0.040	0.016
5	0.035	0.060	0.016
6	0.035	0.080	0.016
7	0.035	0.060	0.020
8	0.035	0.060	0.024
9	0.035	0.060	0.080

#### **3.1.5 Preliminary experiments**

Spouting behavior of wetted black pepper was studied to determine the minimum spouting velocity of the spouted bed of three different configurations namely conventional spouted bed, spouted bed with porous drat tube and spouted bed with non-porous draft tube fitted systems. Experiments were conducted for a selected bed height; 16 cm of wetted black pepper. Way of determining minimum spouting velocity is similar to those of the experiments conducted with raw black pepper in conventional spouted bed. More details are elaborated in section 3.1.6.1.

Effects of geometric factors such as height, diameter and height of entrainment zone of draft tubes on drying were determined. Samples of wetted black pepper with initial stagnant bed height of 16 cm were dried in spouted bed fitted with different draft tubes at their minimum spouting velocity at 75 °C drying temperature. Seven non-porous draft tubes with different dimensions and nine porous draft tubes with different dimensions were used.

Drying experiments with wetted black pepper were conducted using three configurations of spouted bed. A number of experiments were conducted in a limited range of drying conditions to identify the best configuration. Experimental conditions are summarized in Table 3.4 to Table 3.6. When analysing the effect of spouted bed configurations, two superficial air velocities were used. Air velocity was set at their minimum spouting velocities and in a higher velocity; 0.89 m/s to ensure the stable spouting operation with all possible bed heights without pneumatic transport of particles out to the cyclone separator.

Configuration	Air velocity	Stagnant bed	Temperature		
	(m/s)	height (cm)	(°C)		
Conventional spouted bed	0.74	16	55, 65, 75		
*Non porous draft tube fitted spouted bed	0.51	16	55, 65, 75		
*Porous draft tube fitted spouted bed	0.58	16	55, 65, 75		
* $D_T$ = 0.035m, $L_H$ = 0.06 m, $L_T$ = 0.26 m					

Table 3.4: Operating conditions of preliminary experiments conducted to analyze the effect of spouted bed configuration at minimum spouting air velocity

Table 3.5: Operating conditions of preliminary experiments conducted to analyze the effect of spouted bed configuration at superficial velocity of 0.89 m/s

Configuration	Air velocity	Stagnant bed	Temperatur		
	(m/s)	height (cm)	e (°C)		
Conventional spouted bed	0.89	16	75		
*Non porous draft tube fitted spouted bed	0.89	16	75		
*Porous draft tube fitted spouted bed	0.89	16	75		
* $D_T = 0.035 \text{m}, L_H = 0.06 \text{m}, L_T = 0.26 \text{m}$					

Configuration	Air velocity	Stagnant bed	Temperature		
	(m/s)	height (cm)	(°C)		
Conventional spouted bed	0.74	16	45, 55, 65, 75		
*Non porous draft tube fitted spouted bed	0.51	16	45, 55, 65, 75		
*Porous draft tube fitted spouted bed	0.58	16	45, 55, 65, 75		
* $D_T = 0.035 m$ , $L_H = 0.06 m$ , $L_T = 0.26 m$					

Table 3.6: Operating conditions of preliminary experiments conducted to analyze the effective moisture diffusivity of black pepper in three spouted bed configuration

## 3.1.6 Major experiments with raw black pepper

According to the results of preliminary experiments with different spouted bed configurations, best configuration was selected as the conventional spouted bed. Therefore major experiments were conducted with raw black pepper in conventional spouted bed to achieve following goals of the study.

- Determination of minimum spouting velocity
- Effect of operational parameters on drying kinetics
- Examine optimum drying air temperature to achieve black pepper with maximum essential oil yield with consistent quality.
- Analyse the effect of drying air temperature on mass transfer characteristics namely effective moisture diffusivity.
- Model the heat transfer in the spouted bed dryer and validate the model using the experimental data
- Develop suitable drying model to describe the drying behaviour

### 3.1.6.1 Determination of spouting behavior of raw black pepper

Spouting behavior of raw black pepper was studied to determine the minimum spouting velocity and maximum pressure drop of the spouted bed of selected bed heights; 7 cm, 14 cm, 18 cm and 22 cm. All the experiments were conducted using ambient air.

Spouted bed contactor was filled with raw black pepper particles up to the required static bed height. Air flow rate was increased in small increments by adjusting the variable frequency drive connected to the blower. Airflow velocity and total pressure drop across the spouted bed contactor were recorded simultaneously. Air flow rate was increased gradually and air velocity and pressure drop measurements were recorded. Once the bed is in fully spouted bed condition, the airflow rate was reduced gradually and the total pressure drop was recorded. Total pressure drop is the pressure drop caused mainly due to the black pepper bed and the perforated plate fixed at the bottom of the spouted bed contactor.

Pressure drop across the perforated plate fixed at the bottom of the spouted bed contactor was measured in the absence of solid particles over a same range of air velocities. The true pressure drop across the bed of particles was determined by subtracting the measured empty bed pressure drop from total pressure drop value [13].

Then the minimum spouting velocity at the selected bed heights of raw black pepper was determined using the plots of pressure drop versus air velocity at ambient temperature.

# 3.1.6.2 Effect of operational parameters on drying kinetics

The dryer was operated at hot air temperatures between 45 °C to 75 °C and stagnant bed heights of 7-20 cm of raw black pepper. Air velocity was maintained at 1.60 m/s, 1.97 m/s and 2.37 m/s. During drying process, samples were collected in selected time intervals and moisture content was measured using the direct method as

described in section 3.1.3. The drying experiments were conducted until the moisture of the black pepper reached to less than 15%.

Experiments were conducted as described below to determine the effect of drying temperature on drying kinetics. Raw black pepper samples were dried in a spouted bed dryer at 4 different temperatures; 45 °C, 55 °C, 65 °C and 75 °C keeping other parameters constant. The initial stagnant bed height was 14 cm and the airflow rate was maintained at 2.37 m/s.

Effect of air velocity on drying kinetics was determined by drying 14 cm initial stagnant bed height of raw black pepper samples in the spouted bed dryer at three different air velocities; 1.60 m/s, 1.97 m/s and 2.37 m/s keeping drying temperature at 65  $^{\circ}$ C.

Initial static bed height was varied as 7 cm, 14 cm and 20 cm keeping drying temperature at  $65^{\circ}$ C and air velocity at 2.37 m/s to determine the effect of initial stagnant bed height on drying.

Airflow with superficial velocity of 2.37 m/s was used for the experiments conducted to analyze the effect of drying temperature and bed height. Air velocity of 2.37 m/s was selected to have stable spouting operation with all possible bed heights. Since 2.37 m/s was below the terminal velocity of dried black pepper particle and it was also above the minimum spouting velocity of maximum spoutable bed height (25 cm) of the spouted bed column used for the current investigation. All the experiments were duplicated. Terminal velocity of black pepper is 9.1 m/s and it was calculated from the equation available in Appendix H.

## **3.1.6.3 Drying experiments for modeling of heat transfer**

The dryer was operated at hot air temperatures between 45 °C to 75 °C and stagnant bed heights of 7-24 cm of raw black pepper. Air velocity was maintained from 1.59 m/s to 2.60 m/s where all the experiments were conducted beyond the minimum spouting condition and also below the entrainment velocity of black pepper particles respective to their stagnant bed height. Superficial air velocity and inlet gas temperature were kept constant during the drying operation.
Drying experiments were conducted until the moisture of the black pepper reached to less than 15%. 48 experiments were conducted and data was gathered for estimation of heat and mass transfer coefficients.

During drying process, samples were collected in selected time intervals and moisture content was measured using the direct method [100]. Temperature and relative humidity of exhaust air of the dryer and temperature of the bed were measured to estimate the heat transfer coefficient. Due to excellent mixing and heat transfer occur in the spouted bed, gas and particle temperature profile in the spout and annulus regions are almost identical and equal to the bed average temperature [102]. Therefore, temperature of the bed was determined from the average of the temperature measurements of five sensors located inside the bed.

Gas flow rate and inlet gas temperature is kept constant throughout the experiment. Therefore, humidity of inlet gas is also constant assuming that there is no fluctuation of humidity and temperature of the ambient air.

## **3.1.6.4 Drying experiments for development of model to describe drying behavior**

The dryer was operated at hot air temperatures between 45 °C to 75 °C and stagnant bed heights of 14 - 22 cm of raw black pepper. The air velocity was maintained at 2.37 m/s since it was above the minimum spouting velocity of 22 cm stagnant bed height of raw black pepper. The selected velocity was well below the terminal velocity of black pepper. During the drying process, samples were collected in selected time intervals and moisture content was measured using the direct method. The drying experiments were conducted until the moisture of the black pepper reached 15 % dry basis.

All the experiments described in section 3.1.6.1 to 3.1.6.4 were duplicated and average of results are presented and described in Chapter 5.

## 3.1.6.5 Experiments for determination of optimum drying air Temperature

Experiments were conducted as described below to determine the effect of drying temperature in the spouted bed on quality of black pepper and to compare sun drying with the spouted bed drying.

A fresh sample of raw black pepper was dried in sun for several days to reduce the moisture content up to 15%. 14 cm stagnant bed height of raw black pepper was dried in conventional spouted bed just above the minimum spouting velocity of 1.43 m/s at different hot air temperatures; 45 °C, 55 °C, 65 °C and 75 °C. Solar dried black pepper and samples of black pepper dried in the spouted bed were used to extract essential oil.

After drying, all the black pepper samples were packed in air proof dark polythene bags to avoid the contact of oxidative agents with dried seeds and stored in a refrigerator at 3-5  $^{\circ}$ C until they were used for oil extraction. Extraction apparatus consists of a heating mantle, oil bath and 5 L round bottom flask with Clevenger type volatile oil trap and a condenser as shown in Figure 3.7 (a) and (b).



Figure 3.7: (a) Schematic diagram of extraction apparatus (b) Photograph of extraction apparatus

Dried black pepper was chopped and placed in round bottom flask of extraction apparatus with 2 L of distilled water and boiled for 3 - 4 hr. Essential oil was collected in the graduated Clevenger type volatile oil trap. Collected oil was allowed to cool and the volume of the essential oil was measured using graduated scale on the oil trap as shown in Figure 3.8 and 3.9. Oil extraction experiments were conducted for sundried and spouted bed dried black pepper. All the experiments were replicated four times.



Figure 3.8: Clevenger type oil trap



Figure 3.9: (a) Clevenger type oil trap in experiment set up (b) interphase of oil and water in oil trap

The oil samples were subsequently analysed using Gas Chromatography Mass spectrometry(Agilent, American 7890A/5975C GC-MS system) equipped with a fused silica capillary Agilent Technology 5% phenyl dimethyl siloxane column. The injector temperature was 25 °C, and the oscillatory temperature was 100 °C. A volume of  $2\mu$ l of oil was injected in split mode (split ratio of 1:100). The initial temperature was kept at 70 °C for 2 min, and the temperature was gradually increased to 270 °C at a heating rate of 5 °C/min. Figure 8 in Appendix III shows an image of Agilent, American 7890A/5975C GC-MS system.

Components in essential oil were identified according to the NIST (National Institute of Standards and Technology). Microsoft Excel 2010 and IBM SPSS Statistics 20 (IBM, Armonk, US) were used for the analysis of volatile oil components of essential oil derived from black pepper. In order to investigate the significance of the drying temperature on quality and oil yield of black pepper, analysis of variance (ANOVA) was conducted at 95% confidence level.

## 3.1.7 Summary of experimental conditions

Drying conditions used in major experiments were summarized in Table 3.5.

Objective of	Air velocity	Stagnant	Temperature
Experiments*	(m/s)	bed height	(°C)
		( <b>cm</b> )	
Determination of minimum		7, 14, 18,	ambient
spouting velocity		22	temperature
Effect of drying	2.37	14	45, 55, 65, 75,
temperature- extraction of			sun drying
essential oil			
analyzing diffusivity	2.37	14	45, 55, 65, 75
Effect of bed height	2.37	7, 14, 20	65
Effect of air velocity	1.60, 1.97,	14	65
	2.37		
Drying kinetics	1.60, 1.97,	7, 14, 18, 20	45, 55, 65, 75
	2.37		
Thin layer modeling	2.37	14, 18, 20,	45, 55, 65, 75
		22	
Heat and mass transfer	1.6-2.37	7-22	35-75
analysis			

Table 37.	The conditions	used in dry	ing experiment	s with conv	entional spor	uted bed
1 able 5.7.	. The conditions	s useu ili uly	ing experiment	s with conv	entional spor	uteu beu

\*Material: Raw black pepper ; Configuration: conventional spouted bed; Initial moisture content: 350 d.b%

## **3.2 Model for heat transfer**

## 3.2.1 Determination of heat transfer coefficient

The model used to define heat transfer coefficient is derived by applying energy balance to the spouted bed dryer. The main assumptions adopted in the model are as followings.

- Black pepper particles are assumed to be spherical, isotropic (uniformity in all orientations) and identical.
- Physical properties of the dry matter remain constant with time.
- Particles are well mixed in the spouted bed and each particle remove same amount of moisture with time and receive same heat flux from hot gas with time.
- Dryer is well insulated.
- Heat is flown from hot gas to particles only. Conduction of heat and moisture from particle to particle is negligible.
- Air that leaves the dryer is not in thermal equilibrium with the solid particles.
- Particle shrinkage is negligible.
- There is no accumulation of air inside the bed.
- Latent heat of moisture is assumed as constant throughout the process.

Drying rate at time t, N(t) of spouted bed dryer is expressed as kg of moisture evaporated per unit area of bed cross section per unit time.

$$N(t) = \frac{W_b\left(-\frac{dX}{dt}\right)}{S_B} \tag{3.1}$$

where

N(t) - Drying rate at time t (kg of moisture evaporated /m<sup>2</sup>s)

W<sub>b</sub> - mass of dry solid (kg)

 $S_B$  - cross section of the bed measured at right angles to the direction of air flow (m<sup>2</sup>)

The general energy balance can be written as Equation 3.2 for either closed or open system between two instant of time since energy cannot be destroyed or generated [103].

 $Energy \ accumulation = final \ energy \ of \ system- \ initial \ energy \ of \ system \qquad (3.3)$ 

Spouted bed drying process described in this work is a semi batch process where drying material does not cross the boundary but hot gas stream enters into the dryer and exits continuously. Figure 3.10 shows typical schematic diagram for unsteady state spouted bed drying process.



Figure 3. 10: Schematic diagram for unsteady state spouted bed drying process

Where

 $G'_{air}$ - dry air flow rate (kg of dry air/s)

 $T_{gi}$ - inlet gas temperature (°C)

 $T_{go}(t)$  – outlet gas temperature at time t (°C)

W<sub>b</sub>- mass of dry solid (kg)

 $\overline{X}$ - mean moisture content of the solid over the drying period (kg moisture/ kg dry solid)

 $Y_{gi}$  -humidity of air at the inlet of dryer (kg of moisture/ kg of dry air)

 $Y_{go(t)}$  -humidity of air at the outlet of the dryer at time t (kg of moisture/ kg dry air)

When considering right hand side of the Equation 3.2 for unsteady state spouted bed dryer, only the hot gas stream crosses the boundaries. Therefore right hand side (RHS) may be written using the net energy given by hot air and the energy transferred out by the system during moisture evaporation.

Energy balance was applied as follows to obtain an expression for heat transfer coefficients with combination of Equation 3.2 and Equation  $3.3_{\overline{r}}$ 

$$\left(W_b C_{ps} + W_b \bar{X} C_{pw}\right) \frac{dT_{b(t)}}{dt} = \eta G'_{air} c_{pav} \left(T_{gi} - T_{go(t)}\right) - W_b \lambda \left(-\frac{dX(t)}{dt}\right)$$
(3.4)

 $(W_b C_{ps} + W_b \bar{X} C_{pw}) \frac{dT_{b(t)}}{dt}$  expresses the thermal energy required to raise the temperature of bed.  $W_b \lambda \left(-\frac{dX(t)}{dt}\right)$  expresses the thermal energy required to vaporise the moisture from bed of particle.  $\eta G'_{air} c_{pav} (T_{gi} - T_{go(t)})$  expresses the thermal energy given for the drying process by hot air stream.

$$(W_b C_{psw}) \frac{dT_{b(t)}}{dt} + W_b \lambda \left( -\frac{dX(t)}{dt} \right) = \eta G'_{air} c_{pav} \left( T_{gi} - T_{go(t)} \right)$$

$$(3.5)$$

$$C_{pav} = C_{pa} + \bar{Y} C_{pv}$$

$$(3.6)$$

$$C_{psw} = C_{ps} + \bar{X}C_{pw} \tag{3.7}$$

## Where

 $C_{pa}$ - specific heat of dry air (J/kg°C)

 $C_{\text{pav}}\text{-}$  heat capacity of 1kg of dry air and associated water vapour (J/kg $^{\circ}\text{C}\text{)}$ 

 $C_{ps}$  - Specific heat of dry solid, (J/kg°C)

 $C_{psw}$ - Specific heat of wet solid, (J/kg of dry solid °C)

 $C_{pv}$  -specific heat of vapour, (J/kg °C)

 $C_{pw}$  -specific heat of water in solid, (J/kg °C)

 $\overline{Y}$ - mean air humidity, (kg moisture/ kg dry air)

H - Particle to air heat transfer efficiency

 $T_{b(t)}$  - mean temperature of the bed (°C)

X - moisture content of solid( dry basis) (kg of moisture/ kg of dry solid)

 $\lambda$ - Latent heat of water (J/kg )

If the thermal equilibrium between exit gas and bed solid is not achieved, it is convenient to use a gas to particle heat transfer coefficient [13]. Then heat transferred by hot gas is shown in Equation 3.8.

 $\eta = 1$ ; since no heat losses

$$G'_{air}c_{pav}(T_{gi} - T_{go(t)}) = h_p S_p \Delta T_{Lm}$$
(3.8)

Heat transfer area  $S_p$  is the surface area of the particles and  $\Delta T_{Lm}$  is the logarithmic mean temperature difference defined in Eq. (3.9).

$$\Delta T_{Lm} = \frac{(T_{gi} - T_{b(t)}) - (T_{go(t)} - T_{b(t)})}{\ln \frac{(T_{gi} - T_{b(t)})}{(T_{go(t)} - T_{b(t)})}}$$
(3.9)

$$\left(W_b C_{psw}\right) \frac{dT_{b(t)}}{dt} + W_b \lambda \left(-\frac{dX(t)}{dt}\right) = h_p S_p \Delta T_{Lm}$$
(3.10)

Where

 $h_p$  - overall heat transfer coefficient (W/m<sup>2</sup>K)

 $S_p$  - interfacial surface area of particles where heat transfer (m<sup>2</sup>)

Sensible heat gain is neglected as  $(W_b C_{psw}) \frac{dT_{b(t)}}{dt} \ll W_b \lambda \left(-\frac{dX(t)}{dt}\right)$  (3.11)

$$W_b \lambda \left( -\frac{dX(t)}{dt} \right) = h_p S_p \Delta T_{Lm}$$
(3.12)

$$h_p = \frac{-W_b \lambda \int_{X_1}^{X_2} dX}{S_p \int_{t=0}^{t=t} \Delta T_{Lm} dt}$$
(3.13)

Heat transfer coefficient (h<sub>p</sub>) were obtained by solving Equation 3.13 using graphical method for particular time period. Observed data of logarithmic mean temperature difference of the bed ( $\Delta T_{Lm}$ ) vs time should be plotted and the area under the curve gave the value of the integral in the denominator of Equation 3.13. Nevertheless area under the curve of logarithmic mean temperature difference ( $\Delta T_{Lm}$ ) vs time should be plotted and the area under the curve gave the value of the integral in the denominator of Equation 3.13. Nevertheless area under the curve of logarithmic mean temperature difference ( $\Delta T_{Lm}$ ) vs time evaluated directly using the MATLAB codes.

### **3.2.2 Dimensional analysis of heat transfer process**

Prior to dimensional analysis of the heat transfer process, fundamental dimensions should be identified. Fundamental dimensions which is used to express momentum transfer are length, time and mass. They are symbolized as L, t and M respectively. Dimensional analysis of energy problems requires two more fundamental dimensions, namely heat and temperature and symbolized as Q and T [104].

Considering the heat transfer process in drying of black pepper in spouted bed dryer, the important variables, their symbols and dimensions are listed in Table 3.6

Table 3.8: Important variables for heat transfer of black pepper in the spouted bed dryer

	Variable	Symbol	Dimensions
1	Air velocity	U	Lt <sup>-1</sup>
2	Static bed height	Н	L
3	Particle diameter	d <sub>p</sub>	L
4	Gas viscosity	$\mu_{g}$	$ML^{-1}t^{-1}$
5	Air thermal conductivity	kg	$QL^{-1}t^{-1}T^{-1}$
6	Air inlet temperature	T <sub>gi</sub>	Т
7	Temperature difference of gas inlet and gas inlet wet bulb temperature	T <sub>gi</sub> -T <sub>gi</sub> ,w	Т
8	air density	ρ	ML <sup>-3</sup>
9	Heat transfer coefficient-fluid to particle	h <sub>p</sub>	$QL^{-2}t^{-1}T^{-1}$

Where

L, M, Q, T, t - Dimensions of length, mass, heat, temperature and time respectively

Then dimensional matrix for heat transfer processes is formed by tabulating exponents of the fundamental dimensions. Dimensional analysis was carried out according to Buckingham pi theorem and important dimensionless numbers were identified [104]. A Possible correlation was developed using IBM SPSS Statistics 20 by non-linear multiple regression. Results are presented and described in Chapter 7.

## 3.3 Drying model development

Experiments related to accomplish the objective of developing a suitable model to describe drying behaviour is described in section 3.1.6.4. Gathered data of moisture content vs time of the black pepper during the drying process were used to model the drying behavior of wetted and raw black pepper.

## 3.3.1 Mathematical modeling

Moisture content was normalized to dimensionless parameter; moisture ratio (MR).

Moisture ratio is calculated as follows:

$$MR = (X_i - X_e) / (X_o - X_e)$$
(3.14)

 $X_{e,} X_{i}$  and  $X_{o}$  are equilibrium moisture content,  $i^{th}$  experimental moisture content and initial moisture content respectively.

Moisture ratio vs time were used for model development. They were fitted to five thin layer drying models namely Newton model, Page model, Henderson and Pabis model, Two compartment model and Logarithmic model which were recalled in Table 3.7. These models are the basic models among large number of thin layer models and they are widely used by many researchers.

Model name	Model	Equati on no
Newton	$MR = \frac{X_t - X_e}{X_0 - X_e} = e^{-kt}$	2.34
Page	$MR = \frac{X_t - X_e}{X_0 - X_e} = e^{-kt^n}$	2.35
Henderson and Pabis	$MR = \frac{X_t - X_e}{X_0 - X_e} = ae^{-kt}$	2.36
Logarithmic	$MR = \frac{X_t - X_e}{X_0 - X_e} = ae^{-kt} + b$	2.37
Two compartment	$MR = \frac{X_t - X_e}{X_0 - X_e} = ae^{-k_1 t} + be^{-k_2 t}$	2. 38

Table 3.9: Selected thin layer drying models used for analysis

a, b, k, k<sub>1</sub>, k<sub>2</sub> and n are drying coefficients and constants

The moisture ratio is simplified to  $X_t/X_0$  neglecting  $X_e$  term since  $X_e$  term, is relatively small compared to  $X_t$  and  $X_0$  and the continuous fluctuation of the relative humidity of drying air leading to change of equilibrium moisture content of the particle being dried [19, 76]. The software Matlab 2011 was used to facilitate the model fitting.

The performance of these models were compared using three statistical parameters coefficient of determination ( $R^2$ ), Root mean square error (RMSE) and sum of square error (SSE). The statistical values were evaluated using Equation 3.20 to Equation 3.22 [75].

$$SSE = \sum_{i=1}^{n} \left( MR_{pre,i} - MR_{exp,i} \right)^{2}$$
(3.15)

$$RMSE = \sqrt{\frac{\sum_{i=1}^{n} (MR_{pre,i} - MR_{exp,i})^{2}}{N}}$$
(3.16)

$$R^{2} = 1 - \frac{\sum_{i=1}^{n} (MR_{pre,i} - MR_{exp,i})^{2}}{\sum_{i=1}^{n} (MR_{exp,i} - MR_{exp_{mean}})^{2}}$$
(3.17)

Where

MR<sub>exp,I</sub> - i<sup>th</sup> experimental moisture ratio

MR<sub>exp,mean</sub> - mean of experimental moisture ratios

MR<sub>pre,I</sub> - i<sup>th</sup> predicted moisture ratio

N - Number of observation

According to the statistical analysis, best thin layer model which can describe the drying behaviour of black pepper in spouted bed dryer was selected. Since developed models can be used only for specified drying conditions, four models were developed to describe the drying behaviour of raw black pepper in conventional spouted bed. Drying constants and coefficients of best selected model were correlated to bed height at specified drying temperatures using the regression analysis.

## **Chapter specific nomenclature**

a, b, k, k<sub>1</sub>, k<sub>2</sub>, n - drying coefficients and constants

 $C_{pa}$  - specific heat of dry air (J/kg°C)

 $C_{pav}$ - heat capacity of 1kg of dry air and associated water vapour (J/kg°C)

 $C_{ps}$  - specific heat of dry solid, (J/kg<sup>°</sup>C)

 $C_{psw}$  - Specific heat of wet solid, (J/kg of dry solid  $^{\circ}C)$ 

 $C_{pv}$  - specific heat of vapour, (J/kg °C)

 $C_{pw}$  - specific heat of water in solid, (J/kg °C)

D<sub>c</sub> - diameter of the column (m)

D<sub>i</sub>- gas inlet diameter (m)

D<sub>o</sub> - cone base diameter (m)

d<sub>p</sub> - particle diameter (m)

D<sub>T</sub> -diameter of draft tube (m)

 $G'_{air}$ - dry air flow rate (kg of dry air/s)

H, H<sub>o</sub>- static bed height (m)

H<sub>l</sub> - height of cylindrical column (m)

 $h_p$  - overall heat transfer coefficient fluid to particle (W/m<sup>2</sup>K)

H<sub>T</sub> - tube height (m)

k<sub>g</sub> -Air thermal conductivity (J/msK)

L, M, Q, T, t - Dimensions of length, mass, heat, temperature and time respectively

L<sub>H</sub> -entrainment height (m)

L<sub>T</sub> - total height of the draft tube

MR<sub>exp,i</sub>- i<sup>th</sup> experimental moisture ratio

MR<sub>exp,mean</sub> - mean of experimental moisture ratios

 $MR_{pre,I}$  -  $i^{th}$  predicted moisture ratio

N- number of observation

N(t)-Drying rate at time t (kg of moisture evaporated  $/m^2s$ )

 $S_B$  - cross section of the bed measured at right angles to the direction of air flow  $(m^2)$ 

 $S_p$  - interfacial surface area of particles where heat transfer (m<sup>2</sup>)

 $T_{b(t)}$  - mean temperature of the bed (°C)

 $T_{gi}$ ,  $T_d$ - drying air temperature or inlet gas temperature (°C)

 $T_{gi,w}$ - gas inlet wet bulb temperature (°C)

 $T_{go}(t)$  - outlet gas temperature at time t (°C)

U - Superficial air velocity (m/s)

W<sub>b</sub> - mass of dry solid (kg)

W<sub>d</sub> -weight of sample after drying in the oven (kg)

W<sub>w</sub> - initial weight of the sample (kg)

X- moisture content of solid( dry basis) (kg of moisture/ kg of dry solid)

X<sub>e</sub>, -equilibrium moisture content

X -moisture content dry basis (%)

X<sub>o</sub> -initial moisture content

 $X_t \mbox{ -moisture content}$  at time t

 $\overline{X}$  - mean moisture content of the solid over the drying period (kg moisture/ kg dry solid)

 $Y_{gi}$  - humidity of air at the inlet of dryer (kg of moisture/ kg of dry air)

 $Y_{go(t)}$  - humidity of air at the outlet of the dryer at time t (kg of moisture/ kg dry air)

 $\overline{Y}$  - Mean air humidity, (kg moisture/ kg dry air)

 $\mu_q$  - Viscosity of air (kg/ms)

 $\Delta T_{Lm}$  - Logarithmic mean temperature difference

 $\eta$  - Particle to air heat transfer efficiency

 $\lambda$  - Latent heat of water (J/kg )

 $\rho$  - density of fluid/air (kg/m<sup>3</sup>)

## Abbreviations

ANOVA - Analysis of variance

R<sup>2</sup> - Coefficient of determination

RMSE - Root mean square error

SSE - Sum of square error

VSD - variable speed drive

## 4. RESULTS AND DISCUSSION ON PRELIMINARY EXPERIMENTS

This chapter describes the outcome of preliminary experiments conducted with wetted black pepper. Preliminary experiments were conducted using three configurations of spouted bed dryer namely, conventional spouted bed (CSB), and spouted bed equipped with non-porous draft tube (SBNPDT) and spouted bed equipped with porous draft tube (SBPDT). Minimum spouting velocity, drying kinetics and diffusivity of wetted black pepper drying in three configurations are analysed and presented. Furthermore, results of mathematical modelling of wetted black pepper drying in above-mentioned three configurations are presented and discussed.

## 4.1 Minimum spouting velocity

Prior to conducting the drying experiments, minimum spouting velocities of wetted black pepper for two bed loadings in three configurations of spouted bed were determined. Plots of pressure drop versus air velocity were used to determine minimum spouting velocity. As indicated in Table 4.1, spouted bed equipped with non-porous draft tube (SBNPDT) has the lowest minimum spouting velocity followed by the spouted bed equipped with porous draft tube (SBPDT) for a particular bed height.

In the spouted bed equipped with non-porous draft tube, air stream is not percolated in to the annulus region from the spout region as in other two configurations. Therefore, energy required to disrupt the packing is achieved by low airflow rate compare to other two configurations. Conventional spouted bed has the highest minimum spouting velocity. In all cases, minimum spouting velocity increases with increasing stagnant bed heights due to increase of energy required to disrupt the packing. Altzibar et al. also stated that the conical spouted bed equipped with nonporous draft tube had low minimum spouting air velocity compared to the conical spouted bed without draft tube. In addition, minimum spouting velocity was approximately proportional to the stagnant bed of cylindrical spouted beds [46]. More information on determination of minimum spouting velocity from plot of pressure drop versus air velocity is given in section 5.1 for experiments conducted with raw black pepper.

Configuration	Stagnant bed	Minimum spouting	
Comiguration	height; H (m)	velocity; U <sub>ms</sub> (m/s)	
CSB	0.16	0.74	
	0.10	0.50	
SBPDT <sup>b</sup>	0.16	0.58	
	0.10	0.44	
SBNPDT <sup>b</sup>	0.16	0.51	
	0.10	0.43	
<sup>b</sup> $D_T = 0.035 m$ , $L_H = 0.06 m$ , $L_T = 0.26 m$			

Table 4.1: Minimum spouting velocity

## 4.2 Effect of draft tube configurations on drying

Drying experiments were conducted to identify the effect of draft tube configuration on drying. Spouted bed equipped with different dimensions of non-porous and porous draft tubes were used. Geometrical parameters of draft tubes such as diameter  $(D_T)$ , entrainment height  $(L_H)$  and tube height  $(H_T)$  were changed. Drying experiments were conducted until black pepper attained 15% dry basis moisture content from their initial moisture content of 70-80 %. Drying time is defined as the time taken to dry black pepper from initial moisture content to 15 % moisture content.

Use of higher air flow rate is unproductive for spouted bed setting and considering the cost and energy involved, many researches have been conducted at minimum spouting conditions to ensure the regular movement of particles in the three hydrodynamic regions of the bed; spout, annulus and fountain [31]. Therefore, all the drying experiments were conducted using airflow rates just above the minimum spouting conditions. Results of experiments conducted to analyze the effect air velocity on drying kinetics will be presented on section 4.3.

## 4.2.1 Effect of diameter of draft tube

Seven non-porous draft tubes and nine porous draft tubes with different dimensions were used for the analysis. Dimensions of those draft tubes are presented in Table 3.2 and Table 3.3 in Chapter 3.

Three porous draft tubes with diameter of 0.05 m, 0.06 m and 0.07 m and nonporous draft tubes of diameter 0.029 m, 0.035 m and 0.048 m were used. The draft tube diameter is usually chosen to be similar or larger than the diameter of gas inlet nozzle to achieve stable spouting [37]. Therefore gas inlet diameter of spouted bed equipped with non-porous draft tube was set as 0.029 m and a porous draft tube was set as 0.05 m for this study.

Effect of diameter of draft tube on drying kinetics of black pepper was analysed and drying kinetics data are shown in Figure 4.1 to Figure 4.4. When draft tube diameter is increased keeping other parameters (entrainment height and tube height) constant, drying time was decreased and rate of moisture removal was increased. The amount of pepper particles that can enter into the spout region is higher in larger diameter draft tubes than those of smaller diameter draft tubes, and hence intensive heating takes place inside the spout. Therefore drying time decreases.



Figure 4.1: Effect of diameter of non-porous draft tube on drying of wetted black pepper under minimum spouting conditions in SBNPDT;  $T_d = 55$  °C,  $H_o=0.16$  m, Di = 0.029 m,  $L_H = 0.03$  m,  $H_T = 0.16$  m



Figure 4.2: Rate of change of moisture content vs. moisture content; of diameter of non- porous draft tube on drying of wetted black pepper under minimum spouting conditions in SBNPDT.  $T_d = 55$  °C, H =0.16 m, D<sub>i</sub>=0.029 m, L<sub>H</sub> = 0.03 m, H<sub>T</sub>= 0.16 m



Figure 4.3: Effect of diameter of porous draft tube on drying of wetted black pepper under minimum spouting conditions in SBPDT;  $T_d = 55$  °C, H = 0.16 m,  $D_i = 0.05$  m, ,  $L_H = 0.03$  m,  $H_T = 0.16$  m



Figure 4.4: Rate of change of moisture content vs. moisture content; Effect of diameter of porous draft tube on drying of wetted black pepper under minimum spouting conditions in SBPDT;  $T_d = 55$  °C, H = 0.16 m,  $D_i = 0.05$  m,  $L_H = 0.03$  m,  $H_T = 0.16$  m

#### 4.2.2 Effect of entrainment height of draft tube

Three non-porous draft tubes and three porous draft tubes of same diameter and tube height having different values entrainment heights were used to analyse the effect of entrainment height on drying kinetics of black pepper. Figure 4.5 to Figure 4.8 illustrate the drying curves for different entrainment heights of both draft tubes. Increase in entrainment height significantly affected in the moisture removal rate in both systems. For spouted bed equipped with non-porous draft tube, lowest drying time of 110 min was recorded for the black pepper dried at 0.08 m entrainment height at the combination of 75 °C drying air temperature and 0.16 m of non-porous draft tube height. However, for the same geometric configurations when the entrainment height was 0.06 m and 0.03m drying time were greater than 150 min.

The reason for the highest drying rate in the case of 0.08 m entrainment height is due to the fact that when the entrainment height is larger, more number of particles can enter through the draft tube to spout region and hence the spout region in which most of the heat and mass (moisture) transfer occurs [105].



Figure 4.5: Effect of entrainment height of non- porous draft tube on drying of wetted black pepper under minimum spouting conditions in SBNPDT;  $T_d = 75$  °C, H =0.16 m, D<sub>i</sub> =0.035 m, H<sub>T</sub>= 0.16 m, D<sub>T</sub>= 0.035 m



Figure 4.6: Rate of change of moisture content vs. moisture content. Effect of entrainment height of non- porous draft tube on drying of wetted black pepper under minimum spouting conditions in SBNPDT;  $T_d = 75^{\circ}C$ , H = 0.16 m,  $D_i = 0.035$  m,  $H_T = 0.16$  m,  $D_T = 0.035$  m



Figure 4.7: Effect of entrainment height of porous draft tube on drying of black pepper under minimum spouting conditions in SBPDT;  $T_d = 75$  °C, H=0.16 m, D<sub>i</sub> =0.05m, H<sub>T</sub>= 0.16 m, D<sub>T</sub>= 0.035m



Figure 4.8: Rate of change of moisture content vs. moisture content. Effect of entrainment height of porous draft tube on drying of wetted black pepper under minimum spouting conditions in SBPDT;  $T_d=75$  °C, H=0.16 m, D<sub>i</sub> =0.05 m, H<sub>T</sub>= 0.16 m, D<sub>T</sub>= 0.035 m

## 4.2.3 Effect of height of draft tube

Effect of height of draft tubes on drying kinetics is shown in Figure 4.9 to Figure 4.12. A drying time of 150 min was noticed when the draft tube height was kept 0.16 m, and whereas it was 100 min at a draft tube height of 0.24 m at 75 °C air temperature and entrainment height of 0.06 m at the minimum spouting conditions for non- porous draft tube fitted spouted bed. The drying rate was also increased with increasing draft tube height under investigation. In case of 0.24 m height draft tube, the rate of moisture removal was faster than those of other two due to longer exposure of particles to the heated air in the spout region [106]. Similar effect is also shown for porous draft tube fitted spouse bed. The drying rate was also increased with increasing porous draft tube fitted spouted bed. The drying rate was also increased with increasing porous draft tube fitted spouted bed. The drying rate was also increased with increasing porous draft tube fitted spouted bed.



Figure 4.9: Effect of tube height of non- porous draft tube o drying of wetted black pepper under minimum spouting conditions in SBNPDT;  $T_d = 75 \degree C$ , H = 0.16 m,  $D_i = 0.035$  m,  $L_H = 0.06$  m,  $D_T = 0.035$  m



Figure 4.10: Rate of change of moisture content vs. moisture content Effect of tube height of non- porous draft tube on drying of wetted black pepper under minimum spouting conditions in SBNPDT;  $T_d = 75$  °C, H=0.16 m, D<sub>i</sub> =0.035 m, L<sub>H</sub>= 0.06 m, D<sub>T</sub>= 0.035 m



Figure 4.11: Effect of tube height of porous draft tube on drying of black pepper under minimum spouting conditions in;  $T_d$ = 75 °C, H=0.16 m, D<sub>i</sub>=0.05 m, L<sub>H</sub>= 0.06 m, D<sub>T</sub>= 0.035 m



Figure 4.12: Rate of change of moisture content vs moisture content Effect of tube height of non- porous draft tube on drying of wetted black pepper under minimum spouting conditions in SBPDT;  $T_d = 75$  °C, H=0.16 m, D<sub>i</sub>=0.05 m, L<sub>H</sub>= 0.06 m, D<sub>T</sub>= 0.035 m

As a summary, analysis of Figure 4.1 to Figure 4.12 shows that the drying rate of black pepper is influenced by the geometric factors of both type of draft tubes such as diameter, entrainment height and height of the draft tube.

#### 4.3 Effect of air velocity on drying kinetics for spouted bed configurations

Effect of air velocity on drying kinetics of wetted black pepper in three spouted bed configurations was analysed. Experiments were conducted at two velocities namely, minimum spouting velocity ( $U_{ms}$ ) of each configuration and 0.89 m/s ( $v_1$ ) at 75 °C drying air temperature for 16 cm of static bed height. Latter is quite higher velocity where particles will not transport pneumatically away from the drying chamber. Drying time of respective experiments, percentage of increase in air velocity and percentage of time reduction due to use of higher air flow rate are shown in Table 4.2. Percentage of time reduction and percentage of increase in air velocity were calculated using Equation 4.1 and 4.2.

% of time reduction = 
$$\frac{t_{U_{ms}} - t_{\nu_1}}{t_{U_{ms}}} \times 100$$
 (4.1)

% of increase in air velcity  $= \frac{v_1 - U_{ms}}{U_{ms}} \times 100$  (4.2)

Where

 $t_{U_{ms}}$ - drying time when air velocity is set to minimum spouting velocity

 $t_{v_1}$  - drying time when air velocity is set to v<sub>1</sub>

Table 4.2: Drying time	of black pepper at	different spouted	bed configu	urations at
	different air	velocities		

Air velocity	Drying time (minutes)		% of	% of
	TT		increase in	time
Dryer Configuration	U <sub>ms</sub>	$\mathbf{v}_1$	air velocity	reduction
Conventional bed	54	49	20.27	9.26
Spouted bed with porous draft tube	100	92	53.45	8.00
Spouted bed with non- porous draft tube	105	100	74.51	4.76

High air velocity decreases the drying time of black pepper as shown in Table 4.2, Figure 4.13 and Figure 4.14 with increasing drying rate. This is due to the fact that at high air velocity or air flow rate, solid circulation rate is high. Therefore particles get more exposure to hot air in spout region than that of particles in spouted bed of low velocity. Circulation rate of particle is the number of cycles where a particle moves through spout, fountain and annulus region per unit time. However, percentage of time reduction is below 10 % for the three configurations. Therefore using high velocity air flow is not a good solution considering the energy and cost involved. Figure 4.13 and Figure 4.14 show the change of moisture ratio and moisture content of black pepper with time respectively while experiments were going on.

## 4.4 Effect of spouted bed configurations on drying kinetics

Effect of configuration of spouted bed on drying kinetics of wetted black pepper were analysed at minimum spouting condition for three different drying temperatures and results are shown in Figure 4.15 and Figure 4. 16. Installation of both types of draft tubes in the spouted bed dryer reduces the minimum spouting velocity of black pepper compared to that of conventional spouted bed dryer as shown in Table 4.1. However, considering the drying process, non-porous draft tube fitted system requires longer drying time and shows low drying rate due to gas by passing through the spout without percolating into the annulus region. All the drying rate curves are characterized by falling rate periods. According to Figure 4. 16, conventional spouted bed experiences the highest drying rate followed by porous dratt tube fitted system while non-porous draft tube fitted spouted bed has the lowest drying rate. Solid particles are less exposed to the hot air in the annular region of the spouted bed equipped with draft tubes and hence show lower drying rates compared to the conventional spouted bed dryers [107].

Drying air temperature	Drying time (minutes)		utes)
	55°C	65°C	75°C
Dryer configuration			
Conventional bed	100	75	54
Spouted bed with porous draft tube	230	160	100
Spouted bed with non- porous draft tube	310	170	105

# Table 4.3: Drying time of spouted bed configurations at different drying temperatures



Figure 4.13: Effect of air velocity on drying kinetics of black pepper when dried in CSB, SBPDT and SBNPDT. Experiment conditions:  $T_d = 75 \degree C$ ,  $D_o = 0.035$  m, H = 0.16 m,  $D_T = 0.035$  m,  $L_H = 0.06$  m,  $L_T = 0.26$  m



Figure 4.14: Effect of air velocity on drying kinetics of black pepper when dried in CSB, SBPDT and SBNPDT. Experiment conditions:  $T_d = 75$  °C,  $D_o = 0.035$  m, H = 0.16 m,  $D_T = 0.035$  m,  $L_H = 0.06$  m,  $L_T = 0.26$  m



Figure 4.15: Effect of spouted bed configuration on drying kinetics of black pepper when dried in CSB, SBPDT and SBNPDT at minimum spouting conditions. Experiment conditions:  $D_o = 0.035$  m, H = 0.16 m,  $D_T = 0.035$  m,  $L_H = 0.06$  m,  $L_T = 0.26$  m



Figure 4. 16: Rate change of moisture content vs. moisture; Effect of spouted bed configuration on drying kinetics of black pepper when dried in CSB,SBPDT and SBNPDT at minimum spouting conditions; Experiment conditions:  $D_o = 0.035$  m, H = 0.16 m,  $D_T = 0.035$  m,  $L_H = 0.06$  m,  $L_T = 0.26$  m

## 4.5 Diffusivity of wetted black pepper for three spouted bed configurations

Initial static bed height of 16 cm wetted black pepper were dried under minimum spouting conditions in three spouted bed configurations as mentioned in Chapter 3. Experiments were conducted at 4 different drying air temperatures: 45 °C, 55 °C, 65 °C, and 75 °C. During experiments, black pepper samples were collected in selected time intervals for determination of moisture content. Natural logarithm of moisture ratio (Ln (MR)) was plotted against time (t) for each drying temperatures as shown in Figure 4.17 to Figure 4.19.



Figure 4.17: Ln (Moisture ratio) vs. drying time of wetted pepper dried in conventional spouted bed dryer Experiment conditions: H = 0.16m,  $D_i = 0.035 m$ 



Figure 4.18: Ln (Moisture ratio) vs. drying time of raw pepper dried in porous draft tube fitted spouted bed dryer; Experiment conditions: H = 0.16 m,  $D_i = 0.035$  m,  $L_H = 0.06$  m,  $D_T = 0.035$ m,  $H_T = 0.02$  m



Figure 4.19: Ln (Moisture ratio) vs. drying time of wetted pepper dried in nonporous draft tube fitted spouted bed dryer; Experiment condition: H = 0.16 m,  $D_i = 0.035$  m,  $L_H = 0.06$  m,  $D_T = 0.035$  m,  $H_T = 0.02$  m

Effective diffusivity ( $D_{eff}$ ) of black pepper at different drying temperatures was calculated using the plot of Ln (MR) vs. time using Equation 2.5 as mentioned in section 2.13. As per Table 4.4 to Table 4. 6, effective diffusivity of black pepper was increased with increasing drying air temperatures. As stated in Chapter 5, drying rate or moisture removal rate is increased with increasing drying air temperature, drying air temperatures. This is due to the fact that at high drying temperature, drying air provides high external heat fluxes leading to higher moisture diffusivities and hence higher drying rate [108]. Effective diffusivity values were ranged from  $5.64 \times 10^{-11}$  to  $2.05 \times 10^{-10}$  for above-mentioned drying air temperatures in three spouted bed configurations.

Drying air temperature T <sub>d</sub> (°C)	Effective diffusivity D <sub>eff</sub> (m <sup>2</sup> /s)
45	$9.59  imes 10^{-11}$
55	$1.30  imes 10^{-10}$
65	$1.57  imes 10^{-10}$
75	$2.26  imes 10^{-10}$

 Table 4.4: Effective moisture diffusivity of wetted black pepper dried in conventional spouted bed at different temperatures

 Table 4.5: Effective moisture diffusivity of wetted black pepper dried in porous draft

 tube fitted spouted bed at different temperatures

Drying air temperature T <sub>d</sub> (°C)	Effective diffusivity D <sub>eff</sub> (m <sup>2</sup> /s)
45	$6.85  imes 10^{-11}$
55	$9.59  imes 10^{-11}$
65	$1.37  imes 10^{-10}$
75	$1.71 imes10^{-10}$
Drying air	Effective diffusivity
------------------------	------------------------
temperature $T_d$ (°C)	$D_{eff}(m^2/s)$
45	$4.11 \times 10^{-11}$
55	$6.30  imes 10^{-11}$
65	$8.90  imes 10^{-11}$
75	$1.37 \times 10^{-11}$

 Table 4. 6: Effective moisture diffusivity of wetted black pepper dried in non-porous

 draft tube fitted spouted bed at different temperatures

Ln ( $D_{eff}$ ) vs. reciprocal of absolute drying air temperature ( $1/T_d$ ) given in Kelvin were plotted in Figure 4.20 to obtain the pre-exponential factor ( $D_0$ ) and activation energy ( $E_a$ ) as mentioned in Equation 2.6 using the Arrhenius type relationship. Table 4.7 listed the pre exponential factor and activation energy of wetted black pepper drying process in three spouted bed configurations.



Figure 4.20: Ln ( $D_{eff}$ ) vs. 1/T<sub>d</sub> of wetted pepper dried in three spouted bed configurations; Experiment conditions: H=0.16 m, D<sub>i</sub>=0.035 m, L<sub>H</sub>= 0.06 m, D<sub>T</sub>= 0.035 m, H<sub>T</sub>= 0.02 m

According Table 4.7, pre-exponential factor of black pepper dried in conventional spouted bed is the lowest followed by porous draft tube fitted bed while black pepper dried in non-porous draft tube fitted experienced the highest value. Activation energy refers to the minimum energy requirement to overcome the barrier for the moisture diffusion [109] . Activation energy for black pepper drying in conventional spouted bed is 9.13 kJ/mol which is the lowest of three configurations analysed followed by spouted bed with porous draft tube while non-porous draft tube fitted spouted bed has the highest activation energy of 35.21 kJ/mol.

Markowski et al. also used tempered barley for drying experiments in a spouted bed dryer at 30 °C -45 °C drying temperatures and found that the time of barley drying depend on the drying air temperature. The effective diffusivities of barley in their study were in the range of  $2.20 - 4.52 \times 10^{-11}$  m<sup>2</sup>/s [110]. Furthermore, Chapter 5 will present the information on effective diffusivity, pre-exponential factor and activation energy of raw black pepper and other agricultural products.

spouted bed configurations	Pre -exponential factor	Activation energy
	<b>Do</b> (m <sup>2</sup> /s)	E <sub>a</sub> (KJ/mol)
conventional bed	$1.41 \times 10^{-6}$	25.37
SBPDT	$3.56 \times 10^{-6}$	28.61
SBNPDT	$4.00 \times 10^{-5}$	36.39

 Table 4.7: Pre -exponential factor and activation energy of wetted black pepper dried

 in different spouted bed configurations

Equation 2.7 in chapter 2 stated the magnitude of effective diffusivity as a function of exponent ( $E_a/RT$ ). When recalling the meaning of RT, it indicates the average kinetic energy of molecules. Therefore the exponent indicates the ratio of activation energy to kinetic energy of molecules participated in the drying process. When the ratio is too large, effective diffusivity value becomes smaller due to the negative

sign. Also high temperature and low activation energy favor large effective diffusivity thus speed up the drying process [111]. Moreover these  $E_a$  and RT terms occur in an exponent; their effects on effective diffusivity are quite significant.

According to the results, effective diffusivity of black pepper not only depends on drying air temperature but also on the configuration of the spouted bed dryer.

# **4.6 Decision on selecting the optimum configuration for experiment with raw black pepper**

Although minimum spouting velocity is low, spouted bed equipped with draft tubes experiences longer drying time than that of the conventional spouted bed. Hence, it can cause degradation of volatile compounds in the black pepper during long drying periods. Furthermore, either tube plugging or agglomeration of the drying material or any other foreign substance can disrupt the operation.

Conventional spouted bed experiences higher drying rates and higher effective diffusivity values compared to both draft tube fitted configurations. Therefore, conventional spouted bed dryer was selected as the best configuration of spouted bed among draft tube fitted system to continue drying experiments with raw black pepper.

# **4.7 Mathematical modeling of drying kinetics of wetted black pepper in three spouted bed configurations**

3 kg of wetted black pepper were dried in the spouted dryer with 3 different configurations; namely conventional dryer (without draft tube), non-porous daft tube fitted dryer and porous draft tube fitted dryer. The dimensions of draft tubes used were: diameter of draft tube ( $D_T$ ), 0.035m, entrainment height ( $L_H$ ), 0.06 m and height of the draft tube,( $L_T$ ) 0.26 m.

Drying kinetic data were fitted to selected five thin layer drying models using curve fitting tool of Matlab 2011. The results of fitting experimental data to thin layer models and the summary of the statistical analyses are presented in Table 4.8 to Table 4.10. The higher value of  $R^2$  and lowest value of SSE and RMSE, the better the goodness of fit [92].

It was observed that the Logarithmic model presented the highest  $R^2$  values with lower RMSE and SSE values for spouted bed dryer provided with non- porous draft tube at 55 °C, 65 °C and 75 °C drying air temperature.

Page model, Logarithmic model and Two compartment model possessed  $R^2$  values greater than 0.99 with small values of RMSE and SSE values for conventional spouted bed dryer and the spouted bed equipped with porous draft tube at 75 °C.

Two compartment model was the best fitted model with 0.9975 of highest  $R^2$  and 0.0156 of lowest RMSE value for conventional bed at 55 °C drying temperature. Logarithmic model found best with highest  $R^2$  and lowest RMSE values for spouted bed equipped with non-porous draft tube bed at 55 °C drying temperatures while well-known Page model showed the best results with 0.9972 of  $R^2$  and 0.0134 of RMSE for spouted bed equipped with porous draft tube at 55 °C drying air temperature.

Drying constants of thin layer drying models are presented in Table 4.9. For many cases drying constants of selected thin layer drying models for spouted bed drying process are in same order of magnitude with values available in literature [86,80].

Conventional bed				
Model	Equation	$\mathbf{R}^2$	RMSE	SSE
Newton	MR = exp(-0.0206t)	0.9629	0.0508	0.0258
Page	$MR = \exp(-0.0463 t^{0.7923})$	0.9832	0.0360	0.0117
Henderson & Pabis	MR = 0.9604 exp(-0.0196t)	0.9665	0.0509	0.0233
Two	$MR = 0.0526 \exp(0.0114t) +$	0.9975	0.0156	0.0171
compartment	0.9623 exp(0.0281t)	017770	010100	010171
Logarithmic	$MR = 0.8408 \exp(-0.0336t) + 0.1792$	0.9966	0.0173	0.0024
Spouted bed with	th porous draft tube			
Model	Equation	$\mathbf{R}^2$	RMSE	SSE
Newton	$MR = \exp(-0.0086t)$	0.9516	0.0535	0.0344
Page	$MR = \exp(-0.0336t^{0.7158})$	0.9972	0.0134	0.0020
Henderson & Pabis	MR = 0.9116exp(-0.0077t)	0.9719	0.0426	0.0200
Two	$MR = 0.4282 \exp(-0.0077t)$	0.0710	0.0471	0.0200
compartment	$+ 0.4834 \exp(-0.0077t)$	0.9719	0.0471	0.0200
Logarithmic	$MR = 0.8054 \ \exp(-0.0123t) + 0.1562$	0.9871	0.0303	0.0092
Spouted bed with	th non- porous draft tube			
Model	Equation	$\mathbf{R}^2$	RMSE	SSE
Newton	$MR = \exp(-0.0060 t)$	0.9937	0.0197	0.0058
Page	$MR = \exp(-0.0093 t^{0.9138})$	0.9970	0.0141	0.0028
Henderson & Pabis	$MR = 0.9633 \exp(-0.0196t)$	0.9971	0.0138	0.0027
Two	$MR = 0.5077 \exp(-0.0046t) +$	0.0071	0.0140	0.0027
compartment	$0.4555 \exp(-0.073t)$	0.9971	0.0149	0.0027
Logarithmic	MR = 0.9396 exp(-0.0061t)	0.9973	0.0137	0.0024
	+ 0.0294	5.7775	5.0127	0.0021

Table 4.8: Results of statistical analysis for three spouted bed configurations at 55 °C drying air temperature

Conventional bed				
Model	Equation	R <sup>2</sup>	RMSE	SSE
Newton	$MR = \exp(-0.0233t)$	0.9426	0.0622	0.0386
Page	$MR = \exp(-0.0730t^{0.7018})$	0.9883	0.0295	0.0078
Henderson & Pabis	$MR = 0.9291 \exp(-0.0213t)$	0.9538	0.0588	0.0311
Two compartment	$MR = 0.1563 \exp(0.0157t) + 0.8464 \exp(-0.03843)$	0.9998	0.0043	0.0002
Logarithmic	$MR = 0.8206 \exp(-0.0399t) + 0.1831$	0.9998	0.0043	0.0002
Spouted bed wi	th porous draft tube			
Model	Equation	$\mathbf{R}^2$	RMSE	SSE
Newton	MR = exp(-0.0130t)	0.9466	0.0532	0.0425
Page	$MR = \exp(-0.0444t^{0.717})$	0.9961	0.0148	0.0031
Henderson & Pabis	$MR = 0.906 \exp(-0.0115t)$	0.9713	0.0404	0.0229
Two compartment	$MR = 0.253 \exp(-0.0082t) + 0.7512 \exp(-0.0091t)$	0.9974	0.0131	0.0021
Logarithmic	$MR = 0.7889 \exp(-0.0190t) + 0.1712$	0.9879	0.0273	0.0097
Spouted bed wi	th non- porous draft tube			
Model	Equation	$\mathbf{R}^2$	RMSE	SSE
Newton	$MR = \exp(-0.0109t)$	0.9906	0.0247	0.0085
Page	$MR = \exp(-0.0098t^{1.025})$	0.9908	0.0253	0.0083
Henderson & Pabis	$MR = 1.019 \exp(-0.0112t)$	0.9915	0.0243	0.0077
Two compartment	$MR = 0.5096 \exp(-0.0112 t) + 0.5096 \exp(-0.0112 t)$	0.9915	0.0264	0.0077
Logarithmic	$MR = 0.949 \ \exp(-0.0134t) + 0.0856$	0.9932	0.0226	0.0061

Table 4.9: Results of statistical analysis for three spouted bed configurations at 65 °C drying air temperature

Conventional bed				
Model	Equation	$\mathbf{R}^2$	RMSE	SSE
Newton	$MR = \exp(-0.0314t)$	0.8961	0.0835	0.0697
Page	$MR = \exp(-0.1491t^{0.5723})$	0.9931	0.0227	0.0046
Henderson & Pabis	MR = 0.8917 exp(-0.0272t)	0.9180	0.0782	0.0551
Two compartment	$MR = 0.3258 \exp(-0.0080t) + 0.6737 \exp(-0.0766t)$	0.9992	0.0088	0.0005
Logarithmic	$MR = 0.8218 \exp(-0.0566t) + 0.1676$	0.9964	0.0174	0.0024
Spouted bed wi	th porous draft tube			
Model	Equation	$\mathbf{R}^2$	RMSE	SSE
Newton	$MR = \exp(-0.0181t)$	0.9616	0.0499	0.0349
Page	$MR = \exp(-0.0491t^{0.7566})$	0.9926	0.0227	0.0067
Henderson & Pabis	$MR = 0.906 \exp(-0.0115t)$	0.9713	0.0404	0.0229
Two compartment	$MR = 0.1602 \exp(-0.0007t) + 0.840 \exp(-0.0278t)$	0.9940	0.0002	0.0002
Logarithmic	$MR = 0.8537 \exp(-0.0273t) + 0.1458$	0.9940	0.0007	0.0002
Spouted bed wi	th non- porous draft tube			
Model	Equation	$\mathbf{R}^2$	RMSE	SSE
Newton	MR = exp - 0.0163t)	0.9820	0.0389	0.0196
Page	$MR = \exp(-0.0125t^{1.065})$	0.9833	0.0389	0.0182
Henderson & Pabis	MR = 1.033 exp(-0.0169t)	0.9839	0.0383	0.0176
Two	$MR = 0.5164 \exp(-0.0093t)$	0.0830	0.0420	0.0176
compartment	$+ 0.5164 \exp(-0.0169t)$	0.7039	0.0420	0.0170
Logarithmic	$MR = 0.9927 \ \exp(-0.0193t) + 0.0545$	0.9856	0.0378	0.0156

Table 4.10: Results of statistical analysis for three spouted bed configurations at 75 °C drying air temperature



Figure 4. 21: A comparison between the observed moisture ratio and predicted moisture ratio of black pepper dried at different drying air temperatures in conventional spouted bed using Logarithmic model.



Figure 4. 22:A comparison between the observed moisture ratio and predicted moisture ratio of black pepper dried at different drying air temperatures in conventional spouted bed using Henderson and Pabis model.



Figure 4. 23:A comparison between the observed moisture ratio and predicted moisture ratio of black pepper dried at different drying air temperatures in conventional spouted bed using Two Compartment model.



Figure 4. 24:A comparison between the observed moisture ratio and predicted moisture ratio of black pepper dried at different drying air temperatures in conventional spouted bed using Newton model.



Figure 4. 25 : A comparison between the observed moisture ratio and predicted moisture ratio of black pepper dried at different drying air temperatures in conventional spouted bed using Page model.

Figure 4. 21 to Figure 4. 25 show the comparison between the observed moisture ratio and predicted moisture ratio of black pepper dried in conventional spouted bed using selected models. According to the results of statistical analyses shown in Table 4.8 to Table 4.10 and Figure 4. 21 to Figure 4. 25, selected thin layer models can describe the drying kinetics of wetted black pepper in three configurations satisfactorily.

Statistical analysis and the observed moisture ratio vs. predicted moisture ratio data laying around the straight line in Figure 4. 21 and Figure 4. 23 prove the Logarithmic model and the Two compartment model are the best fitting models for wetted black pepper drying in any of three configurations of spouted bed dryer.

However in work of Kalwar and Raghavan, Page model described the drying process of shelled corn in draft plates fitted two dimensional spouted bed with sufficient accuracy [33]. Jittanit et al. conducted drying experiments in a fluidized bed dryer and a spouted bed dryer using rewetted samples of paddy, wheat and maize. They fitted the drying kinetics data into four thin layer models and, Page model and Two Compartment model were found as the best fitted model [80].

# **Chapter specific nomenclature**

 $D_{eff}$  -effective moisture diffusivity (m<sup>2</sup>/s)

 $D_i$  - gas inlet diameter (m)

Do -pre exponential factor of Arrhenius equation

D<sub>T</sub> -diameter of draft tube (m)

E<sub>a</sub> -activation energy (kJ/mol)

H - Stagnant bed height (m)

H<sub>T</sub>- tube height (m)

L<sub>H</sub> -entrainment height (m)

 $L_{T}\xspace$  - total height of the draft tube

t $_{\text{Ums}}$  - drying time when air velocity is set to minimum spouting velocity

t  $v_1$  - drying time when air velocity is set to  $v_1$ 

 $T_d\text{-}$  drying air temperature at the inlet of the drying chamber (  $\r{C}$  )

U<sub>ms</sub> -minimum spouting velocity (m/s)

# Abbreviations

CSB - Conventional spouted bed

SBNPDT - Spouted bed equipped with non-porous draft tube

SBPDT - Spouted bed equipped with porous draft tube

 $R^2$  - Coefficient of determination

RMSE - Root mean square error

SSE - Sum of square error

# 5. HYDRODYNAMICS OF SPOUTING, DRYING KINETICS AND MATHEMATICAL MODELING OF DRYING OF RAW BLACK PEPPER IN CONVENTIONAL SPOUTED BED

This chapter presents the results of major experiments described under section 3.1.6.1, 3.1.6.2 and 3.1.6.4. All the experiments were conducted in conventional spouted bed with raw black pepper. Hydrodynamics of black pepper in the conventional spouted bed and drying kinetics of raw black pepper dried in the conventional spouted bed are discussed. Furthermore, results of mathematical modeling of drying kinetics of raw black pepper in conventional spouted bed dryer are presented and discussed.

## 5.1 Minimum spouting velocity

Spouting behavior of raw black pepper was studied. Measurements of Pressure drop across the bed versus air velocity were taken for four stagnant bed heights at ambient temperature.

The variation of pressure drop across the bed with increasing and decreasing air flow for 0.07 m, 0.14 m, 0.18 m and 0.22 m static bed heights of raw black pepper respectively are shown in Figure 5.1 to Figure 5.4. Curve ABCDE in Figure 5.1 shows the evolution of pressure drop across the bed with increasing air flow rate for a loosely packed bed. At low flow rates, pressure drop across the particle bed increased with increasing air velocity. Gradually the inner cavity converted to an internal spout. Further increase of air velocity made the pressure drop to reach a peak value at point B. This peak in the pressure drop was due to the energy required to disrupt the packing.

Height of internal spout increased and formed a fountain of particles above the packed bed. Then the pressure drop declined significantly with increase of the air velocity approaching the point C. Then the bed approached to the point of incipient

spouting, internal cavity ruptured with slight increase in velocity caused sharp decrease in pressure drop to point D and lead to point of onset spouting such that steady spouting was established. As the air velocity was increased further, pressure drop remained constant as in DE.

Curve EDC'B'A in Figure 5.1 shows the variation of the pressure drop across the bed when the air flow rate was reduced. When the air velocity was gradually decreased in a stable spouted bed, the bed remained in the spouting state and the pressure drop remained constant until point C'. When the spout collapsed or with disappering of external spout pressure drop was suddnely increased to point B' since particles get loosely packed and made an resistance to flow. Pressure drop at point B' was a much lower pressure drop compared to peak value reported in flow rate ascending process. Point C' was identified as the state of minimum spouting condition. Similar trend for conical spouted bed with heavy particles was also reported by Sari et al. [112]. Figure 5. 5(a) to Figure 5. 5(d) appear in similar scale, show the comparison of variation of pressure drop across the bed with increasing and decreasing air flow for above mentioned four diffrent stagnant bed heights of raw black pepper.



Figure 5.1: Pressure drop vs. superficial air velocity for 0.07 m static bed of raw black pepper at ambient temperature



Figure 5.2: Pressure drop vs. superficial air velocity for 0.14 m static bed of raw black pepper at ambient temperature



Figure 5.3:Pressure drop vs. superficial air velocity for 0.18 m static bed of raw black pepper at ambient temperature



Figure 5.4: Pressure drop vs. superficial air velocity for 0.22 m static bed of raw black pepper at ambient temperature



Figure 5. 5: Pressure drop vs. superficial air velocity for different static bed heights of raw black pepper at ambient temperature: (a) 0.07 m, (b) 0.14 m, (c) 0.18 m, (d) 0. 22 m

Similarly pressure drop versus air velocity plots were prepared for all the bed heights. Minimum spouting velocities for each bed height were determined and shown in Table 5.1. The results show that the minimum spouting velocity increases with increasing static bed heights since more energy is required to expand the deeper beds to attain loose state to disrupt the packing state [113]. This is in agreement with the results reported for hydrodynamics of silicon particles in draft tubes fitted spouted fluid bed by Zhang et al. [41].

Widely used Mathur and Gishler equation presented in Equation 2.8 and Wu *et al.* correlation shown in Equation 2.10 were used to compare the experimental values of minimum spouting velocity [13, 62]. Equation 5.1 shows the expression for determining maximum pressure drop in either particulate fluidization or spouting [13]. It was used to compare the experimental values of peak pressure drop and the results are shown in Table 5.1.

$$-\Delta P_{\rm M} = H(\rho_{\rm p} - \rho)(1 - \epsilon)g \qquad (5.1)$$

Figure 5.6 shows the predicted and experimental minimum spouting velocities. The experimentally determined values lie between the values predicted by two correlations given in Mathur and Gishler equation (Equation 2.8) and Wu *et al.* correlation (Equation 2.10). However the accuracy of the Mathur and Gishler equation is  $\pm 15\%$  [37] and therefore experiment results approximately lies within this range.



Figure 5.6: Minimum spouting velocity vs. static bed height

Minimum spouting velocity of raw black pepper for the conventional spouted bed vs static bed height were fitted to a power form of function using Matlab 2011. The relationship is given by Equation 5.2 with 0.97 of coefficient of determination ( $\mathbb{R}^2$ ). This is valid for ambient temperature.

$$U_{\rm ms} = 5.26 \, {\rm H}^{0.67} \tag{5.2}$$

H <sub>o</sub> (m)	Experimental	Predicted U <sub>ms</sub> (m/s)		
	U <sub>ms</sub> (m/s)	Mathur & Gishler equation	Wu at al. (m/s)	
0.07	0.86	0.72	1.54	
0.14	1.43	1.02	2.03	
0.18	1.77	1.15	2.25	
0.22	1.86	1.28	2.44	

 Table 5.1: Experimental and predicted minimum spouting velocity for selected

 stagnant bed heights of raw black pepper

 Table 5. 2- Experimental and predicted maximum pressure drop for selected stagnant

 bed heights of raw black pepper

$H_{o}(m)$	Experimental $\Delta P_M$	Predicted $\Delta P_M$
	$(N/m^2)$	$(N/m^2)$
0.07	374.78	289.38
0.14	653.06	578.76
0.18	1205.43	744.12
0.22	1300.00	909.48

All the experimentally determined  $\Delta P_M$  values are underestimated by the predicted  $\Delta P_M$  by Equation 5.1. The experimental results showed that the exponent of static bed height in power form of function shown in Equation 5.2 derived for current study is 0.67. However, exponent of static bed height in both Mathur and Gishler equation (Equation 2.8) and Wu et al. correlation (Equation 2.10) are 0.5 and 0.405 respectively. This difference is due to the use of different geometrical dimensions of spouted beds with varieties of material used to for investigations.

#### **5.2 Drying kinetics**

# 5.2.1 Effect of drying air temperature

Drying experiments were conducted using 0.14 m of stagnant bed of raw black pepper at four different drying air temperatures; 45 °C, 55 °C, 65 °C and 75 °C. The superficial air velocity of 2.37 m/s was maintained for these experiments as explained in section 3.1.6.2. Data containing moisture content versus time obtained from experiments at different air temperatures were used to determine the effect of drying air temperature on drying kinetics. Figure 5. 7 and Figure 5.8 show the moisture content vs. time and the rate of change of moisture content vs. moisture content different drying air temperatures.

Figure 5. 7 shows that increase in the drying air temperature from 45  $^{\circ}$ C to 75  $^{\circ}$ C causes significant reduction of drying time confirming that the total drying time is reduced with the increase in drying air temperature. Serowick *et. al.* has obtained similar trend in their experiments on drying of semi-refined carrageenan in a spouted bed [114].



Figure 5. 7: Moisture content- dry basis vs. time of black pepper when dried in conventional spouted bed at different drying air temperatures. Experiment conditions:  $D_i=0.05$  m, H =0.14 m, and air flow of 2.37 m/s

Black pepper drying rate curves in Figure 5.8 show only falling rate periods which is due to unsaturated surface drying [14]. As discussed in the literature many agricultural products have shown similar drying behaviour with the falling rate drying periods without appearing the constant rate period [82, 31, 115, 108, 116].



Figure 5.8: Rate of change of moisture content vs moisture content- dry basis of black pepper when dried in conventional spouted bed at different drying air temperatures; experiment conditions:  $D_i=0.05$  m, H =0.14 m and air flow of 2.37 m/s

The highest drying rate was achieved while black pepper was dried at drying air temperature of 75 °C. At higher drying air temperatures, hot air provides higher external heat fluxes leading to higher moisture diffusivities and hence higher drying rate. Table 5.3 shows the air requirement for drying of raw black pepper in the conventional spouted bed at different drying air temperatures. In addition, Figure 5.9, Figure 5.10 and

Table 5.3 provide information on the drying time and volume of air required for drying of raw black pepper in conventional spouted bed dryer regarding the drying of 1 kg of raw black pepper, to achieve 1 kg of dried black pepper and to evaporate 1 kg of moisture in raw black pepper. As per the analysis of air requirement of

different drying air temperatures, low amount of hot air is required at higher drying temperatures.

Despite the fact that rise in drying air temperature reduces the time of drying and the specific energy consumption of the process as described in section 5.3, the quality of the final product may get affected. Therefore, attention was paid to analyse the quality of the final product and presented in chapter 6.

Table 5.3: Drying time and volume of air required for raw black pepper when dried in conventional spouted bed at different drying air temperatures. Experiment conditions:  $D_i=0.05$  m, H =0.14 m (bed weight = 1 kg), and air flow of 2.37 m/s

Description Drying			nperature	(°C)
	45	55	65	75
Drying time (min)	650	439	260	181
Volume of air sent throughout the drying period (m <sup>3</sup> )	1632.41	714.00	652.96	454.56
Volume of air required to evaporate 1kg of moisture (m <sup>3</sup> )	2192.19	962.74	890.13	608.12
Volume of air required to dry 1kg of raw black pepper (m <sup>3</sup> )	1632.41	714.00	652.96	454.56
Volume of air required to have 1kg of dried black pepper after drying of raw black pepper (m <sup>3</sup> )	6392.78	2763.54	2450.69	1800.22
Average moisture removal rate (g/s)	0.019	0.028	0.047	0.069



Figure 5.9: Volume of air required for raw black pepper when dried in conventional spouted bed at different drying air temperatures. Experiment conditions:  $D_i$ =0.05 m, H =0.14 m, and air flow of 2.37 m/s



Figure 5.10: Drying time for black pepper when dried in conventional spouted bed at different drying air temperatures. Experiment conditions:  $D_i=0.05$  m, H =0.14 m, and air flow of 2.37 m/s

### 5.2.2 Effect of bed height

Figure 5.11 and Figure 5.12 show the effect of bed height on the drying kinetics of black pepper in the conventional spouted bed. Figure 5.11 shows that time required for a given moisture reduction increases with the bed height. The time required to reach a moisture content of 15 % dry basis are 225 min, 260 min, 330 min and 335 min in the 0.07 m, 0.14 m, 0.18 m and 0.20 m beds respectively. Figure 5.12 indicates that rate of change of moisture content decreases with increasing bed height showing that deeper beds require longer drying time for moisture removal than the shallow bed. In spouted bed dryers, intensive heating and evaporation takes place in the spout region while particles are tempered in the annulus region, rate of change of moisture content of deep bed is lower than that of shallow beds. According to Law and Mujumdar, this behaviour will appear only up to a certain bed height beyond which the bed height will not have any effect on the drying rate [108].

Table 5.4, Figure 5.13 and Figure 5.14 show the drying time and air requirement for drying of raw black pepper at different bed heights. However, average of moisture evaporated per unit time at 0.07 m, 0.14 m and 0.20 m static bed heights are 0.033 g/s, 0.047 g/s, 0.056 g/s and 0.066 g/ s respectively. This result indicates despite the rate of change of moisture content (dX/dt) decreases with the increasing bed height; the average weight of moisture evaporated per unit time is increased. In addition, volume of air required to dry 1 kg of raw black pepper or to get 1kg of dried black pepper is decreased with increasing static bed height of the dryer. This is due to the fact that at higher static bed heights of spouted bed, a larger fraction of thermal energy of hot air is utilized for drying operation without sending out through the dryer outlet. Therefore it is economical to run spouted bed dryers with deeper bed heights or closer to their maximum spoutable heights.



Figure 5.11: Moisture content (% dry basis) versus time of black pepper when dried in conventional spouted bed at different static bed heights; Experiment conditions:  $D_i = 0.05 \text{ m}$ ,  $T_d = 65 \degree C$  and air flow of 2.37 m/s



Figure 5.12: Rate of change of moisture content, dx/dt versus moisture content (% dry basis) of black pepper when dried in conventional spouted bed at different static bed heights; experiment conditions:  $D_i = 0.05$  m,  $T_d = 65$  °C and air flow of 2.37 m/s

Description	Static bed height (m)			)
	0.07	0.14	0.18	0.2
Drying time (min)	225	260	330	340
Volume of air sent throughout the drying period (m <sup>3</sup> )	562.55	652.96	828.76	841.32
Volume of air required to evaporate 1kg of moisture (m <sup>3</sup> )	1263.75	890.13	742.8	637.22
Volume of air required to dry 1kg of raw black pepper (m <sup>3</sup> )	937.59	652.96	552.51	474.38
Volume of air required to have 1kg of dried black pepper after drying of raw black pepper (m <sup>3</sup> )	3632.84	2450.69	2156.74	1856.25
Average moisture removal rate (g/s)	0.033	0.047	0.056	0.066

Table 5.4: Volume of air required for raw black pepper when dried in conventional spouted bed at different static bed heights; experiment conditions:  $D_i = 0.05 \text{ m}$ ,  $T_d = 65 \text{ °C}$  and air flow of 2.37 m/s



Figure 5.13 :Volume of air required for raw black pepper when dried in conventional spouted bed at different static bed heights; experiment conditions:  $D_i = 0.05 \text{ m}$ ,  $T_d = 65 \text{ °C}$  and air flow of 2.37 m/s



Figure 5.14: Drying time of raw black pepper when dried in conventional spouted bed at different static bed heights; experiment conditions:  $D_i = 0.05 \text{ m}$ ,  $T_d = 65 \text{ °C}$  and air flow of 2.37 m/s

## 5.2.3 Effect of air flow rate

Effect of airflow rate on drying kinetics of black pepper in conventional spouted bed was analyzed. Air velocity was set at 1.6 m/s, 1.97 m/s and 2.37 m/s. Figure 5.15 and Figure 5.16 show the moisture content as a function of time and rate of change of moisture content as a function of moisture content plotted at different air velocities.

Table 5.5, Figure 5.17 and Figure 5.18 show the air requirement for drying of raw black pepper in the conventional spouted bed for different air velocities.

Significant decrease in drying time and increase in drying rate were observed when air velocity changes from 1.6 m/s (closer to minimum spouting velocity) to 1.97 m/s. However, further increase of air velocity up to 2.37 m/s does not show significant change in drying rate or time. Low volume of air is also required for drying at low air velocities. Therefore, it is suitable to adjust the airflow rate just above the minimum spouting velocity to cause stable spouting operation considering the energy and cost involved.



Figure 5.15: Moisture content (% dry basis) vs. time of black pepper when dried in conventional spouted bed dryer at different superficial air velocities; Experiment conditions:  $D_i=0.05$  m,  $T_d=65$  °C and H=0.14 m



Figure 5.16: Rate of change of moisture content vs. moisture content- dry basis of black pepper when dried in conventional spouted bed at different superificial air velocities; Experiment conditions:  $D_i=0.05$  m,  $T_d=65$  °C and H=0.14 m

Table 5.5 : Drying time and volume of air required for raw black pepper when dried
in conventional spouted bed dryer at different superficial air velocities; Experiment
conditions: $D_i=0.05$ m, $T_d=65$ °C and H=0.14 m

Description	Superficial air velocity (m/s)			
	2.37	1.97	1.60	
Drying time (min)	260	266	270	
Volume of air sent throughout the drying period (m <sup>3</sup> )	652.96	557.16	457.89	
Volume of air required to evaporate 1kg of moisture (m <sup>3</sup> )	890.13	751.70	612.15	
Volume of air required to dry 1kg of raw black pepper (m <sup>3</sup> )	652.96	557.16	457.89	
Volume of air required to have 1kg of dried black pepper after drying of raw black pepper (m <sup>3</sup> )	2450.69	2152.80	1816.99	
Average moisture removal rate (g/s)	0.047	0.046	0.046	



Figure 5.17: Drying time of raw black pepper when dried in conventional spouted bed dryer at different superficial air velocities; Experiment conditions:  $D_i$ =0.05 m,  $T_d$ =65 °C and H=0.14 m



Figure 5.18 : Volume of air required for raw black pepper when dried in conventional spouted bed dryer at different superficial air velocities; Experiment conditions:  $D_i=0.05$  m,  $T_d=65$  °C and H=0.14 m

However, airflow with superficial velocity of 2.37 m/s was used for the experiments conducted to analyze the effect of drying temperature and bed height as described in section 3.2.1 and 3.2.2. Air velocity of 2.37 m/s was selected to have stable spouting operation with all possible bed heights. Since 2.37 m/s was below the terminal velocity for the shortest bed height; 0.07 m used and it was also above the minimum spouting velocity of maximum spoutable bed height; 0.25 m of the spouted bed column used for the current investigation.

#### 5.2.4 Summary on effect of operational variables

Drying air temperature, airflow rate and static bed height have effect on rate of change of moisture content from black pepper. Decreasing the bed height or increasing the air temperature, higher drying rates can be obtained.

High moisture content of black pepper particles provides large driving force for mass transfer at the initial stage of drying. Higher concentration gradient from surface to deeper parts of the particle creates low internal resistance for diffusion of moisture [14]. Therefore, initially drying rates are high as shown in Figure 5.8, Figure 5.12 and Figure 5.16. Further, at initial stage of drying when the moisture content is high, the drying rates at given moisture content show a significant difference with the above operational parameters; temperature, bed height and air flow rate. However, the drying curves overlaps at later stages of drying showing less effect of the above parameters.

Particle has high internal resistance to moisture transfer when the material being dried reaches to low moisture content. Usually high internal moisture resistance is observed at the end of the falling rate period [108]. Therefore, when the drying is progressed, the drying rate gradually falls showing that more energy is required for diffusion of moisture.

#### **5.3 Specific energy consumption**

Specific Energy Consumption (SEC) defined as in Chapter 2, were calculated based on the Equation 2.8 and the results are presented in Table 5.6 and Figure 5.19. Maximum SEC values of 54.43 MJ/kg, 38.03 MJ/kg and 26.17 MJ/kg were obtained at 55 °C drying temperature for 0.07 m, 0.14 m and 0.20 m stagnant beds respectively. Then SEC decreased gradually with temperature due to inverse effect of decrease in drying time [21]. Table 5.6 shows that the SEC decreased gradually with increasing static bed height of black pepper in the conventional spouted bed for a particular drying air temperature at selected air velocity. This phenomenon recalls fact that utilization of a larger fraction of thermal energy of hot air at higher static bed heights of the spouted bed as discussed in section 5.2.2.

Figure 5.19 shows that SEC increases with increasing air velocity due to the increase in energy required to heat up higher flow rates of ambient air to drying temperature and the loss of more energy with exhaust air leaving the dryer

Static bed	Drying	SEC		
height	temperature	(MJ/kg of	(kWh/kg of	
<b>H</b> (m)	$T_d$ (°C)	moisture	water removed)	
		evaporated)		
0.07	45	33.98	9.44	
	55	54.43	15.12	
	65	43.06	11.96	
	75	54.29	15.08	
0.14	45	32.22	8.95	
	55	38.03	10.56	
	65	29.80	8.28	
	75	26.51	7.36	
0.20	45	21.69	6.03	
	55	26.17	7.27	
	65	22.66	6.29	
	75	16.62	4.62	

Table 5.6: Specific energy consumption for drying of black pepper at different drying temperaures (air velocity 2.37 m/s)



Figure 5.19: Specific energy consumption vs. air velocity for black pepper dried in conventional spouted bed at different superficial air velocities; Experiment conditions:  $D_i = 0.05 \text{ m}$ ,  $T_d = 65 \text{ °C}$  and H = 0.14 m

# **5.4 Effective moisture diffusivity of raw pepper drying in conventional spouted bed**

During drying of particles, moisture diffuses from interior of the particles to the particle surface and moisture transfer to the particle air interface. Effective diffusivity refers to overall transport coefficient that takes into account all the method of moisture transporting from interior to the surface of the particle namely molecular diffusion, liquid diffusion, vapour diffusion, etc. When the drying process has a falling rate period, mass transfer from particles to gas is controlled internally by internal movement of moisture [13].

Data containing moisture content versus time collected from the drying experiments conducted at four different drying air temperatures: 45 °C, 55 °C, 65 °C and 75 °C were converted to moisture ratio values using Equation 2.4. Plots of Logarithmic of moisture ratio (Ln MR) as a function of time were plotted for different drying air temperatures as shown in Figure 5.20. Effective diffusivity ( $D_{eff}$ ) of black pepper at

different drying air temperatures were calculated using the gradient of the plots as per the Equation 2.6 and presented in Table 5.7. According to the results,  $D_{eff}$  of black pepper depends on drying temperature. In addition,  $D_{eff}$  increased with increasing temperature.

Effective diffusivity ( $D_{eff}$ ) values were ranged from  $6.05 \times 10^{-11} \text{ m}^2/\text{s}$  to  $2.03 \times 10^{-10} \text{ m}^2/\text{s}$  for the temperatures under investigation and the results are in the same order of magnitude as the results obtained for other agricultural crops such as coffee and peanuts [117].

As shown in Table 5.7, effective diffusivity ( $D_{eff}$ ) of black pepper increases with increasing drying temperatures. Usually with increasing drying temperatures, drying rates or moisture removal rate is increased. Because at high drying temperature, drying air provides high external heat fluxes leading to higher moisture diffusivities and hence higher drying rate [108].



Figure 5.20: Ln (MR) vs time of raw black pepper dried in conventional spouted bed dryer

Natural logarithm of effective diffusivity ( $D_{eff}$ ) vs. reciprocal of absolute drying temperature ( $1/T_d$ ) were plotted and shown in Figure 5.21. The pre-exponential factor ( $D_0$ ) and the activation energy for the raw black pepper dried at conventional spouted bed were calculated using Arrhenius type relationship given in Equation 2.7 and found to be  $3.68 \times 10^{-4}$  m<sup>2</sup>/s and 42.41 kJ/mol respectively. This result

corresponds well with those given in the literature for drying of different foods: between 18- 42 kJ/mol for corn [21]; between 19 -22 kJ/mol for red apple [109]; and between 49-54 kJ/mol for grapes [16].



Figure 5.21: Ln (effective diffusivity) vs 1/temperature for raw black pepper dried in conventional spouted bed dryer

Table 5.7 : Effective moisture diffusivity of raw black pepper dried in conventional spouted bed dryer at different temperatures

Drying temperature	Effective diffusivity	$\mathbf{R}^2$ of the	
T <sub>d</sub> (°C)	$D_{eff}(m^2/s)$	plot	
45	$4.00 \times 10^{-11}$	0.9141	
55	$6.28  imes 10^{-11}$	0.9304	
65	$1.07 \times 10^{-10}$	0.9138	
75	$1.55 \times 10^{-10}$	0.9151	

## 5.6 Mathematical modeling of drying kinetics

Drying experiments were conducted for four different stagnant bed heights of raw black pepper at four different drying air temperatures as described in section 3.1.6.4. Moisture content of drying material was recorded at selected time intervals and moisture content was normalized to moisture ratio.

Five thin layer drying models namely Newton, Page, Henderson and Pabis, Two compartment and Logarithmic model were used to fit the drying behaviour of black pepper drying in the conventional spouted bed. RMSE, SSE and coefficient of determination ( $R^2$ ) were calculated to identify the best model which describes the thin layer drying behaviour.

RMSE value closer to zero or very low indicates that the model has a smaller random error component, and that the fit will be more useful for prediction. Just as with RMSE value, SSE value closer to 0 shows model is useful for prediction.  $R^2$  can take on any value between 0 and 1, with a value closer to 1 indicating that a greater proportion of variance is accounted for by the model.

Five thin layer drying models developed for black pepper for each drying temperature and initial stagnant bed height in spouted bed dryer are shown in Table 5.8 to Table 5. 12. Statistical parameters  $R^2$ , RMSE and SSE which measure goodness of the fit were also presented together with the developed thin layer drying models. According to the statistical analysis, all the models were found to be satisfactorily described the drying behaviour of black pepper dried in spouted bed.

Table 5.8: Newton model developed for black pepper drying in conventional spouted bed for various stagnant bed heights and drying air temperatures at 2.37 m/s of air velocity

T <sub>d</sub>	Н	Newton model	$\mathbf{R}^2$	RMSE	SSE
(°C)	( <b>m</b> )	$MR = \exp(-kt)$			
45	14	$MR = \exp(-0.0091t)$	0.9980	0.0133	0.0030
	18	$MR = \exp(-0.0101t)$	0.9983	0.0126	0.0017
	20	$MR = \exp(-0.0041t)$	0.9971	0.0154	0.0062
	22	$MR = \exp(-0.0038t)$	0.9965	0.0162	0.0078
55	14	$MR = \exp(-0.0066 \ t)$	0.9902	0.0296	0.0193
	18	$MR = \exp(-0.0052t)$	0.9937	0.0229	0.0110
	20	$MR = \exp(-0.0064t)$	0.9939	0.0231	0.0091
	22	$MR = \exp(-0.0057t)$	0.9958	0.0184	0.0102
65	7	$MR = \exp(-0.0131t)$	0.9960	0.0191	0.0044
	14	$MR = \exp(-0.0106t)$	0.9938	0.02271	0.0088
	18	$MR = \exp(-0.0091t)$	0.9980	0.0133	0.0030
	20	$MR = \exp(-0.0101t)$	0.9983	0.0126	0.0017
	22	$MR = \exp(-0.0052t)$	0.9956	0.0199	0.0075
75	14	$MR = \exp(-0.0175t)$	0.9975	0.0161	0.0023
	18	$MR = \exp(-0.0137t)$	0.9957	0.0204	0.0050
	20	$MR = \exp(-0.01404t)$	0.9956	0.0215	0.0046
	22	$MR = \exp(-0.0091t)$	0.9985	0.0114	0.0022

Table 5.8 summarizes the statistical analysis of Newton model developed for black pepper drying in conventional spouted bed for various stagnant bed heights and drying air temperatures at 2.37 m/s of air velocity. According to Table 5.8,  $R^2$ , RMSE and SSE values ranged from 0.9902 to 0.9985, 0.0114 to 0.0296 and 0.0017 to 0.0193 respectively.
Table 5.9: Page's model developed for black pepper drying in conventional spouted bed for various stagnant bed heights and drying air temperatures at 2.37 m/s of air velocity

T <sub>d</sub>	Н	Page model	$\mathbf{R}^2$	RMSE	SSE
(°C)	( <b>m</b> )	$MR = \exp(-\mathbf{k} t^n)$			
	14	$MR = \exp(-0.0091t^1)$	0.9980	0.0133	0.0030
15	18	$MR = \exp(-0.0100t^{0.9900})$	0.9983	0.0132	0.0017
43	20	$MR = \exp(-0.0040t^1)$	0.9971	0.0154	0.0062
	22	$MR = \exp(-0.0045t^{0.9701})$	0.9968	0.0157	0.0071
	14	$MR = \exp(-0.0066t^1)$	0.9902	0.0296	0.0193
55	18	$MR = \exp(-0.0052t^1)$	0.9937	0.0229	0.0110
55	20	$MR = \exp(-0.0064t^1)$	0.9939	0.0231	0.0091
	22	$MR = \exp(-0.0057t^1)$	0.9958	0.0184	0.0102
	7	$MR = \exp(-0.0130 t^{0.9993})$	0.9960	0.0199	0.0044
	14	$MR = \exp(-0.01061 \ t^1)$	0.9939	0.0233	0.0087
65	18	$MR = \exp(-0.0091t^1)$	0.9980	0.0133	0.0030
	20	$MR = \exp(-0.0101t^{0.9900})$	0.9983	0.0132	0.0017
	22	$MR = \exp(-0.0057t^1)$	0.9956	0.0199	0.0075
	14	$MR = \exp(-0.0175t^1)$	0.9978	0.0162	0.0021
75	18	$MR = \exp(-0.0175t^1)$	0.9957	0.0204	0.0050
	20	$MR = \exp(-0.0141t^1)$	0.9956	0.0215	0.0046
	22	$MR = \exp(-0.0091 t^1)$	0.9985	0.0114	0.0022

Table 5.9 shows the statistical analysis of Page model developed for black pepper drying in conventional spouted bed for various stagnant bed heights and drying air temperatures at 2.37 m/s of air velocity.  $R^2$ , RMSE and SSE values ranged from 0.9902 to 0.9985, 0.0114 to 0.0296 and 0.0017 to 0.0193 respectively.

Table 5.10: Henderson and Pabis model developed for black pepper drying in conventional spouted bed for various stagnant bed heights and drying air temperatures at 2.37 m/s of air velocity

T <sub>d</sub>	Н	Henderson and Pabis model $MB = a \exp(-kt)$	$\mathbf{R}^2$	RMSE	SSE
(°C)	(cm)	$MK - a \exp(-Kt)$			
	14	$MR = 1.021 \exp(-0.0093t)$	0.9985	0.0117	0.0022
45	18	$MR = 0.9964 \exp(-0.0100t)$	0.9983	0.0131	0.0017
	20	$MR = 1.02 \exp(-0.0041t)$	0.9979	0.0135	0.0045
	22	$MR = 0.9939\exp(-0.0037t)$	0.9965	0.0163	0.0077
	14	$MR = 1.046 \exp(-0.0069t)$	0.9928	0.0260	0.0142
55	18	$MR = 1.017 \exp(-0.0053t)$	0.9941	0.0228	0.0104
	20	$MR = 1.016 \exp(-0.0065t)$	0.9942	0.0232	0.0086
	22	$MR = 1.014 \exp(-0.0058t)$	0.9960	0.0182	0.0097
	7	$MR = 1.005 \exp(-0.0130t)$	0.9960	0.0182	0.0097
	14	$MR = 1.007 \exp(-0.0107t)$	0.9939	0.0233	0.0087
65	18	$MR = 1.021 \exp(-0.0093t)$	0.9985	0.0117	0.0022
	20	$MR = 0.9964 \exp(-0.0099t)$	0.9983	0.0131	0.0017
	22	$MR = 1.038 \exp(-0.0053t)$	0.9974	0.0156	0.0044
	14	$MR = 1.014\exp(-0.0177t)$	0.9978	0.0162	0.0021
75	18	$MR = 1.015 \exp(-0.0139t)$	0.9960	0.0206	0.0047
	20	$MR = 1.028 \exp(-0.0144t)$	0.9967	0.0196	0.0035
	22	$MR = 1.011 \exp(-0.0093t)$	0.9987	0.0111	0.0020

Table 5.10 shows the statistical analysis of Henderson and Pabis model developed for black pepper drying in conventional spouted bed for various stagnant bed heights and drying temperatures at 2.37 m/s of air velocity. R<sup>2</sup>, RMSE and SSE values were ranged from 0.9928 to 0.9985, 0.0111 to 0.0260 and 0.0017 to 0.0142 respectively.

T <sub>d</sub>	Н	Two compartment model	$\mathbf{R}^2$	RMSE	SSE
(°C)	(cm)	$MR = \operatorname{aex} p(-\mathbf{k}_1 t) + \mathbf{b} \exp(-\mathbf{k}_2 t)$			
	14	$MR = 0.4825 \exp(-0.0093t) + 0.5377 \exp(-0.0093t)$	0.9985	0.0125	0.0022
45	18	$MR = 0.512 \exp(-0.0101t) + 0.4843 \exp(-0.0099t)$	0.9983	0.0147	0.0017
15	20	$MR = -1.432 \exp(-0.0040t) + .457\exp(-0.0041t)$	0.9979	0.0140	0.0045
	22	$MR = 0.05509 \exp(-0.0037t) + 0.9390 \exp(-0037t)$	0.9965	0.0169	0.0077
	14	$MR = 0.5646 \exp(-0.0070t) + .4818 \exp(-0.0069t)$	0.9928	0.0273	0.0142
55	18	$MR = 0.7296 \exp(-0.0053t) + .2876 \exp(-0.0053 t)$	0.9941	0.0240	0.0104
	20	$MR = 0.1765 \exp(-0.0065t) + 0.8394\exp(0065t)$	0.9942	0.0248	0.0086
	22	$MR = -0.06805 \exp(-0.0058t) + 1.081 \exp(0058t)$	0.9960	0.0189	0.0097
	7	$MR = 0.5024 \exp(-0.0131t) + 0.5024 \exp(-0.0131t)$	0.9961	0.0220	0.0043
	14	$MR = 0.0065 \exp(0.0044 \text{ t}) + 1.011 \exp(-0.0113 \text{ t})$	0.9955	0.0214	0.0064
65	18	$MR = 0.4825 \exp(-0.0093t) + 0.5377 \exp(-0.0093t)$	0.9985	0.0125	0.0022
	20	$MR = 0.512 \exp(-0.0101t) + 0.4843 \exp(-0.0099t)$	0.9983	0.0147	0.0017
	22	$MR = -0.0026\exp(-0.0054t) + 1.04\exp(-0.0053t)$	0.9974	0.0165	0.0044
	14	$MR = -0.069 \exp(-0.0174t) + 1.08 \exp(-0.0176t)$	0.9978	0.0187	0.0021
75	18	$MR = 0.3968 \exp(-0.0139t) + 0.6177 \exp(-0.0139t)$	0.9960	0.0227	0.0047
,	20	$MR = 0.5207 \exp(-0.0144t) + 0.5069 \exp(-0.0144t)$	0.9967	0.0222	0.0035
	22	$MR = 0.4774 \exp(-0.0094t) + 0.532 \exp(-0.0091t)$	0.9987	0.0119	0.0020

Table 5. 11: Two compartment model developed for black pepper drying in conventional spouted bed for various stagnant bed heights and drying air temperatures at 2.37 m/s of air velocity

Table 5. 11 shows the statistical analysis of Two Compartment model developed for black pepper drying in conventional spouted bed for various stagnant bed heights and drying temperatures at 2.37 m/s of air velocity. R<sup>2</sup>, RMSE and SSE values ranged from 0.9928 to 0.9987, 0.0119 to 0.0273 and 0.0009 to 0.0142 respectively.

Table 5. 12: Logarithmic model developed for black pepper drying in conventional spouted bed for various stagnant bed heights and drying air temperatures at 2.37 m/s of air velocity

T <sub>d</sub>	Ho	Logarithmic model	R <sup>2</sup>	RMSE	SSE
(°C)	(cm)	MR = a exp(-kt) + b			
	14	$MR = 1.027 \ exp(-0.0091 \ t) - 0.0107$	0.9987	0.0116	0.0020
15	18	MR = 0.9979 exp(-0.0099 t) - 0.0023	0.9970	0.0162	0.0055
+3	20	$MR = 1.027 \ exp(-0.0041 \ t) - 0.0028$	0.9979	0.0137	0.0045
	22	$MR = 0.9854 \exp(-0.0040 \ t) + 0.0184$	0.9971	0.0152	0.0065
	14	$MR = 1.07 \ exp(-0.0064 \ t) - 0.0367$	0.9940	0.0242	0.0117
55	18	$MR = 1.018 \ exp(-0.0053t) - 0.0011$	0.9941	0.0234	0.0104
	20	$MR = 1.022 \ exp(-0.0063 \ t) - 0.0088$	0.9943	0.0237	0.0085
	22	$MR = 1.033 \ exp(-0.0053t) - 0.0342$	0.9976	0.0143	0.0058
	7	$MR = 1.003 \ exp(-0.0131t) - 0.0022$	0.9961	0.0208	0.0043
	14	MR = 0.9979 exp(-0.0099 t) - 0.0023	0.9983	0.0138	0.0017
65	18	MR = 0.9925 exp(-0.0116t) + 0.0271	0.9954	0.0209	0.0066
	20	$MR = 1.027 \exp(-0.0091 t) - 0.0107$	0.9987	0.0116	0.0020
	22	$MR = 1.054 \ exp(-0.0050t - 0.0242)$	0.9980	0.0142	0.0034
75	14	$MR = 1.017 \ exp(-0.0175 \ t) - 0.0045$	0.9978	0.0172	0.0021
	18	$MR = 1.019 \ exp(-0.0136 \ t) - 0.0065$	0.9960	0.0214	0.0046
15	20	MR = 1.072 exp(-0.0125 t) - 0.0591	0.9992	0.0104	0.0009
	22	$MR = 1.018 \ exp(-0.0089t) - 0.0113$	0.9988	0.0108	0.0018

According to the statistical analysis of Logarithmic model developed,  $R^2$ , RMSE and SSE values ranged from 0.9940 to 0.9992, 0.0104 to 0.0242 and 0.0117 to 0.0009.

Drying rate constant, k value for Newton models are in the range of 0.003-0.018 per minute, 0.003-0.017 per minute for k value of both Logarithmic model and Henderson & Pabis model. Value of constant 'a' in the Henderson and Pabis model is varied between 0.99- 1.04 which imply constant 'a' is approximately equal to 1. Values of constant, n, in the Page model are approximately equal to the one. This can be compared with the Newton model and the Henderson and Pabis model where 'A'

term is not appearing, which denotes the constant 'a' is equal to the one. In addition 'a' values of Logarithmic model also fluctuate in 0.98 to 1.07 where it is approximately equal to one. Further, the very low parameter 'b' values which fluctuate in -0.05 to 0.02 shows that the effect of the second term in the Logarithmic model is less significant. The above comparison of the parameters shows the similarity of the four models namely Newton, Page, Hendeson & Pabis and Logarithmic equations valid for drying of black pepper in spouted bed. A Similar trend has been shown for the microwave drying of black pepper also [118].

Also  $k_1$  and  $k_2$  values of Two Compartment models are almost similar for many runs while summation of drying constants 'a' and 'b' of every run is approximately equal to the one also.

Considering the statistical analysis as mentioned in Table 5.8 to Table 5. 12, Logarithmic models was appeared to be the best fitting model due to highest  $R^2$  and lowest RMSE and SSE values. In addition, it has been appeared as the best model for many runs. Figure 5.22 to Figure 5.24 show that good relationship between observed moisture ratio with predicted moisture ratio from Logarithmic model of black pepper in conventional spouted bed dryer at different drying air temperatures for selected three bed heights.



Figure 5.22: A comparison between the experimental moisture ratio and predicted moisture ratio (using Logarithmic model) of black pepper dried in various temperatures at static 0.14 m bed height



Figure 5.23: A comparison between the experimental moisture ratio and predicted moisture ratio (using Logarithmic model) of black pepper dried in various temperatures at static 0.10 m bed height



Figure 5.24: A comparison between the experimental moisture ratio and predicted moisture ratio (using Logarithmic model) of black pepper dried in various temperatures at static 0.20 m bed height

The developed models shown in Table 5.8 to Table 5. 12 can be used for the specified drying conditions only. Therefore, four additional models were introduced for black pepper drying in conventional spouted bed by correlating drying constants and coefficients of Logarithmic model shown in Table 5. 12 to stagnant bed height only at specified temperatures; 45 °C, 55 °C, 65 °C, and 75 °C. The optimal results of the models for black pepper drying from 45-75 °C are given in Table 5. 13. The model constants and coefficients are as follows:

Logarithmic model is derived from the Henderson and Pabis model, which related to the first term of the analytical solution of the diffusion equation. Logarithmic model can describe the black pepper drying process in spouted bed for 45-75 °C drying temperatures as a function of initial stagnant bed height. Modified equations for spouted bed drying black pepper are shown Table 5. 14.

<b>T</b> <sub>d</sub> (° <b>C</b> )	coefficie	equation	$\mathbf{R}^2$
	nt		
45	a	$1.0720 + 0.5630H - 1.8660 H^2$	0.6780
	k	$0.0021 + 0.0441 H - 0.1650 H^2$	0.8499
	b	-0.0750 + 0.4097 H	0.7723
55	а	$1.442 - 4.0191 H + 9.5455 H^2$	0.8818
	k	$0.0031 + 0.0475 H - 0.175 H^2$	0.7655
	b	$-0.2944 + 2.7289 H - 6.3441 H^2$	0.9999
65	а	$1.452 - 5.790 H + 18.15 H^2$	0.9649
	k	$-0.05885 + 0.8442 H - 2.519 H^2$	0.9967
	b	$0.5728 + 6.913 H - 20.19 H^2$	0.7965
75	a	$0.9864 + 0.35H - 0.9375 H^2$	0.9900
	k	0.0318 – 0.1010 H	0.9647
	b	0.0079 - 0.0852H	0.9462

 Table 5. 13: The results of the multiple regressions on the coefficients and constants of the logarithmic model for the effects of the stagnant bed height

$T_d$	Modified Logarithmic model	$\mathbf{R}^2$	Equatio
(°C)			n
			No
45	$MR = (1.0720 + 0.5630 H - 1.866 H^2) \exp(0.0021 + 0.0441 H - 0.1650 H^2) t + (-0.075 H^2) \exp(0.0021 + 0.0441 H - 0.1650 H^2) t + (-0.075 H^2) t$	0.8897	5.3
	+ 0.4097 H)		
55	$MR = (1.442 - 4.0191 H + 9.5455 H^2) \exp(0.0031 + 0.0475 H - 0.1750H^2)t$	0.9935	5.4
	$+ (-0.2944 + 2.7289 H - 6.3441 H^2)$		
65	$MR = (1.4520 - 5.79H + 18.15 H^2) \exp((-0.05885 + 0.8442 H - 2.519 H^2) t$	0.9960	5.5
	$+ (0.5728 + 6.913 H - 20.19 H^2)$		
75	$MR = (0.9864 + 0.3500H - 0.9375 H^2) \exp((0.0318 - 0.1010 H)t) + (0.0079 - 0.0852H)$	0.9893	5.6

Table 5. 14: Modified Logarithmic model for black pepper drying for 45  $^{\circ}$ C - 75  $^{\circ}$ C

Equations 5.3 to 5.6 in Table 5. 14 are the modified Logarithmic model. They can be used to estimate the drying time of black pepper in a spouted bed dryer for given moisture reduction. It shows that those models satisfactorily represented the drying of black pepper in the ranges of 45-75  $^{\circ}$ C air temperature, 0.14 -0.22 m stagnant bed heights and in 2.37 m/s air velocity.

Figure 5.25 to Figure 5.28 show that these models shown in Equations 5.3 to 5.6 are in good agreement with the experimental results under all drying conditions. Figure 5.29 to Figure 5.32 show comparison of experimental moisture ratio and the predicted moisture ratio estimated using developed models (Equations 5.3 to 5.6). When comparing the experimental moisture ratio values with the predicted values estimated using developed models at any particular drying condition, the values lay around a straight line for black pepper. This means that the modified model is valid at drying air temperatures of 45-75 °C and velocities  $2.37 \text{ ms}^{-1}$ .



Figure 5.25: A comparison between the experimental moisture ratio and predicted moisture ratio of black pepper dried in 45  $^{\circ}$ C



Figure 5.26: A comparison between the experimental moisture ratio and predicted moisture ratio of black pepper dried in 55  $^{\circ}$ C



Figure 5.27: A comparison between the experimental moisture ratio and predicted moisture ratio of black pepper dried in  $65\degree$ C



Figure 5.28: A comparison between the experimental moisture ratio and predicted moisture ratio of black pepper dried in 75  $^{\circ}$ C



Figure 5.29: A comparison between the experimental moisture ratio and predicted moisture ratio of black pepper dried in various temperatures at static 0.14 m bed height



Figure 5.30: A comparison between the experimental moisture ratio and predicted moisture ratio of black pepper dried in various temperatures at static 0.18 m bed height



Figure 5.31: A comparison between the experimental moisture ratio and predicted moisture ratio of black pepper dried in various temperatures at static 0.20 m bed height



Figure 5.32: A comparison between the experimental moisture ratio and predicted moisture ratio of black pepper dried in various temperatures at static 0.22 m bed height

### **Chapter specific nomenclature**

 $D_{eff}$  -effective moisture diffusivity (m<sup>2</sup>/s)

D<sub>i</sub> - gas inlet diameter (m)

Do -pre exponential factor of Arrhenius equation

- E<sub>a</sub> -activation energy (kJ/mol)
- H<sub>o</sub> -static bed height (m)

 $T_d$  - drying air temperature at the inlet of the drying chamber (°C)

U<sub>ms</sub> -minimum spouting velocity (m/s)

X - moisture content ( kg moisture/ kg dry solid)

 $\Delta P_{\rm M}$  - Peak pressure drop (N/m<sup>2</sup>)

## Abbreviations

- MR Moisture ratio
- R<sup>2</sup> Coefficient of determination
- RMSE Root mean square error
- SEC Specific energy consumption
- SSE Sum of square error

### Chapter 6

## 6. ANALYSIS OF QUALITY OF BLACK PEPPER

This chapter presents the investigations on the essential oil derived from black pepper dried at different drying conditions. This investigation consists of the extraction and analysis of essential oil using GC-MS technique and essential oil yield.

Thermal operations can cause degradation and evaporation of some compounds. Hence, a percentage of certain compounds, which are responsible to generate aroma and pungency of black pepper can be loss during the drying process. Main pungent component of black pepper is non-volatile but most of the aroma compounds in black pepper are volatile [5]. Loss of pungent component is negligible compared to aroma compounds in black pepper during a thermal operation. Therefore, our investigation is focussed to analyse the volatile components of black pepper dried in the conventional spouted bed. As a result of this analysis, optimum drying conditions will be identified for black pepper drying in conventional spouted bed to achieve maximum yield of essential oil with consistent quality.

#### 6.1 Analysis of black pepper essential oil

Essential oil extracted from black pepper dried at five different drying conditions were analysed using Gas Chromatography Mass Spectrometry technique. Five different drying conditions are sun drying and spouted bed dying at 45 °C, 55 °C, 65 °C and 75 °C drying air temperatures with minimum spouting conditions. Section 3.1.6.5 gives more information on the preparation of essential oil. Twenty chromatograms of black pepper essential oil were studied as each experiment was replicated. Appendix D shows the chromatograms of 20 black pepper essential oil samples with their gas chromatography data sheets.

Approximately 230 components were identified in the essential oil extracted from the black pepper dried under sun radiation and spouted bed dryer. Mean composition of each components at different drying conditions are listed in Table 1 in Appendix E. The analysis of components shows that the black pepper oil comprised mainly monoterpenes and sesquiterpenes. 13-66% of total oil was monoterpenes while 24-47% of total oil was sesquiterpenes. Table 6.1 presents a quantitative analysis of terpenes presents in essential oil extracted from black pepper dried under different drying conditions. Moreover, Table 2 in Appendix E presents the percentage of specific constituents in terpenes in essential oil extracted from black pepper dried at different drying conditions. In addition, Figure 6.1 shows the mean percentage of components in black pepper oil.

Oxygenated terpenes are the minor constituents consisting average percentage of 13-23% of total oil for all five different drying conditions. Terpenes and terpene oxide are present up to 97-99.9% of total oil while remaining components are in trace amounts.

Table 6.2 lists out the concentration and the formula of the ten most abundant components of black pepper essential oil which were Caryophyllene (14.79 %), D limonene(13.72 %), Caryophyllene oxide(11.53 %),  $\alpha$ -Copaene (8.19 %), (1R)- $\alpha$ -Pinene(6.98 %),  $\alpha$ -Pinene(6.85 %),  $\alpha$ -Cubebene (6.43 %), L- $\beta$ -Pinene(5.71 %), Sabinene(5.61 %) and  $\delta$ 3-p-Menthene(4.83 %) under this investigation.

Abundant sesquiterpene present in black pepper essential oil was found to be caryophyllene and this factor has been also confirmed by Parthasarathy et. al. [119]. D limonene is the abundant monoterpene present in black pepper essential oil under our investigation however many literature reports that  $\alpha$ -pinene and  $\beta$  pinene as the dominant monoterpene components in black pepper essential oil [95, 96].

	% percentage of components						
Sample No	Drying condition	monoterpene	oxygeneted monoterpene	sesquiterpene	oxygenated sesquterpene		
1	Sun dried	41.79	15.12	30.89	5.60		
2	Sun dried	50.56	7.22	32.69	4.56		
3	Sun dried	37.11	14.02	24.36	8.17		
4	Sun dried	71.47	1.48	25.15	1.79		
5	45 °C	43.22	6.48	37.90	5.64		
6	45 °C	66.27	1.95	27.78	1.19		
7	45 °C	13.33	16.08	24.73	39.03		
8	45 °C	50.76	1.29	39.16	7.37		
9	55 °C	24.95	15.32	27.51	9.59		
10	55 °C	40.49	10.63	33.38	4.91		
11	55 °C	48.59	6.06	38.84	5.26		
12	55 °C	42.14	9.69	32.53	7.00		
13	65 °C	49.96	1.63	34.69	11.13		
14	65 °C	43.82	9.25	37.30	3.74		
15	65 °C	49.64	7.78	32.73	6.48		
16	65 °C	46.56	5.96	36.82	4.97		
17	75°C	50.26	2.51	42.80	2.58		
18	75 °C	42.75	7.77	39.20	2.59		
19	75 °C	55.81	1.30	41.40	1.38		
20	75 °C	27.89	11.82	46.94	10.15		

Table 6.1: Constituents of essential oil extracted from black pepper dried under different drying conditions



Figure 6.1: Percentage of mean sesquiterpenes, monoterpenes and oxygenated compounds in Black pepper essential oil at different drying conditions

Component	Concentration %	Formula
	(w/w)	[120]
Caryophyllene	14.79	C <sub>15</sub> H <sub>24</sub>
D limonene	13.72	C <sub>10</sub> H <sub>16</sub>
Caryophyllene oxide	11.53	C <sub>15</sub> H <sub>24</sub> O
α-Copaene	8.19	C <sub>15</sub> H <sub>24</sub>
(1R)- α -Pinene	6.98	C <sub>10</sub> H <sub>16</sub>
α-Pinene	6.85	C <sub>10</sub> H <sub>16</sub>
α-Cubebene	6.43	C <sub>15</sub> H <sub>24</sub>
L-β-Pinene	5.71	C <sub>10</sub> H <sub>16</sub>
Sabinene	5.61	C <sub>10</sub> H <sub>16</sub>
δ3-p-Menthene	4.83	$C_{10}H_{16}$

Table 6.2: Ten Most Abundant Components of Black Pepper Essential Oil

Average percentage of sesquiterpene compounds have increased from 32% to 43% of total oil when temperature was increased from 45 °C to 75 °C as shown in Figure 6.1. Sesquiterpenes have higher molecular weights and they are larger molecules compared to the monoterpenes. Therefore, sesquiterpenes' ability to leach or transfer from black pepper particle to water is low compared to monoterpene [115]. Further, their boiling points are high and volatility is less than those of monoterpenes. Therefore they are not able to vaporize easily through distillation processes as monoterpenes [121]. Hence, average percentage of monoterpene is always higher than the percentage of sesquiterpene in the extracted essential oil at every drying condition.

Moreover, raising the drying air temperature causes the rupture of cells and increases the permeability of the skin and give more ability to release high percentage of sesquiterpenes during extraction process [122].

According, Figure 6.1 the percentage of oxygenated compounds present in the essential oil is highest at 55 °C. The residence time in the dryer and the rate of oxidation reaction of terpenes are some of the factors, which affect the percentage of oxygenated compounds available in black pepper essential oil. Usually the rate of reaction is substantially low at lower temperatures. Further low temperature drying processes have longer residence times as described in Chapter 5 thus providing more time for oxidization and vice versa. Depending on these two factors, essential oil derived from black pepper dried at 55 °C drying air temperature shows higher percentage of oxygenated compounds compared to those of other drying conditions. Table 3 and Table 4 in Appendix E show more information about the ten most components in black pepper essential oil.

#### 6.2 Analysis of variance

One way analysis of variance (ANOVA) test in IBM SPSS Statistics 20 were conducted to evaluate the significance of the differences among the essential oil extracted from black pepper dried at different drying conditions. One way ANOVA test was used to compare the following components at five different drying conditions:

- Monoterpene concentration
- Sesquiterpene concentration
- oxygenated terpenes concentration
- Caryophyllene concentration
- essential oil yield

Five different drying conditions and mean of above mentioned components of essential oil were considered as the independent variable and dependent variables respectively. Then null hypothesis ( $H_0$ ) and alternative hypothesis ( $H_1$ ) were defined as follows:

$$H_{0}: \mu_{45} = \mu_{55} = \mu_{65} = \mu_{75} = \mu_{sun \text{ dried}}$$
$$H_{1}: \mu_{45} \neq \mu_{55} \neq \mu_{65} \neq \mu_{75} \neq \mu_{sun \text{ dried}}$$

Where,  $\mu$  represents the mean of the selected component.  $\mu_{45}$  represents the mean of selected component of essential oil extracted from black pepper dried at 45 °C drying temperature. Confident interval was set as 0.05. If the significance level is less than 0.05, then there are statistical significant differences between group hence null hypothesis is rejected and alternative hypothesis is accepted.

According to the statistical analysis of the sesquiterpene concentration values obtained by black pepper dried at five different conditions showed that the variation of sesquiterpene concentration was significant. As per ANOVA test, p-value was 0.007. However, a significant level for monoterpene analysis was greater than 5% of confidence level, and thus variation of concentration of monoterpenes was not significant for all the drying air temperatures under investigation. Furthermore, variation of oxygenated terpenes concentration was not significant with variation of drying air temperature. Table 5 in Appendix E presents the results of analysis of variance for comparative study of sesquiterpenes, monoterpenes and oxygenated terpenes with drying conditions.

#### 6.3 Effect of drying condition on Caryophyllene concentration

Caryophyllene also known as beta-Caryophyllene is the primary sesquiterpene contributing to the spiciness of black pepper. It is also the major constituent of cloves [123]. Effect of drying condition on concentration of caryophellene is studied. Figure 6.2 shows the percentage of caryophyllene and caryophyllene oxide in essential oil derived from black pepper dried at different conditions. The results show that the quality of each samples of black pepper dried in sun and at 45 °C drying air temperature were not consistent. The intensity of solar radiation is an uncontrollable factor and intensive sunlight affects the quality of the pepper. Furthermore, at low drying air temperatures, residence time in the dryer is lengthy which gives more opportunity to oxidize and isomerise the components in essential oil.



Figure 6.2: Percentage of Caryophyllene and percentage of caryophyllen oxide vs. drying conditions

As shown in Figure 6.2, the percentage of Caryophyllene was strongly increased when drying air temperature is raised from 55 °C to 75 °C. Percentage of Caryophyllene is moderately high at 45°C drying air temperature despite the fact that the value is not consistent within the investigations. However, the percentage of caryophyllene is consistent at 65°C drying air temperature. Moreover, analysis of variance for concentration of caryophyllene among drying air temperatures was not significant. In addition to the Caryophyllene, there were alpha Caryophyllene. It was non-significant because isomerization and oxides derived from caryophyllene. It was non-significant because isomerization and oxidation of terpenes may take place during the drying process. Similar observations have been discussed for thyme and rosemary by Piga et al. [124]. Table 6 in Appendix E presents the results of analysis of variance for Caryophyllene concentration with drying conditions.

#### 6.4 Effect of drying condition on essential oil yield

The oil yield was calculated as volume of oil extracted per weight of dried black pepper used. According to Figure 6.3, average oil yield was increased from 0.0099 mL/g to 0.0174 mL/g when drying air temperature increased from 45  $^{\circ}$ C to 65  $^{\circ}$ C. Then the average oil yield was reduced to 0.0104 mL/g at 75  $^{\circ}$ C drying temperature. The oil yield for sun-dried black pepper is recorded as 0.0101 mL/g.

SLSI standards does not remark about volatile oil content of black pepper, but other associations such as ESA and ASTA stated that minimum content of volatile matter should be 2 % v/w. To benchmark the globally recognized black pepper quality, variation of essential oil yield with drying conditions is shown Table 9 in Appendix A where oil yield is presented as a percentage of millilitre of oil extracted per gram (V/W %) of black pepper used.

Analysis of variance was conducted for variation of oil yield with drying conditions and results are presented in Table 7 in Appendix E. According to the ANOVA test, effect of drying temperature on the essential oil yield was statistically significant under investigation. The p-value was 0.046. Therefore, optimum drying air temperature to obtain the highest oil yield was selected as 65 °C.



Figure 6.3: Mean oil yield vs. drying conditions



Figure 6.4:-Black pepper essential oil yield vs. drying conditions

Furthermore, 65  $^{\circ}$ C drying air temperature provides consistent quality essential oil with high percentage of caryophyllene and 65  $^{\circ}$ C can be selected as the optimum

temperature to dry black pepper in the conventional spouted bed dryer without degrading large amount of volatile components.

## **Chapter specific nomenclature**

H<sub>1</sub> - alternative hypothesis

H<sub>o</sub> - null hypothesis

 $\mu$  - mean of the selected component

## Abbreviations

ANOVA - analysis of variance

ASTA - American Spice Trade Association

ESA - Europe Spice Association

GC-MS - Gas Chromatography Mass Spectrometry

## Chapter 7

### 7. NON DIMENSIONAL MODEL

This chapter describes the modelling of heat transfer characteristics of black pepper dried in conventional spouted bed dryer in unsteady state. Heat transfer coefficients for different operational conditions were determined from the experimental results. Important non-dimensional numbers were identified by dimensionless analysis according to the Buckingham pi theorem [104]. A non-dimensional model was developed for heat transfer process by non-linear multiple regression. Model development, comparison of experimentally determined heat transfer coefficient and Nusselt number with predicted values from existing correlations and the correlation developed in this work are presented in this chapter.

### 7.1 Dimensional analysis for heat transfer process

As described in section 3.2.2, the dimensional matrix was formed for selected important variables of the heat transfer process of black pepper drying in conventional spouted bed. Table 7.1 shows the dimensional matrix related to heat transfer process.

Table 7.1: Dimensional matrix of the fundamental dimensions related to heat transf	fer
process	

	U	Η	d <sub>p</sub>	$\mu_g$	kg	T <sub>gi</sub>	T <sub>gi</sub> -T <sub>gi,w</sub>	ρ	h <sub>p</sub>
М	0	0	0	1	0	0	0	1	0
L	1	1	1	-1	-1	0	0	-3	-2
t	-1	0	0	-1	-1	0	0	0	1
Q	0	0	0	0	1	0	0	0	1
Т	0	0	0	0	-1	1	1	0	-1

Where

d<sub>p</sub> - particle diameter (m)

H - Static bed height (m)

 $h_p$  - overall heat transfer coefficient fluid to particle(W/m<sup>2</sup>K)

 $k_g$  -Air thermal conductivity (W/mK)

L, M, Q, T, t - Dimensions of length, mass, heat, temperature and time respectively

 $T_{gi}$ - inlet gas temperature (°C)

 $T_{gi,w}$  - gas inlet wet bulb temperature (°C)

U - Superficial air velocity (m/s)

 $\rho$  - Density of fluid/air (kg/m<sup>3</sup>)

 $\mu_g$  - viscosity of air (kg/ms)

Number of variables involved (n') is 9 and the rank of the dimensional matrix (r') is 5. According to Buckingham pi theorem, number of independent dimensionless groups 'i'' is equal to n'-r' [104].

Thus, i'=n'-r'

i'=4

Then four dimensionless parameters namely *pi* groups are symbolized as  $\pi_1$ ,  $\pi_2$ ,  $\pi_3$  and  $\pi_4$ . Group of *r'* number of variables were chosen as core group which consisted of those variables that would be appeared in each *pi* group and other variables were named as non-core group. Usually several fundamental dimensions are included in the variables of core group than the variables of non-core group. Nevertheless thermal conductivity of air was excluded from the core group arbitrarily to isolate its effect [104].

Core group of variables:  $h_p$ ,  $\rho$ ,  $\mu_g$ ,  $T_{gi}$ ,  $d_p$ ,

Non- core group of variables: U, H, kg, Tgi-Tgi,w

Each *pi* group includes variables in core group and one of non-core group. In order to become each dimensionless, variable are raised to certain exponents such as:

$$\pi_{1} = h_{p}{}^{a} \rho^{b} \mu_{g}{}^{c} d_{p}{}^{d} T_{gi}{}^{e} U$$
$$\pi_{2} = h_{p}{}^{f} \rho^{g} \mu_{g}{}^{h} d_{p}{}^{i} T_{gi}{}^{j} H$$
$$\pi_{3} = h_{p}{}^{k} \rho^{l} \mu_{g}{}^{m} d_{p}{}^{n} T_{gi}{}^{p} k_{g}$$
$$\pi_{4} = h_{p}{}^{q} \rho^{r} \mu_{g}{}^{s} d_{p}{}^{u} T_{gi}{}^{w} \Delta T$$

Each *pi* group was arranged using dimensions then each *pi* group was considered independently to evaluate the exponent of as follows:

For 
$$\pi_1$$
  $0 = \left(\frac{Q}{L^2 tT}\right)^a \left(\frac{M}{L^3}\right)^b \left(\frac{M}{Lt}\right)^c (L)^d (T)^e \left(\frac{L}{t}\right)^c$ 

Exponent of each fundamental dimension in both sides of the expression are equated.

For L For M 0 = -2a - 3b - c + d + 1For M 0 = b + cFor t 0 = -a - c - 1For T 0 = -a + eFor Q 0 = aFrom these a = 0; b = 1; c = -1; d = 1; e = 0

Thus first dimensionless group is

$$\pi_1 = \frac{U\rho d_p}{\mu_g} = Re$$

Similarly

For 
$$\pi_2$$
  $0 = \left(\frac{Q}{L^2 tT}\right)^f \left(\frac{M}{L^3}\right)^g \left(\frac{M}{Lt}\right)^h (L)^i (T)^j L$ 

- For L 0 = -2f 3g h + i + 1
- For M 0 = g + h
- For t 0 = -f h

- For T 0 = -f + j
- For Q 0 = f

From these

$$f = 0; g = 0; h = 0; i = -1; j = 0$$

Thus second dimensionless group is

$$\pi_{2} = \frac{H}{d_{p}} = \left(\frac{\text{static bed height}}{\text{particle diamter}}\right)$$

Similarly

For 
$$\pi_3$$
  $0 = \left(\frac{Q}{L^2 tT}\right)^k \left(\frac{M}{L^3}\right)^l \left(\frac{M}{Lt}\right)^m (L)^n (T)^p \left(\frac{Q}{LtT}\right)$ 

similarly

- For L 0 = -2k 3l m + n 1
- For M 0 = l + m
- For t 0 = -k m 1
- For T 0 = -k + p 1
- For Q 0 = k + 1
- From these k = -1; l = 0; m = 0; n = -1

Thus third dimensionless group is

$$\pi_3 = \frac{k_g}{h_p d_p} = \frac{1}{Nu}$$

Nu- Nusselt number

For 
$$\pi_4$$
 
$$0 = \left(\frac{Q}{L^2 tT}\right)^q \left(\frac{M}{L^3}\right)^r \left(\frac{M}{Lt}\right)^s (L)^u (T)^w T$$

Similarly

For L 0 = -2q - 3r - s + u

For M 0 = r + sFor t 0 = -q - sFor T 0 = -q + w + 1For Q 0 = q

From these w = -1; u = 0; r = 0; q = 0; s = 0

Thus fourth dimensionless group is

$$\pi_4 = \frac{T_{gi} - T_{gi,w}}{T_{gi}} = Gu$$

Gu- Gukhman number

The results of the dimensional analysis of heat transfer process of black pepper drying in spouted bed dryer indicate that a possible relation correlating the important variable is of the form of:

$$Nu = f\left(Re, Gu, \frac{H}{d_p}\right)$$

#### 7.2 Experimental results

Drying experiments were conducted as described in section 3.1.6.3. 48 experiments were conducted. Heat transfer coefficients ( $h_p$ ) for 48 experiments with different operating conditions were obtained by solving Equation 3.13. Heat transfer coefficients thus calculated and their operating conditions are presented in Table 7.2.

The analysis of heat transfer indicates that over the operating conditions studied, heat transfer coefficient ( $h_p$ ) for black pepper drying in conventional spouted bed ranges from 35-68 W/m<sup>2</sup>K. Dimensionless parameter Nusselt (Nu) number was calculated and the particle Nu number for the black pepper drying in conventional spouted bed ranges from 6.69 to 11.8. Dimensionless numbers, heat transfer coefficients and their operating conditions are presented in Table 1 in Appendix F.

Table 7.2: Operating conditions, heat transfer coefficient for black pepper dried in conventional spouted bed

Oper	h <sub>p</sub>		
Н	T <sub>gi</sub>	U	$(W/m^2K)$
( <b>cm</b> )	(°C)	(m/s)	
14	45	2.37	49.00
14	55	2.37	45.00
14	65	2.37	54.67
14	75	2.37	57.24
14	65	1.60	51.69
14	65	1.97	59.09
7	65	2.37	65.93
18	65	2.37	51.09
20	65	2.37	46.96
20	75	2.37	47.90
20	55	2.37	47.16
22	75	2.37	45.64
20	45	2.37	40.80
14	45	2.37	51.09
18	45	2.37	53.82
18	55	2.37	43.50
18	55	2.37	53.68
14	45	1.60	47.09
14	60	2.37	57.50
18	75	2.37	56.29
18	75	1.60	53.51
22	55	2.37	45.28
22	45	2.37	40.86

# Table 7.2. (Continued)

Oper	h <sub>p</sub>		
Н	T <sub>gi</sub>	U	$(W/m^2K)$
( <b>cm</b> )	( °C)	(m/s)	
22	60	1.60	40.18
14	35.3	2.37	34.50
24	75	2.37	52.40
24	65	2.37	48.37
22	65	1.60	44.86
24	65	2.57	48.95
24	55	2.57	47.27
24	55	2.37	47.91
7	65	1.60	65.10
7	75	2.37	67.18
7	45	2.06	58.14
7	55	2.33	63.93
7	55	1.72	61.80
7	65	1.72	63.14
7	65	2.54	67.60
7	75	2.49	69.47
7	65	2.06	67.75
14	60	1.60	51.49
14	60	2.54	54.99
14	65	2.59	56.80
7	36	1.68	49.64
7	40	2.15	57.13
14	40	2.20	45.60
18	65	2.49	57.65
18	65	2.06	48.91

#### 7.3 Correlation for heat transfer process.

Important non dimensional numbers were identified for the heat transfer process of black pepper drying in the conventional spouted bed using dimensional analysis as shown in section 7.1. Non-dimensional numbers were Nusselt (Nu), Reynolds (Re), Gukhman (Gu) and ratio of static bed height to particle diameter  $\left(\frac{H}{d_p}\right)$ . Heat transfer coefficient represented by Nu with Re, Gu and ratio of static bed height to particle diameter  $\left(\frac{H}{d_p}\right)$  was correlated to a power form of function as shown in Equation 7.1 using statistical software IBM SPSS Statistics 20.

$$Nu = a' R e^{b'} G u^{c'} \left(\frac{H}{d_p}\right)^{d'}$$
(7.1)

The values of the constant 'a'' and exponents 'b'', 'c'' and 'd'' were determined by the method of multiple non-linear regression of experimental results and the equation thus determined is shown in Equation 7.2

Nu = 12.057Re<sup>0.137</sup>Gu<sup>0.180</sup> 
$$\left(\frac{H}{d_p}\right)^{-0.293}$$
 (7.2)

The correlation for heat transfer is yielded with 0.791 of coefficient of determination  $(R^2)$  with significance level 0.000 and valid for the following conditions.

 $400{\leq}\,Re{\leq}750$ 

 $0.1{\leq}\,Gu{\leq}0.6$ 

 $13 \le H/d_p \le 47$ 

The correlation is valid for a geometrically similar system of conventional spouted bed. Geometric similarity exists between the prototype (scale up spouted bed) and the model (spouted bed used for the investigation), if both of them are identical in shape but different only in size. In addition, ratio of all the linear dimensions should be equal [13,104]. Geometric similarity of the prototype can be described as follows:

- Ratio of column diameter (D<sub>c</sub>) to gas inlet diameter (D<sub>i</sub>) of the conventional spouted bed is equal to three.
- Cone angle of the spouted bed base is  $60^{\circ}$ .
- Ratio of column height (height of cylindrical section of the spouted bed column) to column diameter is 100/15.

Detailed information on statistical analysis is shown in Appendix G and information on dimensionless numbers and correlations for determining density, viscosity and thermal conductivity of air are stated in Appendix H.

The heat transfer coefficient;  $h_p$  was varied over the range of values studied from 35-68 W/m<sup>2</sup>K under the investigation. In addition, Nu number varied with 6.5 to 11.8. However Prachayawarakorn et al. found that heat transfer coefficient for spout region varied from 57 to 123 W/m<sup>2</sup>K while heat transfer coefficient of downcomer(annulus) zone varied in between 19 and 32 W/m<sup>2</sup>K in two dimensional spouted bed of paddy [39]. The reasons for these contradictory results are the difference of material being dried; paddy and black pepper while this study analyses the heat transfer process considering the whole bed including spout, annulus and fountain regions in a conventional spouted bed.

The comparison of experimental heat transfer coefficient and predicted heat transfer coefficient from Equation 7.2 is shown in Figure 7.1. Response surfaces of predicted Nu number as a function of Gu number and  $H/d_p$  at three different Re numbers are shown in Figure 7.2. Figure 7.3 shows the validity of correlated equation (Equation 7.2) and comparison of experimental Nu number values with predicted Nu number from some of existing correlations. Table 2.3 of Chapter 2 shows the existing correlations for heat transfer process of spouled bed drying. Both Figure 7.1 and Figure 7.3 show that predicted values from the developed correlation vs. experimental values data are lying around the straight line. It is seen that experimental heat transfer coefficient and experimental Nu number are in good agreement with predicted values of the developed correlation.



Figure 7.1: Comparison of experimented heat transfer coefficient with predicted heat transfer coefficient



Figure 7.2: Comparison of experimented Nu with response surfaces of predicted Nu as a function of Gu and  $H/d_p$  at three different Re numbers


Figure 7.3: Comparison of experimental Nu with predicted Nu using existing correlations and present work

Existing correlations for evaluation of heat and mass transfer coefficients have been developed considering a particular system of properties such as contactor geometry, geometry of the material being dried (spherical, rectangular slabs) and the characteristics of the fluid being used for heat and mass transfer. In addition, many existing correlations were developed considering a particular material at steady state drying process or constant rate drying periods. Therefore, predicted values from correlations derived by works of Reger et al. and Kmiec et al. are considerably below the experimental values [125,27].

#### 7.4 Effect of operational conditions on heat transfer

The Nu number represents the effectiveness of heat transfer processes. The Nu number is the ratio of heat transferred by convection to the heat transferred by conduction. The Nu number is an important parameter in heat transfer systems for determining the mode of heat transfer and estimating heat transfer coefficient using non dimensional correlations. A System having Nu number equals to unity, interprets that the fluid is stationary and heat is transferred by conduction only. If Nu number is greater than 1 indicates that the fluid motion enhances the heat transfer by

convection. Moreover, Nu number is increased with increasing heat transfer coefficients.

Figure 7.4 and Figure 7.5 present the effect of drying temperature on heat transfer coefficient and Nu number. The trend of Nusselt number and heat transfer coefficient as a function of Re number or Gu number is presented in Figure 7. 6, Figure 7. 7 and Figure 7.10. The values of heat transfer coefficient and Nu number are increased with increasing drying temperature due to increase of driving forces for heat transfer. Temperature difference is the driving force for heat transfer.



Figure 7.4: Effect of drying temperature on heat transfer coefficient for black pepper dried in conventional spouted bed at different drying temperatures; Experiment conditions:  $D_i=0.05$  m, H=0.14 m and U=2.37 m/s



Figure 7.5: Nusselt number as a function of drying temperature for black pepper dried in conventional spouted bed at different drying temperatures; Experiment conditions:  $D_i=0.05$  m, H=0.14 m and U=2.37 m/s



Figure 7. 6: Response surfaces for predicted Nu number as a function of Re number and  $H/d_p$  at different drying air temperatures



Figure 7. 7: Response surfaces for predicted heat transfer coefficient as a function of Re and H/d<sub>p</sub> at different drying air temperatures

Figures 7.8 to 7.13 present the effect of static bed height and superficial air velocity on the both heat transfer coefficient and Nu number. The value of heat transfer coefficient and Nu number increased with drying temperature and air velocity but decreased with increasing static bed height.

According to Figure 7.8 and Figure 7.9, heat transfer coefficient or Nu number has a higher value at the lower static beds. Exponent of  $(H/d_p)$  is negative for the correlation (Equation 7.2) developed for heat transfer processes. Same phenomenal is shown in existing correlations as shown in Table 2.3 of Chapter 2. Figure 7.10 shows the response surface for predicted Nu number as a function of Gu and Re number at different static bed heights.

Low static beds create greater turbulence while spouting, due to the low number of particles in the bed compared to those of deeper beds. Then this greater turbulence caused inside the dryer decreases the gas film thermal resistance around the particle surface, leading to achieve higher heat transfer coefficients [39].



Figure 7.8: Effect of static bed height on heat transfer coefficient for black pepper dried in conventional spouted bed of different bed heights; Experiment conditions:  $D_i=0.05 \text{ m}, T_{gi}=65 \text{ °C} \text{ and } U=2.37 \text{ m/s}$ 



Figure 7.9: Nusselt number as a function of static bed height for black pepper dried in conventional spouted bed of different bed heights; Experiment conditions:  $D_i=0.05 \text{ m}$ ,  $T_{gi}=65 \text{ °C}$  and U=2.37 m/s



Figure 7.10: Response surfaces for predicted Nu number as a function of Gu number and Re number at different static bed heights

The behaviour of the heat transfer coefficient was dependent on air velocity. The values of heat transfer coefficient and Nu number were increased with increasing air velocity as shown in Figures 7.11-7.13. This factor is also interpreted in the correlations developed in present work as shown in Equation 7.2.

However, exponent of Re number in existing correlations; Equation 2.15 to Equation 2.21 (in Table 2.3 of Chapter 2) are more than twice as high as that in the correlation developed for heat transfer by this work; Equation 7.2. This is due to the fact that those correlations were developed for either steady state or constant rate drying processes. Exponent of external parameters are usually high for constant rate drying processes since external parameters such as air velocity, drying temperature, static bed height normally controls the rate of surface evaporation.

Figure 7. 6, Figure 7.10 and Figure 7.13 show that Nu number is increased with increasing Re number and Gu number and decreasing ratio of static bed height to particle diameter. It implies that as a result of an increase in drying air temperature and air velocity and a decrease in bed height, heat and mass transfer between air and the solid particle are increased. However the Re number is less effective than the other two parameters; Gu number and ratio of static bed height to particle diameter on Nusselt number as shown in Figure 7. 6 and Figure 7.10. This is due to the fact that shallow static beds have great turbulence during the spouting operation and higher values of air temperature can cause large driving force for heat transfer while the air velocity or Re number is low. Accordingly use of higher air velocity in a spouted bed dryer is not a wise solution considering the energy and the cost involved.



Figure 7.11- Effect of air velocity and Re number on heat transfer coefficient for black pepper dried in conventional spouted bed at different superficial air velocities; Experiment conditions:  $D_i=0.05 \text{ m}$ ,  $T_{gi}=65 \degree C$  and H=0.14 m



Figure 7.12-: Nusselt number as a function of Reynolds number and air velocity for black pepper dried in conventional spouted bed at different superficial air velocities; Experiment conditions:  $D_i=0.05 \text{ m}$ ,  $T_{gi}=65 \degree C$  and H=0. 14 m



Figure 7.13: Response surfaces for predicted Nu number as a function of Gu number and  $H/d_p$  at different air velocities.

#### **Chapter specific nomenclature**

a, b, c, d, e, f, g, h, i, j, k, l, m, n, p, q, r, s, t, u, w, a', b', c', d' - constants

d<sub>p</sub>- particle diameter (m)

Gu- Gukhman number

H- Static bed height (m)

 $h_p$  – overall heat transfer coefficient fluid to particle(W/m<sup>2</sup>K)

i'- number of independent dimensionless groups

 $k_g$  -Air thermal conductivity (W/mK)

L, M, Q, T, t - Dimensions of length, mass, heat, temperature and time respectively

n' -Number of variables involved

Nu- Nusselt number

r'-rank of the dimensional matrix

Re- Reynolds number

 $T_{gi}$ - inlet gas temperature or drying air temperature (°C)

 $T_{gi,w}$ - gas inlet wet bulb temperature (°C)

U- Superficial air velocity (m/s)

 $\rho$  - Density of fluid/air (kg/m<sup>3</sup>)

 $\mu_g$ - viscosity of air (kg/ms)

### **Chapter 8**

## 8. CONCLUSION & RECOMMENDATIONS FOR FUTURE WORK

#### 8.1 Conclusion

Heat and mass transfer in drying of black pepper in an unsteady state spouted bed dryer was analysed. Experiments were conducted using wetted black pepper and raw black pepper.

Wetted black pepper was used for conducting preliminary experiments in three different spouted bed configurations namely conventional spouted bed, spouted bed equipped with non-porous draft tube and spouted bed equipped with porous draft tube. Spouted bed equipped with non-porous draft tube had the lowest value of minimum spouting velocity followed by the spouted bed equipped with porous draft tube while the conventional spouted bed had the highest minimum spouting velocity for a given bed heights.

Conventional spouted bed required shorter drying time, higher drying rate and higher moisture diffusivity compared to those of spouted beds equipped with draft tubes. Therefore, the conventional spouted bed dryer was identified as the optimum configuration of spouted bed among the spouted bed equipped with draft tubes to continue drying experiments with raw black pepper to achieve the objective of this research.

Main experiments were conducted using raw black pepper in the conventional spouted bed dryer. Minimum spouting velocity of conventional spouted bed was varied from 0.86-1.86 m/s for 0.07-0.22 m stagnant bed height of raw black pepper. Minimum spouting velocity of black pepper particles were increased with increasing static bed height.

Data of minimum spouting velocity of raw black pepper in the conventional spouted bed versus static bed height was correlated to a power form of function and the relationship has 0.97 of coefficient of determination. Equation 5.2 recalls the function of minimum spouting velocity of raw black pepper in the conventional spouted bed versus static bed height.

 $U_{\rm ms} = 5.26 \ {\rm H}^{0.67}$ 

The higher values of drying rates were achieved when the dryer was operated with moderately high temperatures, high airflow rates and shallow beds. Drying kinetic data of raw pepper drying in the conventional spouted bed dominates falling rate periods.

Effective moisture diffusivity of black pepper increased with increasing drying temperature. The highest effective moisture diffusivity of  $1.55 \times 10^{-10}$  m<sup>2</sup>/s was obtained at a hot air temperature of 75 °C. Activation energy for drying of raw black pepper in conventional spouted bed was 42.41 kJ/kmol.

Kinetic data obtained from experiments were fitted to five thin layer drying models namely Newton, Page, Henderson and Pabis, Two compartments, and Logarithmic models. According to the statistical analysis, Logarithmic model show the best results for the drying behaviour of black pepper in conventional spouted bed dryer due to highest coefficient of determination and lowest root mean square error and sum of square error values. In addition, four models were developed for black pepper drying in the conventional spouted bed by correlating drying constants and coefficients of Logarithmic model to stagnant bed height only at specified temperatures; 45 °C, 55 °C, 65 °C, and 75 °C. Developed models can be used to estimate the drying time of black pepper in the conventional spouted bed dryer for given moisture reduction in the ranges of 0.14 -0.22 m stagnant bed heights and in 2.37 m/s air velocity at specified drying air temperatures.

Specific Energy Consumption (SEC) defined as the amount thermal energy supplied to evaporate 1 kg of moisture in drying of black pepper was calculated. SEC values were varied from 16 - 55 MJ/kg of moisture evaporated for stagnant bed height varied from 7 cm to 20 cm at the range of drying air temperature of 45 °C to 75 °C.

SEC increased with increasing air velocity and decreased gradually with increasing static bed height.

The analysis of components in black pepper essential oil shows that the black pepper essential oil comprised mainly monoterpenes and sesquiterpenes. 13-66% of total oil was monoterpenes while 24-47% of total oil was sesquiterpenes. The most abundant sesquiterpene component was Caryophyllene (14.79%) while most abundant monoterpene was D limonene (13.72%). Average percentage of sesquiterpene compounds have increased from 32% to 43% of total oil when drying air temperature was increased from 55 °C to 75 °C. Variation of sesquiterpene concentration in black pepper essential oil was significant while variations of monoterpene, oxygenated terpenes and caryophyllene concentration were non-significant in black pepper essential oil with drying air temperature.

Average oil yield was varied from 0.0099 mL/g to 0.0174 mL/g when drying air temperature increased from 45  $^{\circ}$ C to 65  $^{\circ}$ C. Then the average oil yield was reduced to 0.0104 mL/g at 75  $^{\circ}$ C drying temperature. The variation of oil yield is statistically significant with drying air temperatures from 45  $^{\circ}$ C to 75  $^{\circ}$ C. 65  $^{\circ}$ C drying air temperature provides consistent quality essential oil with high percentage of caryophyllene and 65  $^{\circ}$ C is selected as the optimum temperature to dry black pepper in the conventional spouted bed dryer to achieve higher yield of essential oil without degrading large amount of volatile components.

Therefore black pepper dried at 65 °C temperature in spouted bed dryer is suitable to use in perfumery industry since it had highest percentage of sesquiterpenes and minimum loss of volatile components compared to other drying temperatures.

Heat transfer coefficient was varied from 35-68  $W/m^2$  for different drying conditions of unsteady state drying of black pepper from initial moisture content to final moisture content of 15% dry basis.

Significant dimensionless numbers were Nusselt, Reynolds, Gukhman and ratio of static bed height to particle diameter according to analysis of the Buckingham pi theorem . A correlations was developed for heat transfer as a function of dimensionless groups with 0.791 of coefficient of determination and 0.000 of

significance level. The correlation is valid for a system of conventional spouted bed geometric similar to the unit under investigation. Equation 7.2 recalls the developed correlation.

$$Nu = 12.057 \text{Re}^{0.137} \text{Gu}^{0.180} \left(\frac{\text{H}}{\text{d}_{p}}\right)^{-0.293}$$

Heat transfer coefficients predicted from the developed correlation show a good agreement with the experimentally determined heat transfer coefficients.

#### 8.2 Recommendations for future work

This research outcome provides recommendations for conventional spouted bed dryer for black pepper drying. Furthermore investigations can be carried out to improve the usability of two dimensional(slot -rectangular) spouted bed and two dimensional spouted bed with porous draft plates for black pepper or any other type of agricultural product. Oleoresins of black pepper consists mainly pungent components which can be used in food industry. Work presented in Chapter 6 can be extended to find the optimum drying conditions to achieve consistent quality black pepper with higher yield of either essential oil or the oleoresins.

Then a CFD model can be developed to simulate and describe the drying behaviour and the hydrodynamics of the unsteady state drying process of spouted bed dryer.

#### REFERENCES

- [1] "Pepper." [Online]. Available: http://www.exportagridept.gov.lk/web/index.php?option=com\_content&view =article&id=137&Itemid=159&lang=en. [Accessed: 18-Oct-2018].
- [2] H. G. Kim *et al.*, "Piperine inhibits PMA-induced cyclooxygenase-2 expression through downregulating NF-κB, C/EBP and AP-1 signaling pathways in murine macrophages," *Food Chem. Toxicol.*, vol. 50, no. 7, pp. 2342–2348, 2012.
- [3] M. H. Mehmood and A. H. Gilani, "Pharmacological basis for the medicinal use of black pepper and piperine in gastrointestinal disorders," *J. Med. Food*, vol. 13, no. 5, pp. 1086–1096, Oct. 2010.
- [4] S.-H. Kim and Y.-C. Lee, "Piperine inhibits eosinophil infiltration and airway hyperresponsiveness by suppressing T cell activity and Th2 cytokine production in the ovalbumin-induced asthma model," *J. Pharm. Pharmacol.*, vol. 61, no. 3, pp. 353–359, 2009.
- [5] P. Borges and J. Pino, "Preparation of black pepper oleoresin by alcohol extraction," *Food/Nahrung*, vol. 37, no. 2, pp. 127–130, 1993.
- [6] M. McCarron, A. Mills, D. Whittaker, T. Kurian, and J. Verghese, "Comparison between green and black pepper oils from Piper nigrum L. berries of Indian and Sri Lankan origin," *Flavour Fragr. J.*, vol. 10, no. 1, pp. 47–50, 1995.
- [7] "The Spice Council of Sri Lanka." [Online]. Available: http://www.srilankanspices.com/sl\_spices\_pepper.html. [Accessed: 23-Dec-2017].
- [8] "European Spice Association Quality Minima Document Rev. 5." Adopted at the meeting of the Technical Commission on 27th October 2015, Oct-2015.
- [9] K. J. Chua and S. K. Chou, "Low-cost drying methods for developing countries," *Trends Food Sci. Technol.*, vol. 14, no. 12, pp. 519–528, 2003.

- [10] "National Priorities in Postharvest & Value Addition Research in Agriculture 2011-2015." [Online]. Available: https://www.slcarp.lk. [Accessed: 06-May-2014].
- [11] "National Priorities in on Agricultural Mechanization 2017-2021." [Online]. Available: http://www.slcarp.lk/research-priorities-next/. [Accessed: 20-Sep-2019].
- [12] J. F. Richardson, J. K. Harker, and Backhurst, *Chemical Engineering Volume* 2 - 5th Edition.
- [13] K. M. Norman Epstein, Spouted Beds. A c a d e m i c p r e s s, 1974.
- [14] Treybal R.E., Mass-transfer Operations, Third. McGraw-Hill, 1980.
- [15] S. C. Digvir S. Jayas, "Review of thin-layer drying and wetting equations," *Dry. Technol.*, vol. 9, pp. 551–588, 1991.
- [16] S. Azzouz, A. Guizani, W. Jomaa, and A. Belghith, "Moisture diffusivity and drying kinetic equation of convective drying of grapes," *J. Food Eng.*, vol. 55, no. 4, pp. 323–330, 2002.
- [17] M. Markowski, I. Bia\lobrzewski, and A. Modrzewska, "Kinetics of spouted-bed drying of barley: Diffusivities for sphere and ellipsoid," *J. Food Eng.*, vol. 96, no. 3, pp. 380–387, 2010.
- [18] S. J. Babalis and V. G. Belessiotis, "Influence of the drying conditions on the drying constants and moisture diffusivity during the thin-layer drying of figs," *J. Food Eng.*, vol. 65, no. 3, pp. 449–458, 2004.
- [19] H. Darvishi, A. R. Asl, A. Asghari, M. Azadbakht, G. Najafi, and J. Khodaei, "Study of the drying kinetics of pepper," *J. Saudi Soc. Agric. Sci.*, vol. 13, no. 2, pp. 130–138, 2014.
- [20] S. G. Agrawal and R. N. Methekar, "Mathematical model for heat and mass transfer during convective drying of pumpkin," *Food Bioprod. Process.*, vol. 101, pp. 68–73, 2017.
- [21] R. A. Chayjan, K. Salari, Q. Abedi, and A. A. Sabziparvar, "Modeling moisture diffusivity, activation energy and specific energy consumption of squash seeds in a semi fluidized and fluidized bed drying," *J. Food Sci. Technol.*, vol. 50, no. 4, pp. 667–677, 2013.
- [22] "Drying of Foods Vegetables and Fruits Volume 1.pdf.".

- [23] S. D. George, S. Cenkowski, and W. E. Muir, "A review of drying technologies for the preservation of nutritional compounds in waxy skinned fruit," in North Central ASAE/CSAE Conf, Winnipeg, Manitoba, Canada, 2004, vol. 2425.
- [24] J. CM, G. P. Pittappillil, and P. JOSE, "Drying of black pepper (Piper nigrum L.) using solar tunnel dryer," *TROPICAL*, vol. 25, no. 1, p. 39, 2002.
- [25] "Spice Processing (Practical Action Technical Brief) Appropedia: The sustainability wiki." [Online]. Available: https://www.appropedia.org/Spice\_Processing\_(Practical\_Action\_Technical\_Brief). [Accessed: 01-Jun-2019].
- [26] "Pepper Cultivation and Pocessing -Technichal Bulletin 4." The Department of Export Agriculture, Ministry of Agriculture, Lands and Forestry, 2015.
- [27] N. Epstein and J. R. Grace, Spouted and Spout-Fluid Beds: Fundamentals and Applications. Cambridge University Press, 2011.
- [28] M. S. Bacelos, M. L. Passos, and J. T. Freire, "Characteristics of flow in wet conical spouted beds of unequal-sized spherical Particles," *Braz. J. Chem. Eng.*, vol. 25, no. 1, pp. 27–38, 2008.
- [29] M. L. Passos, A. S. Mujumdar, and V. S. G. Raghavan, "Prediction of the maximum spoutable bed height in two-dimensional spouted beds," *Powder Technol.*, vol. 74, no. 2, pp. 97–105, 1993.
- [30] I. Białobrzewski, M. Zielińska, A. S. Mujumdar, and M. Markowski, "Heat and mass transfer during drying of a bed of shrinking particles – Simulation for carrot cubes dried in a spout-fluidized-bed drier," *Int. J. Heat Mass Transf.*, vol. 51, no. 19–20, pp. 4704–4716, Sep. 2008.
- [31] P. P. Ng, C. L. Law, and S. M. Tasirin, "Drying Kinetics of Malaysian Paddy (Group D Particles) in Spouted Bed Dryer," *Int. J. Food Eng.*, vol. 2, no. 3, Jan. 2006.
- [32] S. Wiriyaumpaiwong, S. Soponronnarit, and S. Prachayawarakorn, "Soybean drying by two-dimensional spouted bed," *Dry. Technol.*, vol. 21, no. 9, pp. 1735–1757, 2003.

- [33] M. I. Kalwar and G. S. V. Raghavan, "Batch drying of shelled corn in twodimensional spouted beds with draft plates," *Dry. Technol.*, vol. 11, no. 2, pp. 339–354, 1993.
- [34] M. Olazar, G. Lopez, H. Altzibar, M. Amutio, and J. Bilbao, "Drying of biomass in a conical spouted bed with different types of internal devices," *Dry. Technol.*, vol. 30, no. 2, pp. 207–216, 2012.
- [35] Z. Wang, H. T. Bi, C. J. Lim, and P. Su, "Determination of minimum spouting velocities in conical spouted beds," *Can. J. Chem. Eng.*, vol. 82, no. 1, pp. 11– 19, 2004.
- [36] M. Olazar, G. Lopez, H. Altzibar, and J. Bilbao, "Modelling batch drying of sand in a draft-tube conical spouted bed," *Chem. Eng. Res. Des.*, vol. 89, no. 10, pp. 2054–2062, Oct. 2011.
- [37] M. E. Fayed and L. Otten, *Handbook of powder science & technology*. New York: Chapman & Hall, 1997.
- [38] D. Geldart, "Types of gas fluidization," *Powder Technol.*, vol. 7, no. 5, pp. 285–292, May 1973.
- [39] S. Prachayawarakorn, S. Ruengnarong, and S. Soponronnarit, "Characteristics of heat transfer in two-dimensional spouted bed," *J. Food Eng.*, vol. 76, no. 3, pp. 327–333, 2006.
- [40] M. R. Parise, Z. Wang, C. J. Lim, and J. R. Grace, "Hydrodynamics of a slotrectangular spouted bed of biomass particles with simultaneous injection of spouting and pulsating air streams," *Chem. Eng. J.*, vol. 330, pp. 82–91, Dec. 2017.
- [41] Y. Zhang, G. Huang, and G. Su, "Hydrodynamic behavior of silicon particles with a wide size distribution in a draft tube spout-fluid bed," *Chem. Eng. J.*, vol. 328, pp. 645–653, 2017.
- [42] H. Altzibar, G. Lopez, M. J. San José, A. Barona, and M. Olazar, "A Draft-Tube Conical Spouted Bed for Drying Fine Particles," *Dry. Technol.*, Mar. 2008.
- [43] H. Altzibar, G. Lopez, M. Olazar, and J. Bilbao, "Study of the minimum spouting velocity in a draft tube conical spouted bed.," presented at the The

13th International Conference on Fluidization - New Paradigm in Fluidization Engineering, 2011.

- [44] D. Evin, H. Gül, and V. Tanyıldızı, "Grain drying in a paraboloid-based spouted bed with and without draft tube," *Dry. Technol.*, vol. 26, no. 12, pp. 1577–1583, 2008.
- [45] T. Ishikura, H. Nagashima, and M. Ide, "Hydrodynamics of a spouted bed with a porous draft tube containing a small amount of finer particles," *Powder Technol.*, vol. 131, no. 1, pp. 56–65, 2003.
- [46] H. Altzibar, G. Lopez, J. Bilbao, and M. Olazar, "Minimum Spouting Velocity of Conical Spouted Beds Provided with Draft Tubes of Different Configuration," *Ind. Eng. Chem. Res.*, 2013.
- [47] T. Madhiyanon, S. Soponronnarit, and W. Tia, "Continuous drying of paddy in two-dimensional spouted bed," *Kasetsart J. Nat. Sci.*, vol. 34, no. 2, pp. 308–314, 2000.
- [48] C. J. Lim and J. R. Grace, "Spouted bed hydrodynamics in a 0.91 m diameter vessel," *Can. J. Chem. Eng.*, vol. 65, no. 3, pp. 366–372, 1987.
- [49] J. Makibar, A. R. Fernandez-Akarregi, L. Díaz, G. Lopez, and M. Olazar, "Pilot scale conical spouted bed pyrolysis reactor: Draft tube selection and hydrodynamic performance," *Powder Technol.*, vol. 219, pp. 49–58, 2012.
- [50] N. Epstein and J. R. Grace, Spouting of Particulate Solids. Chapter 1 1 in Handbook of Powder Science and Technology EM]. New York: Van Nostrand Reinhold Co, 1997.
- [51] M. Olazar, R. Aguado, M. J. San José, S. Alvarez, and J. Bilbao, "Minimum spouting velocity for the pyrolysis of scrap tyres with sand in conical spouted beds," *Powder Technol.*, vol. 165, no. 3, pp. 128–132, 2006.
- [52] M. Olazar, M. J. San Jose, A. T. Aguayo, J. M. Arandes, and J. Bilbao, "Stable operation conditions for gas-solid contact regimes in conical spouted beds," *Ind. Eng. Chem. Res.*, vol. 31, no. 7, pp. 1784–1792, 1992.
- [53] M. J. San José, S. Alvarez, A. Ortiz de Salazar, M. Olazar, and J. Bilbao, "Operating conditions of conical spouted beds with a draft tube. Effect of the diameter of the draft tube and of the height of entrainment zone," *Ind. Eng. Chem. Res.*, vol. 46, no. 9, pp. 2877–2884, 2007.

- [54] H. Altzibar, G. Lopez, R. Aguado, S. Alvarez, M. J. San Jose, and M. Olazar, "Hydrodynamics of conical spouted beds using different types of internal devices," *Chem. Eng. Technol.*, vol. 32, no. 3, pp. 463–469, 2009.
- [55] M. Olazar, M. J. San José, A. T. Aguayo, J. M. Arandes, and J. Bilbao, "Hydrodynamics of nearly flat base spouted beds," *Chem. Eng. J. Biochem. Eng. J.*, vol. 55, no. 1, pp. 27–37, 1994.
- [56] G. Rovero, C. Massimo, and C. Giuliano, "Optimization of a Spouted Bed Scale-Up by Square-Based Multiple Unit Design," *Adv. Chem. Eng.*, 2012.
- [57] J. H. Harker, J. R. Backhurst, and J. F. Richardson, *Chemical Engineering Volume 2, Fifth Edition*, 5 edition. Oxford: Butterworth-Heinemann, 2002.
- [58] A. Belhamri, "Characterization of the first falling rate period during drying of a porous material," *Dry. Technol.*, vol. 21, no. 7, pp. 1235–1252, 2003.
- [59] R. G. Szafran and A. Kmiec, "CFD modeling of heat and mass transfer in a spouted bed dryer," *Ind. Eng. Chem. Res.*, vol. 43, no. 4, pp. 1113–1124, 2004.
- [60] A. Kmiec, "Simultaneous heat and mass transfer in spouted beds," Can. J. Chem. Eng., vol. 53, no. 1, pp. 18–24, 1975.
- [61] T. Kudra, A. S. Mujumdar, and G. S. V. Raghavan, "Gas-to-particle heat transfer in two-dimensional spouted beds," *Int. Commun. Heat Mass Transf.*, vol. 16, no. 5, pp. 731–741, 1989.
- [62] M. H. El-Nans, S. Rognon, R. Legros, and R. C. Mayer, "Hydrodynamics and mass transfer in a spouted bed dryer," *Dry. Technol.*, vol. 18, no. 1–2, pp. 323–340, 2000.
- [63] S. C. S. Rocha, O. P. Taranto, and G. E. Ayub, "Aerodynamics and heat transfer during coating of tablets in two-dimensional spouted bed," *Can. J. Chem. Eng.*, vol. 73, no. 3, pp. 308–312, 1995.
- [64] M. A. Malek and B. C.-Y. Lu, "Heat transfer in spouted beds," Can. J. Chem. Eng., vol. 42, no. 1, pp. 14–20, 1964.
- [65] F. LAP, "Heat transfer in spouted beds," *Dry. Technol.*, vol. 11, no. 2, pp. 303–317, 1993.
- [66] W. P. Oliveira and J. T. Freire, "Study Of The Simultaneous Heat And Mass Transfer During Drying Of Liquid Materials In Conical Spouted Beds.," in

*Proceedings of the 2nd European Thermal-Sciences and 14th UIT National Heat Transfer Conference*, 1996, pp. 1501–1508.

- [67] S. Englart, A. Kmiec, and A. Ludwinska, "Heat transfer in sprayed spouted beds," *Can. J. Chem. Eng.*, vol. 87, no. 2, pp. 185–192, 2009.
- [68] G. K. Jayatunga and B. Amarasinghe, "Mathematical modeling of drying kinetics of black pepper in a spouted bed dryer with and without non porous draft tubes," in *International Conference on "Trends in Multidisciplinary Business and Economic Research (TMBER 25-26 March 2015)TMBER 25-26 March 2015*, Bangkok, Thailand, 2015, vol. 2, pp. 118–134.
- [69] W. Jittanit, "Modelling of seed drying using a two-stage drying concept," PhD Diss Syd. Aust. Univ. New South Wales, 2007.
- [70] C. R. Duarte, V. V. Murata, and M. A. S. Barrozo, "Experimental and numerical study of spouted bed fluif dynamics," *Braz. J. Chem. Eng.*, vol. 25, no. 1, pp. 95–107, 2008.
- [71] R. G. Szafran, A. Kmiec, and W. Ludwig, "CFD modeling of a spouted-bed dryer hydrodynamics," *Dry. Technol.*, vol. 23, no. 8, pp. 1723–1736, 2005.
- [72] T. Thanit, W. Wiwut, T. Tawatchai, T. Toshihiro, T. Toshitsugu, and Y. Yutaka, "Prediction of gas-particle dynamics and heat transfer in a twodimensional spouted bed," *Adv. Powder Technol.*, vol. 16, no. 3, pp. 275–293, 2005.
- [73] W. SOBIESKI, "Influence of selected Eulerian multiphase model parameters on the simulation results for a spouted bed grain dryer," *Task Q.*, vol. 12, no. 1, pp. 511–536, 2008.
- [74] M. Özdemir and Y. O. Devres, "The thin layer drying characteristics of hazelnuts during roasting," J. Food Eng., vol. 42, no. 4, pp. 225–233, 1999.
- [75] H. Kucuk, A. Midilli, A. Kilic, and I. Dincer, "A review on thin-layer dryingcurve equations," *Dry. Technol.*, vol. 32, no. 7, pp. 757–773, 2014.
- [76] A. Midilli and H. Kucuk, "Mathematical modeling of thin layer drying of pistachio by using solar energy," *Energy Convers. Manag.*, vol. 44, no. 7, pp. 1111–1122, 2003.
- [77] W. Jittanit, G. Srzednicki, and R. Driscoll, "Seed drying in fluidized and spouted bed dryers," *Dry. Technol.*, vol. 28, no. 10, pp. 1213–1219, 2010.

- [78] J. Fernando and A. Amarasinghe, "Drying kinetics and mathematical modeling of hot air drying of coconut coir pith," *SpringerPlus*, vol. 5, no. 1, p. 807, 2016.
- [79] T. N. Tulasidas, T. Kudra, G. S. V. Raghavan, and A. S. Mujumdar, "Effect of bed height on simultaneous heat and mass transfer in a two-dimensional spouted bed dryer," *Int. Commun. Heat Mass Transf.*, vol. 20, no. 1, pp. 79– 88, 1993.
- [80] W. Jittanit, G. Srzednicki, and R. Driscoll, "Modelling of seed drying in fluidised and spouted bed dryers," *Julius-Kühn-Arch.*, no. 425, p. 275, 2010.
- [81] H. Kucuk, A. Kilic, and A. Midilli, "Common Applications of Thin Layer Drying Curve Equations and Their Evaluation Criteria," in *Progress in Exergy, Energy, and the Environment*, Springer, 2014, pp. 669–680.
- [82] A. Vega, P. Fito, A. Andrés, and R. Lemus, "Mathematical modeling of hotair drying kinetics of red bell pepper (var. Lamuyo)," *J. Food Eng.*, vol. 79, no. 4, pp. 1460–1466, 2007.
- [83] C. L. Hii, C. L. Law, and M. Cloke, "Modeling using a new thin layer drying model and product quality of cocoa," *J. Food Eng.*, vol. 90, no. 2, pp. 191– 198, 2009.
- [84] A. E. Ekow, M. A. Haile, O. John, and E. F. Narku, "Microwave-vacuum drying effect on drying kinetics, lycopene and ascorbic acid content of tomato slices," *J. Stored Prod. Postharvest Res. Vol.*, vol. 4, no. 1, pp. 11–22, 2013.
- [85] H. O. Menges and C. Ertekin, "Modelling of air drying of Hacıhaliloglu-type apricots," J. Sci. Food Agric., vol. 86, no. 2, pp. 279–291, 2006.
- [86] E. K. Akpinar, "Mathematical modelling of thin layer drying process under open sun of some aromatic plants," *J. Food Eng.*, vol. 77, no. 4, pp. 864–870, 2006.
- [87] E. Meisami-asl, S. Rafiee, A. Keyhani, and A. Tabatabaeefar, "Determination of suitable thin layer drying curve model for apple slices (variety-Golab)," *Plant Omics*, vol. 3, no. 3, p. 103, 2010.
- [88] W. P. da Silva, C. M. D. P. S. e Silva, F. J. A. Gama, and J. P. Gomes, "Mathematical models to describe thin-layer drying and to determine drying

rate of whole bananas," J. Saudi Soc. Agric. Sci., vol. 13, no. 1, pp. 67–74, Jan. 2014.

- [89] C. Ertekin and O. Yaldiz, "Drying of eggplant and selection of a suitable thin layer drying model," *J. Food Eng.*, vol. 63, no. 3, pp. 349–359, Aug. 2004.
- [90] İ. Doymaz, "Drying of Leek Slices Using Heated Air," J. Food Process Eng., vol. 31, no. 5, pp. 721–737, Oct. 2008.
- [91] M. m. i. Chowdhury, B. k. Bala, and M. a. Haque, "Mathematical Modeling of Thin-Layer Drying of Jackfruit Leather," J. Food Process. Preserv., vol. 35, no. 6, pp. 797–805, Dec. 2011.
- [92] T. Madhiyanon, A. Phila, and S. Soponronnarit, "Models of fluidized bed drying for thin-layer chopped coconut," *Appl. Therm. Eng.*, vol. 29, no. 14–15, pp. 2849–2854, 2009.
- [93] T. Arumuganathan, M. R. Manikantan, R. D. Rai, S. Anandakumar, V. Khare, and others, "Mathematical modeling of drying kinetics of milky mushroom in a fluidized bed dryer," *Int. Agrophysics*, vol. 23, no. 1, pp. 1–7, 2009.
- [94] M. L. Passos, A. S. Mujumdar, and G. Massaranl, "Scale-up of spouted bed dryers: Criteria and applications," *Dry. Technol.*, vol. 12, no. 1–2, pp. 351– 391, 1994.
- [95] A. N. Menon, K. P. Padmakumari, A. Jayalekshmy, M. Gopalakrishnan, and C. S. Narayanan, "Essential oil composition of four popular Indian cultivars of black pepper (Piper nigrum L.)," *J. Essent. Oil Res.*, vol. 12, no. 4, pp. 431– 434, 2000.
- [96] H. H. Jeleń and A. Gracka, "Analysis of black pepper volatiles by solid phase microextraction–gas chromatography: A comparison of terpenes profiles with hydrodistillation," *J. Chromatogr. A*, vol. 1418, pp. 200–209, 2015.
- [97] K. A. Buckle, M. Rathnawathie, and J. J. Brophy, "Compositional differences of black, green and white pepper (Piper nigrum L.) oil from three cultivars," *Int. J. Food Sci. Technol.*, vol. 20, no. 5, pp. 599–613, 1985.
- [98] Z. de L. Cardeal, M. D. R. Gomes da Silva, and P. J. Marriott, "Comprehensive two-dimensional gas chromatography/mass spectrometric analysis of pepper volatiles," *Rapid Commun. Mass Spectrom.*, vol. 20, no. 19, pp. 2823–2836, 2006.

- [99] "Publications." [Online]. Available: http://www.esa-spices.org/indexesa.html. [Accessed: 04-Oct-2016].
- [100] A. S. Mujumdar, Handbook of Industrial Drying, Third Edition. CRC Press, 2014.
- [101] G. K. Jayatunga, B. M. W. P. K. Amarasinghe, and S. M. N. D. Martino, "Thin Layer Drying Models for Drying of Black Pepper in Spouted Bed Dryer with Internal Devices," *IESL Annu. Sess.*, 2016.
- [102] A. Mendes, A. Dollet, C. Ablitzer, and G. Flamant, "Numerical simulation of spouted bed reactors using process engineering models: application to coal gasification," p. 9.
- [103] R. M. Felder and R. W. Rousseau, *Elementary principles of chemical processes*. John Wiley, 2000.
- [104] J. R. Welty, C. E. Wicks, G. Rorrer, and R. E. Wilson, Fundamentals of momentum, heat, and mass transfer. John Wiley & Sons, 2009.
- [105] G. K. Jayatunga and B. Amarasinghe, "Effect of configuration of non-porous draft tubes drying kinetics of black pepper in a spouted bed dryer," *IESL Annu. Sess.*, 2014.
- [106] M. Balakrishnan, G. S. V. Raghavan, V. V. Sreenarayanan, and R. Viswanathan, "Batch drying kinetics of cardamom in a two-dimensional spouted bed," *Dry. Technol.*, vol. 29, no. 11, pp. 1283–1290, 2011.
- [107] G. K. Jayatunga and B. M. W. P. K. Amarasinghe, "Drying Kinetics of Black Pepper Dried in A Spouted Bed Dryer with or without Draft Tubes," *Int. J. Manuf. Ind. Eng.*, vol. 1, no. 2, p. 5, Jun. 2014.
- [108] C. Lim Law and A. Mujumdar, "Fluidized Bed Dryers," in Handbook of Industrial Drying, Third Edition, A. Mujumdar, Ed. CRC Press, 2006.
- [109] A. Kaya, O. Aydın, and C. Demirtaş, "Drying kinetics of red delicious apple," *Biosyst. Eng.*, vol. 96, no. 4, pp. 517–524, 2007.
- [110] M. Markowski, W. Sobieski, I. Konopka, M. Tańska, and I. Bia\lobrzewski, "Drying characteristics of barley grain dried in a spouted-bed and combined IR-convection dryers," *Dry. Technol.*, vol. 25, no. 10, pp. 1621–1632, 2007.

- [111] J. Anuththara, E. Edirisinghe, B. Amarasinghe, and G. K. Jayatunga, "Kinetics and Mathematical Modeling of Drying of Parboiled Paddy in a Packed Bed Dryer," in *MERCON*, University of Moratuwa, Sri Lanka, 2019.
- [112] S. Sari, G. Kulah, and M. Koksal, "Characterization of gas-solid flow in conical spouted beds operating with heavy particles," *Exp. Therm. Fluid Sci.*, vol. 40, pp. 132–139, 2012.
- [113] A. S. Franca, M. L. Passos, A. L. T. Chattel, and G. Massarani, "Modeling and simulation of airflow in spouted bed dryers," *Dry. Technol.*, vol. 16, no. 9–10, pp. 1923–1938, 1998.
- [114] M. Serowik, A. Figiel, M. Nejman, A. Pudlo, D. Chorazyk, and W. Kopec, "Drying characteristics and some properties of spouted bed dried semi-refined carrageenan," *J. Food Eng.*, vol. 194, pp. 46–57, 2017.
- [115] G. K. Jayatunga and B. M. W. P. K. Amarasinghe, "Drying kinetics, quality and moisture diffusivity of spouted bed dried Sri Lankan black pepper," J. *Food Eng.*, vol. 263, pp. 38–45, Dec. 2019.
- [116] D. Evin, "Paraboloid-Based Spouted Bed Drying of Paddy: Aerodynamics, Temperature Distribution, and Moisture Degradation," *AgriEngineering*, vol. 1, no. 2, pp. 257–264, Jun. 2019.
- [117] N. M. Panagiotou, M. K. Krokida, Z. B. Maroulis, and G. D. Saravacos, "Moisture diffusivity: literature data compilation for foodstuffs," *Int. J. Food Prop.*, vol. 7, no. 2, pp. 273–299, 2004.
- [118] B. Amarasinghe, A. Aberathna, and K. Aberathna, "Kinetics and Mathematical Modeling of Microwave Drying of Sri Lankan Black Pepper (Piper nigrum)," *Int. J. Environ. Agric. Res.*, vol. 4, no. 2.
- [119] V. A. Parthasarathy, B. Chempakam, and T. J. Zachariah, *Chemistry of spices*. CABI, 2008.
- [120] "Chemical Formula Search." [Online]. Available: https://webbook.nist.gov/chemistry/form-ser/. [Accessed: 24-Sep-2019].
- [121] D. Stewart, *The Chemistry of Essential Oils Made Simple: God's Love Manifest in Molecules*. Care Publications, 2005.

- [122] H. T. Sabarez, "Chemical and physical changes during dehydration of prunes (Prunus domestica), Doctor of Philosophy thesis," Department of Chemistry, University of Wollongong, 1998.
- [123] J. A. Hartsel, J. Eades, B. Hickory, and A. Makriyannis, "Chapter 53 -Cannabis sativa and Hemp," in *Nutraceuticals*, R. C. Gupta, Ed. Boston: Academic Press, 2016, pp. 735–754.
- [124] A. Piga *et al.*, "Influence of different drying parameters on the composition of volatile compounds of thyme and rosemary cultivated in Sardinia," 2007.
- [125] C. Ratti and G. H. Crapiste, "Determination of heat transfer coefficients during drying of foodstuffs," J. Food Process Eng., vol. 18, no. 1, pp. 41–53, 1995.
- [126] A. Mujumdar and C. L. Law, "Fluidized bed drying," 2006.
- [127] M. J. San José, S. Alvarez, F. J. Peñas, and L. B. López, "Modelling of analysis of influence parameters on cycle time in a novel conical spouted bed dryer with a draft tube," 2012, vol. 3.
- [128] M. H. Morgan III and H. Littman, "Predicting the maximum spoutable height in spouted beds of irregularly shaped particles," *Ind. Eng. Chem. Fundam.*, vol. 21, no. 1, pp. 23–26, 1982.
- [129] X. Bi, "Initiation of spouting," in Spouted and Spout-Fluid Beds, Cambridge University Press, 2010.
- [130] M. Olazar, M. J. San José, A. T. Aguayo, J. M. Arandes, and J. Bilbao, "Pressure drop in conical spouted beds," *Chem. Eng. J.*, vol. 51, no. 1, pp. 53– 60, Feb. 1993.
- [131] Y.-L. He, C. J. Lim, and J. R. Grace, "Scale-up studies of spouted beds," *Chem. Eng. Sci.*, vol. 52, no. 2, pp. 329–339, 1997.
- [132] P. T. Tsilingiris, "Thermophysical and transport properties of humid air at temperature range between 0 and 100°C," *Energy Convers. Manag.*, vol. 49, no. 5, pp. 1098–1110, May 2008.

## APPENDIX A

Туре	Description
Tray dryer	Operated in batch mode.
(direct	• The material which may be lumpy solid or a pasty solid is
contact)	spread uniformly on a metal tray.
	• Hot air is re-circulated by a fan over and parallel to the
	surface of the trays.
	• After drying, the cabinet is opened and the trays are replaced
	with a new batch of trays.
Flash dyer	• Wet material is dispersed in to a stream of hot air which
(Convective/	conveys through a duct.
pneumatic/)	• Drying takes place during short time in a drying duct.
	• Therefore pneumatic driers are suitable for removal of
	surface moisture, only.
	• Ability to handle in continuous operation.
	• Flash dryers are suitable for agricultural products such as tea,
	coir, fish products, dairy products starch and etc.
Rotary dryer	• Consists of a hollow cylinder which is rotated and usually
(Direct/	slightly inclined toward the outlet.
indirect)	• The wet granular solids are fed at the high end and move
	through the shell as it rotates.
	• The heating is by direct contact with hot gases in counter
	current or co-current flow.
	• In some cases the heating is by indirect contact through the
	heated wall of the cylinder.
	• The lifting flights extending from the cylinder wall lift the
	solid and allow to fall and simultaneous by pushing to the
	outlet.

Table 1 : Brief description of widely used dryers

	• Rotary dryers are suitable for powders, granules, flakes,
	pastes, Slurries and etc.
Fluidized	• Usually Fluidized bed dryer consist of a shell of cylindrical
bed	or rectangular cross section.
	• A grid is provided in the column over which the wet material
	is rests. In this type of dryer, the drying gas is passed through
	the bed of solids at a velocity sufficient to keep the bed in a
	fluidized state.
	• Mixing and heat transfer are very rapid in this type of dryers.
	• The dryer can be operated in batch or continuous mode.
	• Ability to operate in batch mode and continuous mode.
	• Fluidized bed dryer are suitable for granular and crystalline
	materials, pharmaceutical product, sludge, tea and etc.
Spouted bed	• Hot gas is introduced vertically upward into the solid bed
dryer	through a small opening which is located at the center of the
	base of the vessel.
	• They are suitable to handle coarse particles such as grains,
	pastes, slurries and etc.
Spray dryer	• Liquid to be dried is atomized and introduced to a drying
	chamber.
	• Air fed through the filler and heater enters at the top of the
	drying chamber flowing down co-current with the droplets
	being dried.
	• The moisture evaporates and the dried product is collected to
	the bottom of the drier.
	• They are suitable for dairy product, dye stuff, coffee, tea,
	yeast and etc.
Drum dryer	• A drum dryer consists of a heated metal roll on the outside of
(indirect	which a thin layer of liquid or slurry is evaporated to dryness.
dryer)	• The final dry solid is scraped off the roll which is revolving
	slowly.

### Table 1 Continued

	•	Drum drier may be either a single drum, double drum, twin		
		drum or vacuum double drum		
	•	Drum driers are suitable for drying slurries, pastes and		
		sludge.		
Continuous	•	The solids are placed on trays or trucks which move		
tunnel dryer		continuously through a tunnel with hot gas passes over the		
		surface of each tray.		
	•	The air flow is either perpendicular to the material flow, co-		
		current or counter-current. (However in practice it will be a		
		combination).		



Figure 1: Powder classification diagram for fluidization by air at ambient conditions [12]

Table 2: Categorization of powders in	relation of fluidization characteristics
[12]	

Group	Typical	Fluidization/Powder Characteristics	Examples
	particle size		of materials
	(µm)		
А	30-100	Particulate expansion of bed will take place over	Cracker
		significant velocity range.	catalyst
		Small particle size and low density	
		$(\text{density} < 1400 \text{ kg/m}^3)$	
В	100-800	Bubbling occurs at velocity > minimum	sand
		fluidization velocity.	
		Most bubbles have velocities greater than	
		interstitial gas velocity. No evidence of	
		maximum bubble size.	
С	20	Fine cohesive powders, difficult to fluidize and	Flour
		readily form channels.	Fine silica
D	1000	All but largest bubble rise at velocities less than	Wheat
		interstitial gas velocity. Can be made to form	Metal shot
		spouted beds. Particle large and dense	

Table 3: Some of the key characteristics of gas spouted and gas fluidized bed dryers

Туре	Advantages	Limitations
Spouted	Reduced pressure drop	• Complex mechanism of particle movement and difficulty in
bed	• Relatively lower gas flow rate	control of cyclic pattern
	• Prevention of particle segregation	• Limitation of scale up considering the size and the bed
	• Possibility of handling coarse particles [56]	height
	Produce material circulation	• Lack of data available for the design and applications [9]
	• Producing excellent solid gas contact	• Low thermal efficiency [28]
	• Possibility of using high temperatures.	
	• Fast drying rates and uniform	
Fluidized	High rate of moisture removal	High pressure drop
bed	High thermal efficiency	High electrical power consumption
	• Easy material transport inside dryer	• Poor fluidization quality of some particulate products such
	• Ease of control	as bubbling or slugging.
	• Low maintenance cost [126]]	• Non uniform product quality of some materials
		• entrainment of fine particles [126]
		• Attrition or pulveriazation of particles
		• Agglomeration of fine particles may occur in high
		temperature operations or with sticky particles,
		• Erosion of vessel, internals and pipes
		• A very noticeable gas flow rate is required to reach
		fluidization of large particles
		• To confer the system adequate mixing and avoid dead
		zone, the gas rate should be doubled tripled with respect to
		the minimum fluidization velocity [56]

Type of bed	Advantages	Limitations	Applications
Conventional spouted bed	<ul> <li>Simple grid design</li> <li>Regular solid circulation</li> <li>Good solid mixing for coarse solids</li> <li>Low pressure drop</li> <li>Good fluidization for sticky or lumpy solid [45]</li> </ul>	<ul> <li>Limitation of geometry and operation</li> <li>Limitation on bed height</li> <li>Somewhat narrower range of particle size</li> <li>Poor spouting for finer particles [45]</li> </ul>	<ul> <li>Drying of grains and suspensions</li> <li>Coating of particles</li> <li>Granulation of particles</li> <li>Coal gasification</li> <li>Chemical product [45]</li> </ul>
Spouted bed equipped with non-porous draft tube (SBNPDT)	<ul> <li>Greater operation flexibility</li> <li>low gas flow and pressure drop required for operations</li> <li>Narrower spread of solid residence time distribution [127]</li> <li>Better control of solid circulation</li> <li>High bed stability under any operation condition</li> <li>Prevent grain from cracking during the drying [46]</li> </ul>	<ul> <li>Complex design</li> <li>Deuces solid circulation rate</li> <li>Plugging draft tube</li> <li>Reduces contact between gas and solids [46]</li> <li>Lower eat ans mass transfer</li> <li>Prevent gas percolation from spout to annulus</li> </ul>	<ul> <li>Drying of grains and suspensions</li> <li>Granulation of particles</li> <li>Coal gasification</li> <li>Combustion</li> <li>Pyrolysis of hydrocarbon</li> <li>Pneumatic conveying [45]</li> </ul>
Spouted bed equipped with porous draft tube (SBPDT)	<ul> <li>Advantages of SBNPDT are retained.</li> <li>More control of gas percolation through annulus</li> <li>Higher heat and mass transfe compare to SBNPDT [45]</li> </ul>	<ul> <li>More coplex design</li> <li>Plugging drat tube [45]</li> </ul>	• Drying of grain and chemical product [45]

Table 4: Characteristics of the conventional spouted bed versus spouted bed with draft tubes

# Table 5: Correlation for Minimum Spouting Velocity

Author	correlation	comments
Mathur and Gishler	$(d_{\lambda} (D_{\lambda})^{1/3} 2aH(a_{\mu} - a))$	Conventional spouted bed
1955	$U_{ms} = \left(\frac{u}{D}\right) \left(\frac{u}{D}\right) \qquad \sqrt{\frac{-gm(pp-p)}{\rho}}$	
[37]	N	
Fane and	$(d_{r}) (D_{i})^{1/3} \overline{2aH(a_{r}-a_{i})}$	Conventional spouted beds
Mitchell (1984)	$U_{ms} = 2.0 D^n \left(\frac{a_p}{D}\right) \left(\frac{a_1}{D}\right) - \sqrt{\frac{-gm(p_p - p)}{\rho}}$	For $D > 0.4m$
[48]	N	$n = 1 - exp\{-7D^2\}$
Wu et al. (1987)	$U_{ms} = 10.6  d_p ^{1.05}  D_o ^{0.266}  H_o ^{-0.095}  \rho_s - \rho ^{0.256}$	Conical spouted bed at elevated
[51]	$\frac{1}{\sqrt{2gH_o}} = 10.6 \left[ \frac{1}{D_c} \right] \qquad \left[ \frac{1}{D_c} \right] \qquad \left[ \frac{1}{D_c} \right] \qquad \left[ \frac{1}{D_c} \right]$	temperatures
Olazar et al. (1992)	$(P_a) = 0.126 4r^{0.5} \left(\frac{D_b}{D_b}\right)^{1.68} \left(\tan \frac{\gamma}{D_b}\right)^{-0.57}$	Conical spouted bed
[52]	$(Re_o)_{ms} = 0.120 AT \left(\frac{D_o}{D_o}\right) \left(\frac{\tan 2}{2}\right)$	
San José et al., (2007)	$(P_{a}) = 0.1264r^{0.5} (D_{b})^{1.68} [top \gamma]^{-0.57} (H_{o} - l_{d})^{0.45} (D_{i})^{0.17}$	Conical – cylindrical with draft tubes
[53]	$(He_o)_{ms} = 0.120H$ $(\overline{D_o})$ $ \tan \frac{1}{2} $ $(\overline{H_o})$ $(\overline{D_i - d_d})$	where $d_d \leq \overline{D_s}$ ,
		$d_d \leq \overline{D_s} < D_i$ , $l_d = H_o - h_d$
		and $h_d \ge 10d_p$
Altzibar et al. (2009)	$(R_{e}) = 0.2044r^{0.475} \left(\frac{H_{o}}{L_{H}}\right)^{1.240} \left(\frac{L_{H}}{L_{H}}\right)^{0.168} \left(\tan\frac{\gamma}{L_{H}}\right)^{-0.135}$	Conical spouted bed with draft tubes
[54]	$(D_o) = 0.204AT \qquad (D_o) \qquad (D_T) \qquad (\tan \frac{1}{2})$	

Table 6: Correlations available for prediction of maximum spoutable height

Author	Correlation	Geometry
McNab-Briggwater		Conventional spouted bed
[37]	$H_m = \left(\frac{D^2}{d_p}\right) \left(\frac{D}{D_i}\right)^{2/3} \left(\frac{568b^2}{Ar}\right) \left(\sqrt{(1+35.9Ar \times 10^{-6})}\right)$	
	$(-1)^{2}$	
Malek- Lu		Conventional spouted bed
[56]	$\frac{H_m}{D_c} = 418 \left(\frac{D_c}{d_p}\right)^{0.75} \left(\frac{D_c}{d_i}\right)^{0.40} \left(\frac{\lambda^2}{\rho_s^{1.2}}\right)$	
Passos et al., 1993	$H_m = 0.605 + \frac{6.21 \times 10^{-2}}{2.9 \times 10^{-3}}$	Two dimensional
[29]	$\frac{1}{L_1} = 0.003 + \frac{1}{A_{2D}} = \frac{1}{A_{2D^2}}$	spouted bed
		$A_{2D} = \frac{Re_{mf}Re_Td_p}{ArL_N}$
Morgan III & Littman, 1982	$\frac{H_m d_i}{H_m d_i} = 0.218 \pm \frac{0.005}{10000000000000000000000000000000000$	Conventional spouted bed
[128]	$D_c^2 = 0.210 + A$ , $H > 0.02$	For spherical particles
		$A = \frac{\rho_f}{\left(p_s - \rho_f\right)} \frac{U_{mF} U_T}{g d_i}$

 Table 7: Correlation available for prediction of maximum pressure drop

Author	correlation	geometry
Gelperin et al.(1961)		Conical spouted beds
[129]	$\frac{\Delta P_M}{H\rho_b g} = 1 + 0.062 \left(\frac{D_H}{D_i}\right)^{2.54} \times \left(\frac{D_H}{D_i} - 1\right) \left[\tan\frac{\theta}{2}\right]^{-0.18}$	
Gorshtein and Mukhlenov		Conical spouted beds
(1965)	$-\Delta P_{M} = 1 + 6.65 \left(\frac{H}{H}\right)^{1.2} (\tan^{\theta}\theta)^{0.5} 4\pi^{0.2}$	
[130]	$\overline{\Delta P_s} = 1 + 6.65 \left(\frac{D_i}{D_i}\right)  \left(\tan \frac{1}{2}\right)  Ar^{3/2}$	
Olazar et al. (1993)		Conical spouted beds
[130]	$\frac{-\Delta P_{M}}{\Delta P_{s}} = 1 + 0.116 \left(\frac{H}{D_{i}}\right)^{0.5} \left(\tan\frac{\theta}{2}\right)^{-0.8} Ar^{0.0125}$	
Olazar et al. (1994) [55]		Conical cylindrical
	$\frac{-\Delta P_{M}}{\Delta P_{s}} = 1 + 0.35 \left(\frac{H}{D}\right)^{0.1} \left(\frac{D_{i}}{D}\right)^{1.1} Ar^{0.1}$	
Manurung	$\frac{-\Delta P_M}{\Delta P_M} = \left[0.8 + \frac{6.8}{6.8} \left(\frac{D_i}{D_i}\right)\right] - 34.4 \frac{d_p}{d_p}$	Conical- cylindrical
[55]	$\Delta P_s \qquad \left\lfloor \cos^{-1} \tan \gamma \left\lfloor D \right\rfloor \right\rfloor \qquad H$	

Table 8: Options for scaling up of conventional spouted be	ed
--	----

Options available for scale up		Main characteristics
changing the geometry of the	increasing the size of a single unit [56]	Simple geometry or mechanical construction
conventional spouted bed	convert in to two dimensional spouted bed	Require low pressure drop
(CSB)	(2DSB) [33]	Require high fluid flow rates
	Convert into conical spouted bed (CCSB).	Require low pressure drop
	[131]	Require high fluid flow rates
		No limitation of the bed height
	Multiple unit (repeating the side by side	Complex design and construction
	several units) [56]	High investment and operating cost
		High heat loss
Modification of spouting	Insertion of draft tube or draft plates [43]	No limitation of the bed height
operation		Low pressure drop
		Low fluid flow rates
		Reduction of solid circulation rate
		Low heat transfer rate
	Insertion of any mechanical device to	
	improve or produce solid circulation such	
	as mechanical conveyor spouted bed or	
	vibrated spouted bed [94].	
	Introduction of addition fluid in the annular	Complex design and construction
	region	

Table 9: Specifications of black pepper defined by some institutes.

Institute	Ash	Acid	moisture	Volatile oil
	% W/W	insoluble ash	% W/W	% V/W
	Max	% W/W	Max	Min
		Max		
ESA	7	1.5	12	2
CSA	7	1	12	2
Sri Lanka spice			12-14	2-2.4
council				

## Table 10: Standard quality specifications of black pepper approved by Sri Lanka Standard Institute.

	Sp. Grade I	Grade I
Moldy berries %	1	1
Other extraneous matter %( insects	1	1
live or dead, stones, sand, plant		
parts, mammalian fecal matter etc.)		
Light berries %	Max. 4	Max. 4
Moisture %	12	14
Appearance	Dark black color	Dark black to
	with surface	brownish black
	grooves	color with
		surface grooves
#### Nomenclature for Appendix A

Ar- Archimedes number  $Ar = \frac{d^3 \rho (\rho_p - \rho)g}{\mu^2}$ b=Um/Umf D,  $D_c$ - diameter of the column (m) d, d<sub>p</sub> - particle diameter , horizontally projected diameter (m)  $D_b$ ,  $D_{H}$ - upper diameter of the stagnant bed in conical spouted bed (m) Di - diameter of the contactor base(m)  $D_0$  - diameter of the gas inlet (m)  $D_T$ , d<sub>d</sub>- diameter of the draft tube (m) g- Acceleration of gravity  $(m/s^2)$ H,  $H_0$  - height of the stagnant bed (m)  $H_m$  – maximum spoutable bed height (m)  $L_1$ - column width and column thickness in two dimensional spouted beds (m)  $L_{H}$ - height of entrainment zone (m)  $L_N$ - nozzle width in two dimensional spouted beds(m)  $l_d$ - length of the draft tube (m) Re<sub>mf</sub>- Reynolds number of minimum fluidization (Re<sub>o</sub>)<sub>ms</sub>- Reynolds number of minimum spouting, referred to Do Re<sub>T</sub>- Reynolds number of terminal velocity  $U_m$ -  $U_{ms}$  at maximum spoutable bed height (m/s) U<sub>mf</sub> -velocity at minimum fluidization (m/s)  $U_{ms}$ - minimum spouting velocity at the inlet orifice (m/s) U<sub>T</sub> - Terminal settling velocity of isoplated particle in spouting fluid (m/s)  $\Delta P_{M}$ - maximum pressure drop (Pa)  $\Delta P_s$  - Spouting pressure drop across bed (Pa)  $\lambda$  - Thermal conductivity of fluid (W/mK)  $\rho_{\rm f}$ ,  $\rho$ - Density of fluid or gas( kg/m<sup>3</sup>)

 $\rho_{\rm s}$ ,  $\rho_{\rm p}$  - density of solid or particle ( kg/m<sup>3</sup>)

 $\gamma, \theta$ - angle of the conical base of the contactor (degrees)

## **APPENDIX B**



Figure 1: Precision balance; VWRECN 611-2300



Figure 2: Protable digital balance; KERN PCB 350-3



Figure 3: Drying Oven; A lab Tech LDO-060E



Figure 4: EXTECH CFM Thermo anemometer model 407113; (b)- Thermo-Hygrometer Model GMK-920HT



Figure 5: GC-MS)-7890A gas chromatograph (Agilent, American) equipped with a 5975C plus mass spectrometer (Agilent, American)

# **APPENDIX C**



Figure 1: Spouted bed column

### **Dimensions of draft tubes**

- $D_T \quad \ \ \, \ \, diameter \ \, of \ \, tube$
- L<sub>H</sub> entrainment height
- $L_T$  whole length
- w Width of draft tube faces



Figure 2: non porous draft tube

### Dimensions of the cyclone separator

Appendix C Table 1: dimensions of cyclone separator

D	28	cm
a	14	cm
b	7	cm
De	14	cm
S	17.5	cm
h	56	cm
Н	112	cm
В	7	cm
•		



Figure 3: Reverse flow cyclone separator



Figure 4: Arrangement of temperature sensors

## **APPENDIX D**

Chromatograms and chromatagrapgy data sheets of essential oil derived from black pepper

Area Percent Report

Data Path : C:\msdchem\1\data\Commercial 2015.08.07\ Data File : 15.08.07.SAMPLE 1.D Acq On : 7 Aug 2015 12:17 'Acq On Operator Sample Misc ALS Vial : 2 Sample Multiplier: 1 Integration Parameters: autoint1.e Integrator: ChemStation Method : C:\MSDCHEM\1\METHODS\SMART.M Title : autointl.e Signal : TIC: 15.08.07.SAMPLE 1.D\data.ms peak R.T. first max last PK peak # min scan scan scan TY height corr. corr. area % max. % of total 259 BV 2 480898 21075700 1.10% 841 BV 2 7025794 555624092 28.89% -----1 1.531 211 238 1.10% 0.146% 2 5.013 788 828 3.850% 3 5.177 841 856 895 VV 3 14974489 1923420924 100.00% 13.328% 4 895 947 PB 5.545 918 2161139 111208802 895 918 947 PB 2161139 111208802 5.78% 0.771% 961 1014 1061 BV 2 10335150 1761225038 91.57% 12.204% 5 6.110 106110731081VV3090390931537454.84%108111161124PV1014920533204782.77%112411441160VV3621838049733331525.86% 6 6.458 0.645% 6.712 7 0.369% 8 6:879 3.446% 1160 1209 1235 VV 6 5792842 712248559 37.038 1235 1243 1251 VV 2 3411587 132897262 6.918 9 7.262 4.935% 7.463 10 0.921% 
 1269
 1299
 1306
 VV
 1179225
 39125979

 1306
 1328
 1334
 VV
 1967431
 65403496

 1334
 1391
 1403
 VV
 3
 2714041
 169277119

 1403
 1425
 1444
 VV
 5513407
 216786759

 1444
 1454
 1459
 VV
 1291134
 36696324
 11 7.794 2.03% 0.271% 12 7.962 3.40% 0.453% 13 8.335 8.80% 1.173% 14 8.536 11.27% 1.502% 15 8.704 1.91% 0.254% 
 1459
 1466
 1475
 VV
 4
 793572
 39575319

 1475
 1492
 1502
 VV
 6
 1545607
 89966015

 1502
 1507
 1515
 VV
 1462585
 4332876

 1515
 1534
 1538
 VV
 5
 1698074
 87917967

 1529
 1541
 1553
 W
 5
 050774
 1571767
 16 8.777 2.06% 0.274% 17 8.928 4.68% 0.623% 18 9.021 0.300% 19 9.177 4.57% 0.609% 20 9.281 1538 1551 1558 VV 4 2836488 108126035 5.62% 0.749% 21 9.360 1558 1565 1573 VV 2 2067253 67903550 3.53% 0.471% 
 1613
 1616
 1624
 VV
 2
 1420191
 42455595
 2.21%

 1648
 1665
 1675
 VV
 8356189
 335454274
 17.44%
 22 9.675 2.21% 0.294% 23 9.953 2.324% 10.089 1675 1688 1693 VV 2 2506144 114904121 10.152 1693 1699 1707 VV 2 3703306 97833340 24 5.97% 0.796% 5.09% 25 0.678% 26 1707 1716 1723 VV 2 3851557 109663590 1723 1730 1734 VV 2 2243811 64407955 1734 1738 1747 VV 3 2523389 71736369 10.254 5.70% 0.760% 27 10.337 3.35% 0.446% 28 10.383 3.73% 0 497% 29 10.469 1747 1752 1766 VV 5 1241874 69600824 3.62% 0.482% 30 1766 1772 1779 VV 2681187 69797202 10.587 3.63% 0.484% 31 10.689 1779 1790 1797 VV 5 1676378 72247997 3.76% 0.501% 32 10.772 1797 1804 1815 VV 1961371 69337366 3.60% 0.480% 33 
 1815
 1822
 1835
 VV
 4
 1326844
 66109011

 1835
 1841
 1849
 VV
 2245955
 64605387

 1849
 1868
 1886
 VV
 10
 1567030
 125528782

 1849
 1868
 1886
 VV
 10
 1567030
 125528782
 10.878 3.44% 0.458% 34 10.989 3.36% 0.448% 35 11.150 6.53% 0.870% 36 11.391 1902 1909 1932 VV 2 1505167 107575220 5.59% 0.745% 
 1932
 1964
 1976
 VV
 8
 1719906
 178355385

 1997
 2019
 2023
 VV
 2295448
 121910405
 37 11.718 9.27% 1.236% 12.042 38 6.34% 0.845% 39 12.109 2023 2030 2051 VV 3 3064321 163938893 8.52% 1.136% 40 12.320 2051 2066 2070 VV 1820955 83711244 4.35% 0.580%

*41 42 43 44 45	12.410 12.574 12.863 13.008 13.182	2070 2096 2139 2168 2200	2081 2109 2158 2183 2212	2096 2139 2168 2200 2248	VV VV VV VV VV	2 3 4	1203296 4332183 2382523 1337147 3796567	0 430153953 241046005 140557553 6 821537485 323619675	22.36% 12.53% 7.31% 42.71% 16.83%	2.981% 1.670% 0.974% 5.693% 2.242%
46 47 48 49 50	13.648 13.882 13.979 14.150 14.335	2248 2316 2341 2356 2391	2291 2331 2347 2376 2407	2316 2341 2356 2391 2417	VV VV VV VV VV	2 4 8	1081764 6963431 557333 5088834 795425	9 114761384 438870615 11535334 156145175 40538063	5 59.67% 22.82% 0.60% 8.12% 2.11%	7.952% 3.041% 0.080% 1.082% 0.281%
51 52 53 54 55	14.424 14.505 14.621 14.694 14.857	2417 2430 2442 2460 2476	2423 2436 2456 2468 2496	2430 2442 2460 2476 2500	VV VV VV VV VV	4	1026088 1951518 3265374 4836245 3188175	22647409 36848849 120556786 132594875 77070780	1.18% 1.92% 6.27% 6.89% 4.01%	0.157% 0.255% 0.835% 0.919% 0.534%
56 57 58 59 60	14.929 15.172 15.331 15.409 15.775	2500 2539 2566 2583 2615	2508 2549 2576 2589 2651	2520 2566 2583 2599 2665	VV VV VV BV	2 2	5077460 447543 4747920 869671 8689315	129696814 21674350 100403768 21704927 610847443	6.74% 1.13% 5.22% 1.13% 31.76%	0.899% 0.150% 0.696% 0.150% 4.233%
61 62 63 64 65	16.019 16.105 16.199 16.323 16.392	2684 2701 2712 2729 2750	2693 2707 2723 2744 2756	2701 2712 2729 2750 2768	VV VV VV VV VV	2 3 4	2052667 718660 3626466 3382375 4985748	38541663 13914848 90551046 104281522 105690582	2.00% 0.72% 4.71% 5.42% 5.49%	0.267% 0.096% 0.627% 0.723% 0.732%
66 67 68 69 70	16.529 16.673 16.806 17.376 17.897	2768 2787 2817 2911 2998	2779 2803 2826 2922 3011	2787 2817 2847 2933 3024	VV VV VV PV VV	2 2 2 4	1417881 1714492 626620 776492 595731	30747228 53052837 21545496 16229965 21086188	1.60% 2.76% 1.12% 0.84% 1.10%	0.213% 0.368% 0.149% 0.112% 0.146%
71 72 73 74	18.540 19.499 24.977 25.669	3099 3262 4166 4293	3120 3282 4210 4327	3133 3288 4219 4358	BV BV BV BB	3	424812 410582 526396 665768	7355786 11483280 16014516 20375334	0.38% 0.60% 0.83% 1.06%	0.051% 0.080% 0.111% 0.141%

Sum of corrected areas: 14431326310

SMART.M Tue Sep 08 13:51:16 2015





```
Area Percent Report
        Data Path : C:\msdchem\1\data\Commercial 2015.08.07\
        Data File : 15.08.07.SAMPLE 2.D
       Àcq On
                                  7 Aug 2015 15:20
       Operator
       Sample
      Misc
      ALS Vial : 2 Sample Multiplier: 1
       Integration Parameters: autointl.e
      Integrator: ChemStation
      Method
                          : C:\MSDCHEM\1\METHODS\SMART.M
      Title
                            : autointl.e
      Signal
                             : TIC: 15.08.07.SAMPLE 2.D\data.ms
   peak R.T. first max last PK peak
# min scan scan scan TY height
                                                                                                     corr.
                                                                                                                     corr.
                                                                                                                                            % of
                                                                                                   total
               ----- -----
    ___
                                           233 255 BV 2 795577 27095255
     1
               1.506
                                203
                                                                                                                                           0.224%

        205
        235
        255
        bV 2
        195517
        21095255
        1.278
        0.2248

        785
        826
        836
        BV 2
        6518875
        496333472
        23.298
        4.1078

        836
        853
        907
        VV 2
        14599325
        2131072142
        100.008
        17.6348

        907
        919
        965
        VB
        1686035
        99078228
        4.65%
        0.820%

        969
        1015
        1055
        BV 3
        9715195
        1634628394
        76.70%
        13.526%

      2
               5.001
      3
               5.164
      4
              5.551
      5
              6.117
     6
              6.376

        1055
        1058
        1065
        VV
        4410115
        113981538
        5.35%

        1065
        1077
        1093
        VB
        3148130
        108268156
        5.06%

        1102
        1147
        1177
        EV
        4
        5512570
        571899152
        26.84%

        1177
        1212
        1238
        VV
        4
        4951262
        596833752
        28.01%

        1238
        1248
        1259
        VV
        4
        3099677
        142690063
        6.70%

                              1055 1058 1065 VV
                                                                                                                                           0.943%
              6.484
                                                                                                                                           0.896%
     8
              6.901
                                                                                                                                           4.732%
              7.284
     9
                                                                                                                                           4.939%
  10
              7.493
                                                                                                                                         1.181%
  11
              7.603
                             1259 1266 1277 PV
                                                                             431872
                                                                                                 6036437
                                                                                                                        0.28%
                                                                                                                                          0.050%
  12
                            1277 1302 1318 VV 1692395 32678391
1318 1329 1359 PB 1816278 37710996
              7.815
                                                                                                                        1.53%
                                                                                                                                          0.270%
  13
             7.974
                                                                          1816278 37710996
                                                                                                                        1.77%
                                                                                                                                          0.312%
                             1362 1393 1405 BV 3 3885131 106776754
  14
             8.348
                                                                                                                        5.01%
                                                                                                                                          0.884%
  15
             8.541 1405 1425 1445 VV
                                                                      5769619 163349071
                                                                                                                        7.67%
                                                                                                                                          1.352%
  16
             8,708
                            1445 1453 1459 VV
                                                                            603436 10004000

        1445
        1453
        1453
        VV
        603436
        10004000

        1475
        1492
        1501
        VV
        4
        1174616
        37427396

        1501
        1507
        1516
        VV
        1560778
        29184916

        1516
        1532
        1538
        VV
        3
        891392
        31721739

        1538
        1550
        1557
        VV
        3
        1862381
        57488696

                                                                                                                        0.47%
                                                                                                                                          0.083%
  17
             8.938
                                                                                                                        1.76%
                                                                                                                                          0.310%
 18
             9.024
                                                                                                                       1.37%
                                                                                                                                          0.241%
  19
             9.174
                                                                                                                        1.49%
                                                                                                                                          0.262%
 20
             9.277
                                                                                                                        2.70%
                                                                                                                                         0.476%
 21
                            1557 1564 1583 VV 2 1301210 38971289
            9.359
                                                                                                                     1.83%
                                                                                                                                         0.322%

        1613
        1618
        1623
        VV
        764332
        15677249

        1648
        1664
        1673
        VV
        8757184
        280002089

 22
            9.677
                                                                                                                       0.74%
                                                                                                                                         0.130%
 23
            9.950
                                                                                                                   13.14%
                                                                                                                                         2.317%

        1643
        1664
        1673
        1680
        1691
        VV
        2
        1486192
        58070163

        1691
        1697
        1705
        VV
        3122577
        63305029

 24
          10.043
                                                                                                                       2.72%
                                                                                                                                         0.481%
 25
          10.146
                                                                                                                     2.97%
                                                                                                                                         0.524%
                        170517141722VV3238490857606715172217291745VV7172816765979183176517711778VV144351727736410177817891797VV287177726208849179718021814VV90163719400397
26
          10.248
                                                                                                                       2.70%
                                                                                                                                         0.477%
27
          10.333
                                                                                                                       3.10%
                                                                                                                                         0.546%
28
          10.581
                                                                                                                       1.30%
                                                                                                                                         0.230%
29
          10.686
                                                                                                                       1.23%
                                                                                                                                         0.217%
30
          10.768
                                                                                                                       0.91%
                                                                                                                                        0.161%
                          1814 1821 1834 VV 3 606828 17277429
1834 1839 1847 VV 1168535 23001692
1847 1867 1886 VV 7 743701 34311723
1886 1907 1930 VV 4 689985 39227821
1943 1962 1975 VV 10 638971 42727729
31
         10.876
                                                                                                                       0.81%
                                                                                                                                         0.143%
32
          10.987
                                                                                                                      1.08%
                                                                                                                                        0.190%
33
         11.150
                                                                                                                      1.61%
                                                                                                                                         0.284%
34
         11.386
                                                                                                                     1.84%
                                                                                                                                        0.325%
35
         11.712
                                                                                                                    2.00%
                                                                                                                                        0.354%
                          199520182022VV2114313440710097202220282048VV3192961973831448
36
        12.038
                                                                                                                      1.91%
                                                                                                                                       0.337%
37
         12.098
                          2048 2080 2094 VV 3 1929619 73831448 3.46% 0.611%
2048 2080 2094 VV 11676369 372225870 17.47% 3.080%
2094 2108 2135 VV 3730152 124882968 5.86% 1.033%
2150 2157 2167 VV 3 1652038 64345574 5.566%
38
        12.407
        12.571 2094 2108 2135 VV 3730152 124882968
12.861 2150 2157 2167 VV 3 1652038 64345574
39
40
                                                                                                                                     0.532%
                                                                                                                     3.02%
```

41 42 43 44 45	13.010 13.181 13.652 13.958 14.125	2167 2195 2241 2334 2351	2182 2211 2291 2343 2371	2195 2241 2334 2351 2386	VV VV VV PV VV	3 5 4	1431898 3884948 1265909 437642 5058779	2 732052256 242808158 3 158125471 7748540 136035062	5 34.35% 11.39% 17 74.20% 0.36% 6.38%	6.058% 2.009% 13.085% 0.064% 1.126%
46 47 48 49 50	14.256 14.323 14.413 14.493 14.610	2386 2397 2414 2426 2439	2393 2405 2420 2433 2453	2397 2414 2426 2439 2458	VV VV VV VV	4 7	436024 659106 1152669 1832190 3477558	7795951 25499888 22398732 35314606 126002170	0.37% 1.20% 1.05% 1.66% 5.91%	0.065% 0.211% 0.185% 0.292% 1.043%
51 52 53 54 55	14.678 14.842 14.913 15.310 15.383	2458 2473 2497 2562 2579	2465 2492 2505 2572 2584	2473 2497 2518 2579 2593	VV VV VV VV VV	3	5116575 3575259 6129872 3461727 513942	127905497 78616709 146422109 61827372 11639868	6.00% 3.69% 6.87% 2.90% 0.55%	1.058% 0.651% 1.212% 0.512% 0.096%
56 57 58 59 60	15.740 16.002 16.095 16.183 16.309	2593 2681 2699 2710 2726	2645 2689 2705 2720 2741	2661 2699 2710 2726 2748	VV VV VV VV VV	3 2 2 4	9693039 1419829 802083 3849246 1921722	476215158 30343394 14030817 82924781 73072297	22.35% 1.42% 0.66% 3.89% 3.43%	3.941% 0.251% 0.116% 0.686% 0.605%
61 62 63	16.380 16.517 16.662	2748 2765 2785	2753 2776 2801	2765 2785 2810	VV PV PV	5	5656998 679456 932616	110066545 14410659 20755199	5.16% 0.68% 0.97%	0.911% 0.119% 0.172%

Sum of corrected areas: 12084898760

SMART.M Fri Sep 11 10:17:31 2015





	1				Area Percent	Report					
	Data Pat Data Fil Acq On Operator Sample Misc ALS Vial	th : C:\msdc e : 15.08.2 : 20 Aug : : : : : : 1 Sam	hem\1\data 0.SAMPLE_3 2015 14:0 ple Multipj	\Commercial .D 6 Lier: 1	2015.08.07\						
	Integrat	ion Paramete	ers: autoir	ntl.e							
	Integrator: ChemStation										
	Title : autointl.e										
	Signal	: TIC: 15	.08.20.SAM	PLE_3.D\data	.ms						
р 	eak R.T. # min 	first max scan scan	last PK scan TY	peak height	corr. corr. area % max.	% of total					
	1 1.428	202 227	253 BV	12802398 48	1822728 35.09	\$ 2.774%					
	2 4.343	712 721	793 VV	18103311 51	9883445 37.87	\$ 2.994%					
	3 4.920	793 819	836 PV	5338789 299	917308 21.85%	1.727%					
	4 5.118	836 852	884 VV 3	17680147 12	53753977 91.32	2% 7.219%					
	5 5.428	884 905	934 VB	2704874 86	910375 6.33%	0.500%					
6	6.033	956 1007	1015 BV 4	14254918 12	41002000 90.39	9% 7.146%					
7	6.107	1015 1020	1037 VV 3	12468590 59	5523106 43.388	3.429%					
8	6.295	1037 1051	1074 PB	4978035 109	071645 7.94%	0.628%					
9	6.596	1079 1103	1114 BV 3	554155 27	535505 2.01%	0.159%					
10	6.757	1114 1130	1141 VV	12041216 49	7830095 36.26%	5 2.867%					
11	6.868	1141 1149	1158 VV	4514382 834	270956 6.08%	0.481%					
12	7.148	1158 1196	1224 VV 7	13239391 136	50045167 99.06	* 7.831%					
13	7.365	1224 1233	1238 PV	1586241 278	225778 2.03%	0.160%					
14	7.438	1238 1245	1254 VV	3651305 684	64535 4.99%	0.394%					
15	7.696	1254 1289	1306 VV 2	1560114 1224	56780 8.92%	0.705%					
16	7.867	1306 1318	1326 VV 2	4947316 1182	28581         8.61%           83688         2.26%           35605         2.81%           36493         13.97%           65906         15.24%	0.681%					
17	7.956	1326 1333	1339 VV 2	833217 309		0.178%					
18	8.055	1339 1350	1353 VV	1198491 385		0.222%					
19	8.122	1353 1361	1370 VV	9116371 1917		1.104%					
20	8.266	1370 1385	1410 VV 4	2855906 2092		1.205%					
21	8.478	1410 1421 1	1443 VV	L2391799 403	403707 29.38%	2.323%					
22	8.650	1443 1450 1	1457 VV 3	648634 197	71111 1.44%	0.114%					
23	8.898	1482 1492 1	1500 VV 4	1912409 1392	25751 10.14%	0.802%					
24	8.979	1500 1506 1	1515 VV 4 1	L228235 310	94988 2.26%	0.179%					
25	9.145	1515 1534 1	1541 VV 3 2	2886053 917	34417 6.68%	0.528%					
26	9.243	1541 1551 1	L560 VV 9 7	170060 22794	12925 16.60%	1.313%					
27	9.336	1560 1567 1	L575 VV 2 5	146296 10735	58643 7.82%	0.618%					
28	9.598	1591 1611 1	L626 BV 3 2	414509 9603	38019 7.00%	0.553%					
29	9.760	1626 1638 1	L652 VV 6	870257 5534	6191 4.03%	0.319%					
30	9.948	1652 1670 1	L681 VV 4 1	6607300 8880	003182 64.68%	5.113%					
31	10.051	1681 1688 1	698 VV 4	210823 15685	2303 11.42%	0.903%					
32	10.148	1698 1704 1	711 VV 2 8	286404 17025	7580 12.40%	0.980%					
33	10.246	1711 1721 1	737 VV 2 6	844295 21400	2213 15.59%	1.232%					
4	10.400	1737 1747 1	752 VV 2 3	050260 8785	5477 6.40%	0.506%					
5	10.454	1752 1756 1	771 VV 6 1	813101 7349	6356 5.35%	0.423%					
6	10.581	1771 1777 1	784 VV 51	522124 10529	6380 7.67%	0.606%					
7	10.682	1784 1795 1	803 VV 3 21	552249 7180	5420 5.23%	0.413%					
8	10.765	1803 1809 1	821 VV 30	086774 5424	1410 3.95%	0.312%					
9	10.981	1821 1845 1	853 FV 36	861028 8586	8148 6.25%	0.494%					
0	11.114	1853 1868 18	382 VV 7 16	523965 5743.	3590 4.18%	0.331%					

Sum of corrected areas: 17367048848

MART.M Fri Sep 11 10:24:42 2015







	•							Area P	ercent Re	port		
Da Da Ac Og Sa Mi	Data Path : C:\msdchem\l\data\Commercial 2015.08.07\ Data File : 15.09.03.SAMPLE_H.D Acq On : 3 Sep 2015 12:17 Operator : Sample : Misc : ALS Vial : 1 Sample Multiplier: 1											
Ir Ir	Integration Parameters: autoint1.e Integrator: ChemStation											
M∈ Ti	Method : C:\MSDCHEM\1\METHODS\SMART.M Title : autointl.e											
Signal : TIC: 15.09.03.SAMPLE_H.D\data.ms												
pea #	k R.T. min	first scan	max scan	last scan	PI T	K Y	peak height	corr. area	corr. % max.	% of total		
1 2 3 4 5	1.503 1.575 2.813 4.979 5,189	213 248 447 797 840	240 252 462 829 864	248 276 473 840 881	BV VB BB BV VB	3	6634414 9048946 1094950 1685321 7277523	138336107 220686582 25914118 88311134 606461253	22.81% 36.39% 4.27% 14.56% 100.00%	4.112% 6.560% 0.770% 2.625% 18.026%		
6 7 8 9 10	5.481 6.015 6.092 6.182 6.352	900 986 1009 1026 1042	914 1004 1017 1032 1061	941 1009 1026 1042 1071	BB VV VV VV PB	3	633523 5506356 4873573 4434092 1837987	22617632 211079539 281036047 179290653 43816848	3.73% 34.81% 46.34% 29.56% 7.23%	0.672% 6.274% 8.353% 5.329% 1.302%		
11 12 13 14 15	6.663 6.821 6.902 7.231 7.610	1098 1122 1148 1166 1263	1114 1141 1154 1210 1274	1122 1148 1166 1244 1287	BV VV PV VB VV	2 2 6	1585352 3761298 422878 4020995 1519915	52959817 197963666 9067153 376119691 37205632	8.73% 32.64% 1.50% 62.02% 6.13%	1.574% 5.884% 0.270% 11.180% 1.106%		
16 17 18 19 20	7.728 8.292 8.478 9.905 12.383	1287 1372 1406 1597 2067	1294 1390 1421 1663 2083	1311 1406 1431 1675 2089	VV BV PV BV BV		458686 644683 1007953 1423351 1819995	7756058 15921342 18932479 28222722 31638217	1.28% 2.63% 3.12% 4.65% 5.22%	0.231% 0.473% 0.563% 0.839% 0.940%		
21 22 23 24 25	12.976 13.152 13.656 14.057 14.583	2168 2204 2278 2348 2442	2183 2213 2298 2366 2455	2204 2242 2328 2375 2460	VV VB BB BV VV	4 9 5	4631681 794942 6341412 1579805 564346	117901120 26429930 468257675 28756334 18038170	19.44% 4.36% 77.21% 4.74% 2.97%	3.504% 0.786% 13.918% 0.855% 0.536%		
26 27 28 29 30	14.642 14.805 14.873 15.686 16.357	2460 2477 2498 2610 2737	2465 2493 2505 2642 2756	2477 2498 2519 2665 2770	VV VV VV BB BV	2	1236786 574939 1436171 1418399 678752	22771880 10455659 25052992 37373557 15977249	3.75% 1.72% 4.13% 6.16% 2.63%	0.677% 0.311% 0.745% 1.111% 0.475%		

Sum of corrected areas: 3364351256

MART.M Fri Sep 11 11:49:02 2015





Area Percent Report

41 42 43 44 45	16.380 16.448 16.584 16.731 17.939	2741 2761 2778 2797 3012	2756 2768 2791 2815 3020	2761 2778 2797 2827 3033	VV VV PV PV VV	4 6 3	1060173 1432674 715767 750949 452383	29692825 33450465 14386193 21251529 12794486	4.01% 4.52% 1.94% 2.87% 1.73%	0.565% 0.636% 0.274% 0.404% 0.243%
46	33.629	5619	5678	5704	BV	7	436330	56121092	7.58%	1.067%
47	44.419	7363	7506	7563	BV	8	665763	275794650	37.24%	

Sum of corrected areas: 5259580425

SMART.M Fri Sep 11 10:35:02 2015





Area Percent Report Data Path : C:\msdchem\1\data\Commercial 2015.08.07\ Data File : 15.09.03.SAMPLE\_17.D Acq On : 3 Sep 2015 10:32 Operator : Sample Misc : ALS Vial : 1 Sample Multiplier: 1 Integration Parameters: autoint1.e Integrator: ChemStation : C:\MSDCHEM\1\METHODS\SMART.M Method Title : autoint1.e Signal : TIC: 15.09.03.SAMPLE 17.D\data.ms peak R.T. first max last PK peak
 # min scan scan scan TY height corr. corr. area % max. % of total ------\_\_\_\_\_ 
 800
 824
 839
 BV 2
 1583932
 71108643
 15.45%

 839
 860
 886
 VB
 3
 7550933
 460249717
 100.00%

 886
 908
 933
 BB
 693608
 18027353
 3.92%

 970
 1007
 1013
 BV
 2
 5025703
 331003471
 71.92%
 4.951 1 2.723% 2 5.162 17.622% 3 5.449 0.690% 4 6.030 12.674% 5 9.632% 6.127 1013 1023 1043 VV 2 5609027 251562310 54.66% 6 6.301 1043 1053 1079 VB 8.18% 4.18% 1911508 37660231 1.442% 
 1092
 1105
 1116
 BV
 670882
 19248409
 4.18%

 1116
 1132
 1142
 VV
 4631057
 160022917
 34.77%

 1160
 1204
 1234
 BB
 5 5289960
 370115444
 80.42%
 7 6.610 0.737% 8 6.770 6.127% 7.196 9 80.42% 14.171% 10 8.268 1368 1386 1404 BV 472826 11869161 2.58% 0.454% 11 8.459 1404 1418 1442 VB 1200866 23324956 5.07% 0.893% 12 9.894 1599 1661 1674 BV 1341652 27498163 1769008 33416890 437616 8742089 5.978 7.268 1.053% 13 12.380 2062 2082 2099 BV 1.279% 14 12,551 2099 2111 2134 PB 437616 8742089 1.90% 0.335% 4761566 103109479 15 12.973 2172 2183 2192 BV 22.40% 3.948% 16 13.150 2192 2213 2239 PB 5 873317 27310408 5.93% 1.046% 2274 2297 2328 BB 7 6600321 448314029 17 13.649 97.41% 17.165% 18 2356 2366 2375 BV 1611253 27656632 2441 2455 2459 VV 3 564722 17379718 2459 2465 2476 VV 1434020 24691325 1.059% 14.052 6.01% 19 14.580 3.78% 0.665% 20 14.640 5.36% 0.945% 2476 2493 2498 VV 588439 2498 2504 2520 VV 2 1386647 21 2.25% 14.804 588439 10361346 0.397% 22 14.871 5.35% 2.53% 24618690 0.943% 
 2436
 2504
 2520
 VV
 2
 1386847

 2520
 2574
 2581
 PV
 8
 427389

 2600
 2644
 2665
 BV
 2620080
 23 15.285 11655781 0.446% 2.739% 24 15.695 15.54% 71534876 25 16.163 2703 2723 2729 BV 3 455806 9512234 2.07% 0.364% 26 16.357 2751 2756 2769 VV 667613 11742589 2.55% 0.450%

Sum of corrected areas: 2611736861

SMART.M Fri Sep 11 11:25:47 2015





• Area Percent Report ÷C Data Path : C:\msdchem\1\data\Commercial 2015.08.07\ \*Data File : 15.09.03.SAMPLE\_D.D Acq On : 3 Sep 2015 14:02 Operator : Sample Misc ALS Vial : 1 Sample Multiplier: 1 Integration Parameters: autoint1.e Integrator: ChemStation Method : C:\MSDCHEM\1\METHODS\SMART,M Title : autoint1.e Signal : TIC: 15.09.03.SAMPLE\_D.D\data.ms peak R.T. first max last PK peak
 # min scan scan scan TY height peak % of corr. corr. area % max. area % max. total - --------- ---- ---- -------------------\_\_\_\_ 9434131 394697267 100.00% 1683190 46933579 11.89% 1951406 45223835 11.46% 1 1.495 208 238 287 BB 33.879% 985 1010 1025 BE 1158 1188 1207 BB 1951406 1552 1559 PV 1088004 2 6.047 4.029% 3 7.103 3.882% 1540 1552 1559 PV 1088D04 24079577 1708 1718 1743 VB 2 1270965 26956239 9.248 4 6.10% 2.067% 10.227 5 6.83% 2.314% 1766 1776 1792 BV 2 1019359 20708112 1828 1845 1857 BV 1172376 22483727 2064 2090 2100 BV 2 1178449 34468998 6 10 574 5.25% 1.778% 7 10.980 5.70% 1.930% 8 12.423 8.73% 2.959% 
 2166
 2180
 2191
 PV
 1547748

 2440
 2448
 2458
 VV
 4
 462250
 9 12,956 28482261 7.22% 2.445% 10 14.540 12861758 3.26% 1.104% 1114.636 2458 2464 2486 VV 1576565 26518458 6.72% 2.276% 12 14.807 2486 2494 2499 PV 435759 6861575 1.74% 7.72% 0.589% 13 15.299 15.745 2567 2577 2584 BV 1771420 30486740 2.617% 14 2630 2652 2667 BV 5998024 254187394 64.40% 21.819% 15 16.001 2680 2696 2711 VV 834918 18273041 4.63% 1.568% 16 16.298 2731 2746 2752 VV 3 500394 13764811 3.49% 1.182% 16.365 2752 2757 2776 VV 2 631824 2776 2783 2799 VV 2 737793 17 15959541 4.04% 1.370% 18 16.517 16950660 4.29% 11.75% 1.455% 19 16.690 2799 2812 2822 VV 3 2013085 46382226 3.981% 20 16.843 2822 2838 2852 VV 3 691547 28602298 7.25% 2.455% 21 18.241 3061 3075 3082 VV 1865446 50122617 12.70% 4.302%

Sum of corrected areas: 1165004715

SMART.M Fri Sep 11 11:37:44 2015





Area Percent Report Data Path : C:\msdchem\1\data\Commercial 2015.08.07\ Data File : 15.09.03.SAMPLE G.D Acq On : 3 Sep 2015 11:09 Operator : Sample Misc : ALS Vial : 1 Sample Multiplier: 1 Integration Parameters: autointl.e Integrator: ChemStation Method : C:\MSDCHEM\1\METHODS\SMART,M Title : autointl.e Signal : TIC: 15.09.03.SAMPLE\_G.D\data.ms peak corr. peak R.T. first max last PK peak # min scan scan scan TY height corr. % of % max. total area \_\_\_\_\_ 1 1.510 230 241 264 BB 7397099 287066405 53.57% 10.758% 
 04/2/7
 21393900
 3.99%
 0.802%

 0.126
 984
 1023
 1032
 BV 3
 6664334
 233249213
 43.52%
 8.741%

 6.126
 984
 1023
 1032
 BV 3
 6654467
 535914218
 100.00%
 20.084%

 6.298
 1032
 1052
 1071
 PB
 1357798
 24754224
 4.62%
 0.928%

 7.191
 1160
 1203
 1202
 2 3 4 5 6 7.191 1160 1203 1229 BE 5 6375939 393376164 73.40% 14.742% 
 1503
 1203
 1225
 BE 3
 03/3535
 593370104

 1542
 1550
 1558
 BV
 423280
 8422658

 1649
 1661
 1674
 BV
 594138
 11204200
 1.57% 0.316% 2.09% 0.420% 9.239 8 9.890 10.219 1716 1716 1740 BB 3 470516 11067354 12.382 2065 2083 2097 BV 1738171 34930245 9 2.07% 0.415% 1.309% 0.415% 10 6.52% 11 12.557 2097 2112 2138 PB 3 510593 15060167 2.81% 0.564% 2171 2184 2204 VV 5346563 138003876 2204 2213 2239 VB 6 1272384 36554374 5.172% 12 12,981 25.75% 13 13,152 6.82% 14 13.652 2277 2298 2328 BB 4 6608584 443417114 82.74% 16.618% 15 14.055 2356 2366 2376 BV 1815854 31919400 5.96% 1.196% 2430 2436 2442 VV 720830 12865725 2442 2456 2461 VV 4 1062386 34769990 16 14.466 2.40% 0.482% 17 14.584 6.49% 1.303% 2442 2456 2401 VV 2461 2471 2479 VV 18 14.675 4399037 120057872 22.40% 4.499% 2479 2497 2501 VV 1200491 23854753 2501 2507 2520 VV 2 1752836 32696170 0.894% 1.225% 19 14.827 4.45% 20 14.887 6.10% 
 2565
 2577
 2583
 BV
 822467
 14063390

 2613
 2647
 2665
 BV
 3623725
 118194677

 2715
 2725
 2732
 VV
 2
 961181
 19667654

 2722
 2746
 2753
 VV
 5
 529385
 17747704

 265986
 30397559
 30397559
 30397559
 30397559
 21 15.297 0.527% 2.62% 15.713 22 22.05% 4.430% 23 16.173 3.67% 0.737% 3.31% 24 16.299 0.665% 25 16.371 2753 2758 2770 VV 1685986 30397559 5.67% 1.139% 16.514 2770 2783 2796 VV 446387 8506690 16.659 2796 2807 2820 VV 4 437083 9189728 16.514 2770 2783 2796 VV 26 1.59% 0.319% 27 1.71% 0.344%

7

Sum of corrected areas: 2668345430

MART.M Fri Sep 11 11:44:47 2015




#### Sample No: 9

Ŀ Area Percent Report Data Path : C:\msdchem\1\data\Commercial 2015.08.07\ Data File : 15.09.01.SAMPLE\_9.D Acq On : 1 Sep 2015 14:53 Operator : Sample : Misc Misc : ALS Vial : 1 Sample Multiplier: 1 Integration Parameters: autoint1.e Integrator: ChemStation Method : C:\MSDCHEM\1\METHODS\SMART.M Title : autointl.e Signal : TIC: 15.09.01.SAMPLE 9.D\data.ms peak R.T. first max last PK peak # min scan scan scan TY height 
 last
 PK
 peak
 corr.
 corr.

 scan
 TY
 height
 area
 % max.

 ---- ---- ---- ---- ---- 

 254
 BV 2
 12898895
 461289755
 25.71%
 % of corr. % of % max. total -----1.487 1 211 237 3 3308 2 1.649 
 287
 VV
 6353032
 191407677
 10.67%

 841
 BV
 2918640
 153061078
 8.53%
 254 265 1.382% 3 4.962 799 826 
 841
 BV
 2918640
 153061078
 8.53%
 1.105%

 871
 VV
 4
 15449994
 946084732
 52.74%
 6.830%

 894
 VV
 2108282
 67147612
 3.74%
 0.485%
 4 5.167 841 861 6.830% 5 5.259 871 876 
 894
 911
 949
 VB
 3
 1426814
 65239634
 3.64%
 0.471%

 964
 1016
 1045
 BV
 2
 13389017
 1793918449
 100.00%
 12.951%

 1045
 1059
 1080
 VB
 4774164
 106221163
 5.92%
 0.767%

 1085
 1106
 1117
 BV
 3
 573439
 22707884
 1.27%
 0.164%

 1117
 1129
 1159
 VV
 4937959
 164395716
 9.16%
 1.187%
 6 5.464 7 6.088 8 6.336 9 6.616 10 6.753 11 7.161 1159 1198 1232 VV 5 13964923 1648095362 91.87% 11.899% 12 7.887 1232 1321 1329 PV 1913224 74434985 4.15% 1329 1334 1348 VV 507571 13868777 0.77% 1369 1392 1405 BV 7 824272 39865868 2.22% 0.537% 13 7.961 0.100% 14 8,305 0.288% 1405 1422 1444 VV 4 6113902 177574876 9.90% 1.282% 15 8.480 

 1444
 1452
 1460
 PV
 3869746
 71440071
 3.98%

 1460
 1467
 1484
 VV
 3
 1175679
 35243363
 1.96%

 1484
 1493
 1503
 VV
 3
 6517779
 169576026
 9.45%

 1503
 1509
 1517
 VV
 2
 966291
 25098396
 1.40%

 1517
 1530
 1534
 VV
 2
 3744143
 87303408
 4.87%

 16 8.656 0.516% 17 8.749 0.254% 18 8.899 1.224% 19 8.995 0.181% 20 9.117 0.630% 

 9.184
 1534
 1541
 1548
 VV
 2
 5268919
 166481130
 9.28%

 9.284
 1548
 1558
 1564
 VV
 8972343
 293584962
 16.37%

 9.349
 1564
 1569
 1579
 VV
 2
 3511487
 74883558
 4.17%

 9.535
 1591
 1600
 1614
 VV
 3
 1037702
 35553068
 1.98%

 9.660
 1614
 1622
 1629
 VV
 1998504
 39723584
 2.21%

 21 1.202% 22 2.120% 23 0.541% 24 0.257% 25 0.287% 1629 1637 1653 VV 2 987524 27023454 1.51% 1653 1667 1678 PV 7969418 229439234 12.79% 26 9.754 0.195% 27 9.927 1.656% 
 1678
 1691
 1697
 VV
 4
 2707555
 94695814

 1697
 1704
 1713
 VV
 2
 4459399
 107651374
 28 10.068 5.28% 0.684% 29 10.149 6.00% 0.777% 30 1713 1723 1738 VV 2 8311442 244346429 10.259 13.62% 1.764% 31 10.459 1751 1757 1767 VV 2 808942 16419767 0.92% 0.119% 32 10.596 1767 1780 1790 VV 2 5093487 124518567 6.94% 0.899% 
 1790
 1796
 1805
 VV
 3
 569840
 22706419

 1805
 1811
 1823
 VV
 3396492
 68778544

 1823
 1848
 1856
 PV
 4927677
 103805875
 33 10.691 1.27% 0.164% 34 10.780 3.83% 0.497% 35 10.997 5.79% 0.749% 11.150 1856 1874 1887 VV 3 1182745 33441070 11.424 1908 1920 1940 VV 6 794132 56342689 11.681 1940 1964 1980 VV 8 1187409 74535383 36 1.86% 0.241% 37 3.14% 0.407% 38 4.15% 0.538% 39 11.819 1980 1987 2011 VV 6 1061759 55991907 3.12% 0.404% 40 12.037 2011 2024 2033 VV 6 712844 33527809 1.87% 0 242%

Sum of corrected areas: 13851071441

SMART.M Fri Sep 11 10:49:03 2015









								Ar	ea Pei	ccent Repo	ort				
Dat Dat Acq Ope: Samj Mis	a Path a File On rator ple c	: C:\r : 15.0 : 7 2 : :	nsdche 08.07 Aug 20	em\1\0 .SAMP1 015 :	data LE_1 L4:0	\( 1. 0	Commercia D	1 201	5.08.(	)7\					
ALS	Vial	: 2	Samp:	le Mu	tip.	11	er: 1								
Int. Int	Integration rataneters: autointile Integrator: ChemStation Method : C:\MSDCHEM\1\METHODS\SMART M														
Met) Tit	Method : C:\MSDCHEM\1\METHODS\SMART.M Title : autoint1.e														
Sig	Signal : TIC: 15.08.07.SAMPLE_11.D\data.ms														
peak #	R.T. min	first scan	max scan	last scan	PK TY		peak height	cc ar	ea	corr. % max,	% of total				
1	1.473	197 748	223	276	BB BV		14068658	6196 32823	08971	30.26%	4.251%				
2	5.154	831	846	885	VV :	2	15072032	1370	211275	5 66.92%	9.401%				
4	5.480	885	902	939	PB		1227859	7246	6229	3.54%	0.497%				
5	6.085	955	1004	1046	BV 3	2	12665159	2047	609708	3 100.00%	14.048%				
6	6.396	1046	1057	1062	PV		3127290	7378	5889	3.60%	0.506%				
~ 7	6.647	1062	1099	1110	PV		332895	1040	5096	0.51%	0.071%				
8	6.780	1110	1122	1142	VB		2152667	8205	1494	4.01%	0.563%				
9	7.205	1151	1194	1236	BV '	7	9812089	14290	12522	69.79%	9.804%				
10	7.914	1292	1314	1320	vv		940484	1798	0422	0.88%	0.123%				
11	7.981	1320	1325	1335	VB		196751	490	0187	0.24%	0.034%				
12	8.301	1354	1380	1394	BV	6	806039	3531	6797	1.72%	0.242%				
13	8.494	1394	1412	1434	VV :	2	4198271	12215	0350	5.97%	0.838%				
14 15	8.670	1434	1442	1450	VV VV	2	806523	3182	5833	∠.0⊥≋ 1.55%	0.2838				
10															
16	8.911	1473	1483	1492	VV ·	4	3716558	10644	7088	5.20%	0.730%				
10	9.002	1492	1498	1506	VV .	3	8/8831 2199885	2466	8925	2 70%	0.1095				
19	9,120	1506	1520	1525	vv .	2	3120073	8857	6293	4.33%	0.608%				
20	9.288	1536	1547	1553	vv :	2	5406382	17458	9745	8.53%	1.198%				
0.1	0 252	1653	1	1567	\$ 75 7	2	1006176	1762	0452	0 338	0 3278				
21	9.303	1603	1611	1617	VV /	4	1497012	3580	1930	1.75%	0.246%				
23	9.763	1617	1627	1641	vv	5	877995	4168	1910	2.04%	0.286%				
24	9,946	1641	1658	1667	vv :	3	9130199	33246	2576	16.24%	2.281%				
25	10.079	1667	1681	1687	VV :	3	2900140	10951	7334	5.35%	0.751%				
26	10.161	1687	1695	1702	vv	2	5811500	15021	1509	7.34%	1.031%				
27 3	10.266	1702	1712	1728	vv :	2	7140490	22045	9445	10.77%	1.513%				
28 3	10.465	1741	1746	1755	vv		746095	2228	5725	1.09%	0.153%				
29 :	10.596	1755	1768	1778	VV	_	4856398	12468	2302	6.09%	0.855%				
30 :	10.697	1778	1785	1792	VV :	3	1189206	4268	7575	2.08%	0.293%				
31 3	10.782	1792	1800	1812	vv		3670113	8863	7555	4.33%	0.608%				
32	10.996	1812	1836	1846	vv		4176015	11381	1545	5.56%	0.781%				
33 3	11.153	1846	1863	1876	VV 2	2	1413461	5792	4735	2.83%	0.397%				
34 3	11.392	1897	1903	1927	VV !	5	1400650	8205	5857	4.01% 5 0.20	0.203% 0 2330				
35 .	11.081	т977	1925	тара	VV i	Ø	1400000	12143	1208	0.938	0.0008				
36 3	11.818	1969	1975	2001	vv ·	7	1251419	8895	6592	4.34%	0.610%				
37 3	12.413	2055	2076	2090	VV		13128263	5175	33601	25.28%	3.551%				
38 3	12.573	2090	2103	2144	VV .	2	4399303	27948	2658	13.65%	1,917%				
39 3	12.861	2144	2152	2163	VV :	2	2706002	12037	4589	5.88%	U.826%				
40 .	17.011	∠163	21//	2190	VV.	3	TSA05201	0021	12010	42.108	リ・ツエンタ				

41 42 43 44 45	13.189 13.658 13.856 13.976 14.141	2196 2242 2312 2331 2348	2208 2287 2320 2341 2369	2242 2312 2331 2331 2348 2376	VV VV VV VV VV	7 5 4	3793224 1021304 6568588 1019671 4864596	257136991 1 110029614 284400634 18166081 148043850	12.56% 47 53.74% 13.89% 0.89% 7.23%	1.764% 7.549% 1.951% 0.125% 1.016%
46 47 48 49 50	14.345 14.518 14.626 14.747 14.973	2384 2425 2438 2457 2483	2403 2433 2451 2471 2510	2425 2438 2457 2483 2520	VV VV VV VV VV	7 6 4 8	1382092 3554178 4528591 5513636 3310657	84300516 78607942 173905036 360530715 154367164	4.12% 3.84% 8.49% 17.61% 7.54%	0.578% 0.539% 1.193% 2.473% 1.059%
51 52 53 54 55	15.202 15.366 15.476 15.789 16.065	2537 2565 2590 2605 2670	2548 2576 2595 2648 2695	2565 2590 2605 2670 2701	VV VV VV VV VV	2 2 2 7 2	559189 3761196 598054 6872527 1264856	26833730 115396551 20445548 629451668 40998465	1.31% 5.64% 1.00% 30.74% 2.00%	0.184% 0.792% 0.140% 4.318% 0.281%
56 57 58 59 60	16.227 16.456 16.573 16.709 16.838	2711 2731 2769 2788 2815	2722 2761 2781 2804 2826	2731 2769 2788 2815 2844	VV VV VV PV VV	6 2 2 2	1658450 3451534 900311 700984 655287	58582054 173706116 17636814 29235502 18918833	2.86% 8.48% 0.86% 1.43% 0.92%	0.402% 1.192% 0.121% 0.201% 0.130%
61 62 63 64 65	17.074 17.284 17.409 17.814 17.925	2844 2878 2914 2972 3002	2866 2901 2922 2991 3010	2878 2914 2938 3002 3023	PV VV VV VV VV	7 9 3 6 3	538077 529505 1290857 692402 1445220	21304319 33705520 39220985 46479428 47420483	1.04% 1.65% 1.92% 2.27% 2.32%	0.146% 0.231% 0.269% 0.319% 0.325%
66 67 68 69 70	18.099 18.222 18.466 18.560 18.673	3023 3050 3077 3109 3131	3039 3060 3101 3117 3137	3050 3077 3109 3131 3143	VV VV VV VV	3 6 6 6	718906 693271 552120 1676060 654127	45079555 38226643 43985993 53781292 22426855	2.20% 1.87% 2.15% 2.63% 1.10%	0.309% 0.262% 0.302% 0.369% 0.154%
71 72 73 74 75	19.157 19.232 19.314 19.839 20.671	3207 3224 3239 3322 3447	3219 3231 3245 3334 3475	3224 3239 3263 3359 3491	VV VV VV VV VV	4 5 4 2	778972 672593 775528 525123 458175	25460881 23539244 27575992 24302244 17934736	1.24% 1.15% 1.35% 1.19% 0.88%	0.175% 0.161% 0.189% 0.167% 0.123%
76 77 78 79 80	20.965 22.245 22.463 22.785 24.688	3491 3736 3770 3816 4134	3525 3742 3778 3833 4155	3540 3754 3796 3842 4161	VV VV PV BV VV	3	657004 614894 876148 521808 475073	17203704 12945953 21589538 16822281 9679764	0.84% 0.63% 1.05% 0.82% 0.47%	0.118% 0.089% 0.148% 0.115% 0.066%

Sum of corrected areas: 14575738308

SMART.M Fri Sep 11 11:02:02 2015



File Operat	or	:C:\ma :	sdchem	1∖da	ta\Co	mmercia	al 201	5.08.07	\15.I	08.07.	SAMPLE_1	1.E	)			
Instru Sample	iment Name	່ ບໍ	SJP GC	MSD	4.00	us.	LIIY AL	qme troc	Fiai	IL EXI	IACLS.M					
Vial N	.nio Jumber	: 2								E	RR					
Abundan		2 2 3	<b>4</b> Ω	ŋ	66 6	<u>დ</u> თ	TIC: 15.0	18.07.SAMP	LE_11.	D\date.m	°040		20	812	N 60	ເພ
1.5e+C	97 - 17 aus)- 17	S*)]- 5	aene: «	llene: 5	(E)-: 9 (E)-: 8 llene: 9	ha.)-: 5 ta.)]-: 9 ha (-: 9	(S)-: 9	rene: 9 liene: 8 er 2): 4	xide: 9	xide: 6 -ene: 8	aene: 9 (thyl-: 4 ans-: 6 exo)-: 5	cis)-: 4	xide: 5 10ne: 3	ester: 3 rmin: 3	thyl-: 2 thyl-: 3	nyl)- 4 -one: 3 ane: 3
1.5e+0	thyl)-, (3R-tra alobe -Cubel	a.7a.beta.8	.7.aipha.,7a	Caryophy	R-(1R*,4Z,9 -methylene-, aCarvophv	lipha, 8a.alp alpha, 8a.be alpha, 8a.be	4-hexenyt)-, lethyl)-, (1S-	phaCalaco e-1,6-heptad ethyl- (isom	/ophyllene o	/ophyllene o 2.2.1]hept-2	Copa re, 2,7-dime a-methyl-, tr (2-endo,5-∈	lethyl)-, (1S-	abolene epo -4a-yl}-ethar	iyl-, methyl ∈ yisotrichode	dro-5,5-dime amino-4-me	methylethe anyi)butan-2 cyclohexylid
1.4e+0	17 12 13 13	ha.,5.alpha	sta. 4. beta		thylene[1 dirrrethyl-3 alpha	alpha.,4a.a alpha.,4a.a alpha.,4a.a	nethylene- -(1-methy	.al Methylene ,10-tetram	Can	Caŋ Tylbicyclo[	Quinoli Xahydro-8 -trimethyl-,	(1-methy	alphaBìs ta[b]pyran	-tetrameth 8-Hydrox	1-2.5-dihyo 1-2-one, 7-	nethyl-2-(1 clohex-3-6 anone, 2-0
1.3e+0	7 ( <del> </del>	beta.,4.alp	oha., 3b.be		thyŀ-8-mei ne, 7,11-<	thyl)-, (1.a )-, [2R-(2. thyb-, (1.a	nethyl-1-r dimethyl-1	3. -ol, 2,6,10		iethyl-2-vii	,5,8,8a-he dìol, 1,7,7	dimethyl-4	trans-Z-, -cyclopen	id, 3,3,8,8	3,4-triethy enzopyrar	e, 2,5-din limethylcy Cyclohex
1.2e+0	t-methyle	lipha.,3a.t	3aS-(3a.al		1,11-trime Dodecatrie	(1-methyle hylathenyl (1-methyle	sthyl-4-(5- hydro-4,7-	Indecan-5		1,7,7-Trin	ne, 3,4,4a ptane-2,5-	hydro-1,6-	ahydro-2H	boxylic ac	aborole, 2, 2H-1-B	dohexanor xy-4-(4,6-c
1.1e+0	nethyl-3-(	alpha.,2.4	ylethyl)-, [	13.658	4-ene, 4,1 1,6,10-	thylene-1- 1-2-(1-met imethv!-1-	xene, 1-m 6,8a-hexa	2.0.0(2,6)]			phthaleno o[2.2.1]he	2,3,4-tetra	3,7,7a-tetr	tane-5-ca	1,2-Oxa	Cyc 4-Hydro
1e+0	ethenyl-4-	yl)-, [1S-(1	4-(1-meth		2.0]undec-	sthyl-4-me .8-dimethy vdro-4.7-d	Cyclohe: e, 1,2,3,5,	ricyclo[7.2			1(2H)-Na Bicycl	nalene, 1,	thylene-5,	).0(2,4)]oc		
900000	exeñe, 4a	methyleth	thylene-		Bicyclo[7.2	ydro-7-me ahydro-4a, 8a-hexah	aphthalen	4				Napht	thyl-2-met	oycio[5.1.0		
800000	Cycloh	ethyl-5-(1-	nethyl-3-m			6,8a-octah 6,6,8a-octa 1,2,4a,5,6,	Ż						(5,5-Dime	T U		1994, super- 1 - 4 Gan, supering Maring - 2 - 2
700000		-1,7a-dime	ahydro-7-n	13	.856 /	,3,4,4a,5,6 ,2,3,4,4a,5 ithalene, 1			15.789				÷		an dalar tak	r 1. La Astron
600000		octahydro	Izene, oct	V		alene, 1,2 tinalene, 1 7, Napi	1.747 K		1						a series and a series of the s	
5000000	12.5	H-indene,	ipa[1,2]bei		14.14	tan Napt Napt Napt	6									
4000000		Metheno-1	3jcyclosic 6890			14.518	14.973	15.366		18	5.456					
3000000		12361	<u>अठ्ठ</u> र्सावभि											N.		
2000000		$\mathbb{N}$	1H-Cyc			4.345			18	16.227 6.065			17.409	17.925	i	18.560
1000000		$\mathbb{A}$	VV		19.976	$\mathbb{N}$		5.20215.476	W	$\mathbb{W}$	16.573 16.783387	.074 N	.284 M	17.81418	. <b>08.922</b> A A . A	18.67 8.466 //
0 Time>	12.50	13.0	0 13.5	0	14.00	14.50	15.00	15.50	16.	.00	6.50 17.	00	17.50	18.0	0	18.50

• File



						Area P	ercent Re	eport						
i i i i i i i i i i i i i i i i i i i	Data Pat Data Fil Acq On Dperator Sample Misc ALS Vial	h : C:\msd e : 15.09. : 1 Sep : : : : : 1 Sa	Chem\1\c 01.SAMPI 2015 1 mple Mul	lata\( E_12 2:25 tipli	Commercia .D ier: 1	al 2015.08	.07\							
1	Integrati Integrato	ion Parame or: ChemSt	ters: au ation	toint	:1.e									
M	Method : C:\MSDCHEM\1\METHODS\SMART.M Title : autointl.e													
S	ignal	: TIC: I	15.09.01	. SAMP	LE_12.D\	data.ms								
ре 	ak R.T. # min 	first ma scan sca	ax last an scan	PK TY	peak height	corr. area	corr. % max.	% of total						
1 2 3 4 5	4.969 5.159 5.467 6.106 6.304	804 82 842 85 898 91 967 101 1043 105	27 842 E 59 898 V 1 937 V 9 1043 E 3 1080 V	3V /V /B 2 3V 3 /B	3189108 : 17719307 1149635 15694184 7471052 :	110520493 770527340 36861518 160898498 161374793	6.87% 47.89% 2.29% 6 100.00% 10.03%	1.297% 9.044% 0.433% 18.886% 1.894%						
6 7 8 9 10	6.607 6.738 7.163 7.702 7.872	1080 110 1116 112 1160 119 1269 129 1299 131	4 1116 E 7 1142 F 8 1245 E 0 1299 B 9 1329 V	V B 5 1 V V 2	440619 2733883 16135362 445754 762685	11462684 68430375 133851888 8379934 19215715	0.71% 4.25% 3 83.19% 0.52% 1.19%	0.135% 0.803% 15.712% 0.098% 0.226%						
11 12 13 14 15	8.275 8.458 8.644 8.876 9.132	1367 138 1410 141 1443 144 1482 148 1516 153	7 1410 B 8 1443 V 9 1456 P 9 1502 V 2 1538 V	V 2 V 2 2 V V 4 1 V 9	699330 2120904 429479 102175 880037	19492428 49850064 7493200 29310940 26937410	1.21% 3.10% 0.47% 1.82% 1.67%	0.229% 0.585% 0.088% 0.344% 0.316%						
16 17 18 19 20	9.238 9.322 9.899 10.051 10.111	1538 1559 1558 1564 1650 1662 1675 1688 1693 1698	0 1558 V 4 1588 V 2 1675 P 3 1693 V 3 1707 V	V 3 1 B 3 V 7 V 5 V 3	.904315 540495 467529 1 723107 503087	47834124 14141478 52568415 26333254 65431078	2.97% 0.88% 9.48% 1.64% 4.07%	0.561% 0.166% 1.791% 0.309% 0.768%						
21 22 23 24 25	10.218 10.560 10.748 10.969 12.383	1707 1716 1764 1774 1801 1806 1832 1843 2054 2083	5 1736 V 1783 B 5 1817 V 8 1857 V 8 2097 B	732 71 7 7 8	255479 440180 853008 853423 061151 1	53733740 25703433 15442464 15983001 53767760	3.34% 1.60% 0.96% 0.99% 9.56%	0.631% 0.302% 0.181% 0.188% 1.805%						
26 27 28 29 30	12.555 12.842 12.981 13.154 13.443	2097 2112 2145 2161 2170 2184 2204 2213 2246 2262	2123 PV 2170 BV 2204 VV 2246 VE 2278 BV	2 2 3 1 1 2 4 7	079865 218901 4856606 328059 13 436529 2	50437206 26463381 438223618 33680097 23363547	3.13% 1.64% 27.24% 8.31% 1.45%	0.592% 0.311% 5.144% 1.569% 0.274%						
31 32 33 34 35	13.638 13.919 14.057 14.306 14.384	2278 2295 2337 2343 2349 2366 2396 2409 2418 2422	2337 VV 2349 PV 2375 VV 2418 VV 2430 VV	4 15 61 5 12 4 6	5709173 1 948513 1 142399 12 275129 5 517340 1	119595931 4091452 0727292 3556081 7010685	69.58% 0.88% 7.50% 3.33% 1.06%	13.142% 0.165% 1.417% 0.629% 0.200%						
36 37 38 39 40	14.466 14.584 14.674 14.823 14.891	2430 2436 2442 2456 2461 2471 2478 2496 2501 2508	2442 VV 2461 VV 2478 VV 2501 VV 2522 VV	29 5 40 11 34 2 73	962592 5 000841 14 049173 3 69327 7 69950 15	4262504 8733313 41783095 0135988 3877080	3.37% 9.24% 21.24% 4.36% 9.56%	0.637% 1.746% 4.012% 0.823% 1.806%						

41 42 43 44 45	15.042 15.151 15.296 15.389 15.709	2522 2541 2566 2583 2602	2533 2552 2576 2592 2646	2541 2566 2583 2602 2665	PV VV VV VV	3	555119 809249 2583780 571745 9667535	16015572 22044661 49308626 19197274 354747191	1.00% 1.37% 3.06% 1.19% 22.05%	0.188% 0.259% 0.579% 0.225% 4.164%
46 47 48 49 50	15.914 15.988 16.083 16.171 16.377	2665 2686 2704 2714 2732	2681 2693 2710 2725 2759	2686 2704 2714 2732 2770	PV VV VV VV VV	6 3 2 2	418606 1185963 619159 4037820 7898649	16269289 28186230 10822109 89294408 251203800	1.01% 1.75% 0.67% 5.55% 15.61%	0.191% 0.331% 0.127% 1.048% 2.949%
51 52	16.514 16.658	2770 2791	2783 2807	2791 2820	PV VV	2 5	1403889 1330636	27341040 30670532	1.70% 1.91%	0.321% 0.360%

Sum of corrected areas: 8519341511

SMART.M Fri Sep 11 11:05:45 2015





### Sample No :12

Area Percent Report Data Path : C:\msdchem\1\data\Commercial 2015.08.07\ Data File : 15.08.25.SAMPLE\_13.D Acq On : 25 Aug 2015 12:19 Operator : Sample Misc ALS Vial : 1 Sample Multiplier: 1 Integration Parameters: autoint1.e Integrator: ChemStation Method : C:\MSDCHEM\1\METHODS\SMART.M Title : autointl.e : TIC: 15.08.25.SAMPLE 13.D\data.ms Signal corr. corr. % of area % max. total peak R.T. first max last PK peak # min scan scan scan TY height peak \_\_\_\_\_ ------ 
 205
 240
 282
 BB
 15175034
 410899076
 22.08%
 3.494%

 806
 828
 844
 BV
 3517200
 176342533
 9.30%
 1.471%
 1,502 1 2 4.973 
 806
 628
 644
 BV
 351/200
 1/0342333
 9.30%
 1.4/1%

 844
 862
 896
 VB
 2
 17362090
 1061464272
 55.96%
 8.855%

 898
 913
 940
 BB
 1324678
 47775226
 2.52%
 0.399%

 970
 1017
 1047
 BV
 3
 14659560
 1896871904
 100.00%
 15.823%
 3 5,174 4 5.475 5 6.092 
 1047
 1060
 1083
 VB
 3739011
 73546037
 3.88%
 0.614%

 1120
 1130
 1146
 VB
 2907359
 80236750
 4.23%
 0.669%

 1156
 1203
 1232
 BV
 5
 12359090
 1427062314
 75.23%
 11.904%

 1304
 1322
 1329
 BV
 825632
 14173519
 0.75%
 0.118%

 1356
 1390
 1407
 BV
 7
 496566
 25742301
 1.36%
 0.215%
 6 6.345 6.759 8 7.188 9 7.893 10 8.292 11 8.479 1407 1422 1445 VV 5 3254249 91078848 4.80% 0.760% 
 1407
 1422
 1445
 vy
 5
 22.92.93
 91.07.001

 1445
 1452
 1460
 VV
 1455591
 26009330

 1484
 1492
 1502
 PV
 4
 3065444
 70726541

 1517
 1538
 1544
 VV
 4
 2391031
 84952209

 1517
 1556
 1562
 VV
 5065634
 133528098
 12 13 8.657 1.37% 3.73% 0.217% 8.897 0.590% 14 9.167 0.709% 4.48% 1544 1556 1562 VV 5065934 133528928 15 9.271 7.04% 1.114% 16 1473259 29835816 9.342 1562 1568 1577 VV 1.57% 0.249% 
 9.641
 1615
 1622
 1628
 1VV
 14/32/59
 29835816
 1.5/8

 9.661
 1615
 1622
 1628
 VV
 868031
 16481233
 0.878

 9.926
 1652
 1667
 1678
 VV
 28329097
 223581313
 11.798

 10.069
 1678
 1691
 1696
 VV
 3
 2021176
 71000751
 3.74%

 10.141
 1696
 1703
 1712
 VV
 4481412
 96705266
 5.10%
 17 0.137% 18 1.865% 19 0.592% 20 0.807% 21 10.249 1712 1721 1738 VV 2 6711802 163551310 8.62% 1.364% 
 10.249
 17/2
 17/2
 17/3
 VV
 2
 0/1802
 103331310
 6.028

 10.584
 1765
 1778
 1786
 VV
 4051647
 83423250
 4.408

 10.689
 1786
 1796
 1803
 VV
 744063
 23974600
 1.268

 10.771
 1803
 1810
 1821
 VV
 2744797
 52548856
 2.778

 10.987
 1821
 1846
 1857
 VV
 3099485
 65314948
 3.448
 22 4.40% 0.696% 23 1.26% 0.200% 24 0.438% 25 0.545% 

 11.147
 1857
 1874
 1887
 VV
 2
 821479
 25191961
 1.33%

 11.388
 1908
 1914
 1939
 VV
 5
 791981
 43224807
 2.28%

 11.676
 1939
 1963
 1980
 VV
 9
 956481
 65346653
 3.44%

 11.814
 1980
 1986
 2012
 VV
 6
 841242
 46871575
 2.47%

 12.401
 2065
 2086
 2100
 VV
 12741174
 340696570
 17.96%

 26 0.210% 27 0.361% 28 0.545% 29 0.391% 30 2.842% 2100 2114 2127 VV 3 4304895 146543880 7.738 2127 2133 2155 VV 2 794238 46704918 2.468 2155 2163 2173 VV 2494187 77774716 4.108 2173 2189 2204 VV 3 15194111 736725236 38.848 2204 2217 2246 VV 2 5180082 230823490 12.173 31 12.567 1.222% 32 12.681 0.390% 33 12.855 0.649% 34 13.008 6.146% 35 13.174 1.926% 36 13.458 2246 2265 2281 VV 6 852335 60083290 3.17% 0.501% 2281 2297 2334 VV 5 13167853 1224667854 64.56% 10.216% 2246 2265 2281 VV 6 852335 60083290 37 13.647 38 2334 2347 2353 PV 668288 12087041 0.64% 0.101% 2353 2373 2380 VV 4700683 113352161 5.98% 0.946% 13.941 39 14.095 2400 2411 2429 VV 8 1286617 59034205 40 14.322 3.11% 0.492%

말랐

1. 6. 4

*										
41 42 43 44 45	14.492 14.601 14.724 14.931 15.179	2429 2446 2465 2487 2546	9 2440 5 2458 5 2479 7 2514 5 2556	244 246 248 252 52570	6 VV 5 VV 7 VV 7 VV 7 VV 0 VV	7 5 7 7	3358927 3904252 7355404 3970773 989547	70620244 159728777 347480191 159771678 30813322	3.72% 8.42% 18.32% 8.42% 1.62%	0.589% 1.332% 2.899% 1.333% 0.257%
46 47 48 49 50	15.337 15.429 15.761 16.034 16.208	2570 2591 2616 2684 2708	2583 2599 2655 2701 2731	2591 2616 2674 2708 2738	VV VV VV VV VV	5 1 3 9	4343180 969908 0 8686491 1925472 1986657	105438005 33540325 702012627 50904683 79931143	5.56% 1.77% 37.01% 2.68% 4.21%	0.880% 0.280% 5.856% 0.425% 0.667%
51 52 53 54 55	16.342 16.424 16.557 16.700 16.827	2738 2760 2777 2797 2825	2753 2767 2790 2814 2836	2760 2777 2797 2825 2854	VV VV VV VV VV	4 5 4	2336189 4698257 1874465 1832301 747146	94516809 122444013 38513576 56060114 23826343	4.98% 6.46% 2.03% 2.96% 1.26%	0.788% 1.021% 0.321% 0.468% 0.199%
56 57 58 59 60	17.065 17.271 17.396 17.806 17.915	2854 2889 2924 2985 3013	2876 2911 2932 3001 3020	2889 2924 2949 3013 3033	PV VV PV BV VV	7 8 4 10 4	565928 456206 1111579 537451 1193534	19885949 24856836 26307888 25877536 32989192	1.05% 1.31% 1.39% 1.36% 1.74%	0.166% 0.207% 0.219% 0.216% 0.275%
61 62 63 64 65	18.094 18.214 18.554 19.153 19.227	3033 3060 3119 3218 3235	3050 3071 3128 3230 3242	3060 3088 3142 3235 3251	PV VV VV PV VV	8 9 4 3	493038 451054 948717 449400 427892	24371054 16872680 22386874 10456777 9709602	1.28% 0.89% 1.18% 0.55% 0.51%	0.203% 0.141% 0.187% 0.087% 0.081%
66	22.463	3770	3790	3807	вv	2	589658	14436467	0.76%	0.120%

Sum of corrected areas: 11987708191

SMART.M Fri Sep 11 11:07:44 2015







## Sample No : 13

```
Area Percent Report
      Data Path : C:\msdchem\1\data\Commercial 2015.08.07\
     Data File : 15.09.03.SAMPLE_C.D
Acq On : 3 Sep 2015 13:27
      Operator
      Sample
      Misc
     ALS Vial : 1 Sample Multiplier: 1
      Integration Parameters: autoint1.e
     Integrator: ChemStation
     Method
                        : C:\MSDCHEM\1\METHODS\SMART.M
     Title
                       : autoint1.e
                      : TIC: 15.09.03.SAMPLE C.D\data.ms
     Signal
  peak R.T. first max last PK peak
# min scan scan scan TY height
                                                                                   corr. corr.
area % max.
                                                                                                                      % of
                                                                                                     % max.
                                                                                                                      total
   ____
            ____
                                           - ---- ----
                                                                 -----
                                                                                                     _____
                                                                                                                      ----

        228
        237
        263
        BB
        7754719
        261734853
        42.24%

        805
        825
        840
        BV
        742984
        28384873
        4.58%

        840
        860
        885
        VB
        6834177
        282378970
        45.58%

            1,485
    1
                                                                7754719 261734853 42.24%
                                                                                                                      8.142%
    2
            4.955
                                                                                                                      0.883%
    3
            5.162
                                                                                                                      8.784%
                         980 1025 1042 BV 2 6104801 619591719 100.00%
1042 1054 1080 PB 1366499 24684309 3.98%
    Δ
            6.138
                                                                                                                   19.273%
    5
            6.312
                                                                                                                     0.768%
                        1153 1206 1242 BB 5 5789283 520309669 83.98% 16.185%
    6
            7.206

        1133
        1200
        1242
        Bit
        5
        5765205
        52030500

        1540
        1552
        1559
        PV
        904647
        18506172

        1650
        1663
        1676
        BV
        1589151
        31689679

    7
            9.249
                                                                                                    2.99%
                                                                                                                    0.576%
   8
           9.905
                                                                                                                     0.986%

        1636
        1609
        1709
        VV
        3
        519214
        16550776

        1709
        1718
        1723
        PV
        874913
        13743234

         10.118
   g
                                                                                                     2.67%
                                                                                                                     0.515%
 10
         10.228
                                                                                                    2.22%
                                                                                                                     0.428%
                       176317761789BV700285124626942.01%183718441876BB46339982614271.33%206520842099BV2590457528640798.53%209921132142VB3<647360</td>202900593.27%217121862205VV544165715714123425.36%
 11
       10.570
                                                                                                                     0.388%
 12
        10.975
                                                                                                                     0.257%
 13
        12.387
                                                                                                                     1.644%
        12.558
 14
                                                                                                                     0.631%
        12.990
 15
                                                                                                                     4.888%
 16
        13.156 2205 2214 2242 VB 5 1252154 36815837
                                                                                                     5.94%
                                                                                                                    1.145%
                       2276 2299 2326 BB 8 6073763 423124801 68.29%
2349 2367 2386 PV 1880558 35130581 5.67%
2430 2437 2443 BV 787442 13191188 2 13%
17
        13.661
                                                                                                                  13.162%
18
        14.060
                                                                                                                    1.093%
                       2430 2437 2443 BV
19
        14,471
                                                                787442
                                                                               13191188
                                                                                                    2.13%
                                                                                                                    0.410%
20
        14.586 2443 2456 2461 PV 5 1119379 38524463
                                                                                                   6.22%
                                                                                                                    1.198%
21
        14,683
                       2461 2472 2480 VV
                                                              4084526 125665909 20.28%
                                                                                                                    3.909%
22
        14.835
                      2480 2498 2503 PV
                                                                993173
                                                                              19445307
                                                                                                  3.14%
5.04%
                                                                                                                    0.605%

        2503
        2508
        2521
        VV
        3
        1508986
        31213960
        5.04%

        2570
        2579
        2585
        BV
        1741707
        30664887
        4.95%

        2616
        2654
        2667
        BV
        4678883
        235501421
        38.01%

23
       14.894
                                                                                                                    0.971%
24
        15.311
                                                                                                                    0.954%
25
       15.752
                                                                                                                    7.326%

        16.004
        2684
        2696
        2706
        BV
        2
        565076
        10672897

        16.182
        2706
        2726
        2733
        PV
        4
        714346
        17017176

        16.314
        2733
        2749
        2755
        VV
        4
        1030094
        31767750

26
                                                                                                    1.72%
                                                                                                                    0.332%
27
                                                                                                    2.75%
                                                                                                                    0.529%
28
                                                                                                    5.13%
                                                                                                                    0.988%
29
       16.386
                       2755 2761 2773 VV
                                                            2000893
                                                                               38592850
                                                                                                    6.23%
                                                                                                                    1.200%
                                                                                                  6.23% 1.200%
2.47% 0.476%
30
       16.530 2773 2785 2793 PV
                                                              859875 15289661
       16.677 2793 2810 2822 VV 4 901500 19934360
18.217 3035 3071 3087 BB 4 783220 23602907
31
                                                                                                 3.22%
                                                                                                               0.620%
0.734%
32
                                                                                                   3.81%
```

Sum of corrected areas: 3214749703

4ART.M Fri Sep 11 11:34:52 2015





### Sample No: 14

Area Percent Report Data Path : C:\msdchem\1\data\Commercial 2015.08.07\ Data File : 15.08.07.SAMPLE 14.D Acq On : 7 Aug 2015 14:42 Acq On : Operator : Sample Misc Misc : ALS Vial : 2 Sample Multiplier: 1 Integration Parameters: autoint1.e Integrator: ChemStation Method : C:\MSDCHEM\1\METHODS\SMART.M : autointl.e Title Signal : TIC: 15.08.07.SAMPLE\_14.D\data.ms peak R.T. first max last PK peak # min scan scan scan TY height 
 last
 PK
 peak
 corr.
 corr.
 % of

 scan
 TY
 height
 area
 % max.
 total

 267
 EV
 14393010
 616662446
 29.17%
 4.519%

 303
 VB
 2448102
 64279668
 3.04%
 0.471%

 842
 EV
 4527115
 334495822
 15.82%
 2.451%

 884
 VV
 4
 15004135
 1422264009
 67
 67
 -----7 1,504 208 237 2 1.731 267 275 3 4.986 801 827 
 842
 859
 884
 VV
 4
 15004135
 1422264089
 67.27%
 10.422

 884
 891
 904
 VV
 1378105
 52811859
 2.50%
 0.387%
 4 5.179 10.422% 5 5.366 6 5.501 904 914 942 VV 5 1028942 69650297 944 942 VV 5 1028942 09650297 3.29% 0.510% 942 1016 1061 VV 2 12542075 2114205066 100.00% 15.492% 1061 1072 1090 VB 3335643 88506176 4.19% 0.649% 1123 1134 1164 VB 1400463 59505487 2.81% 0.436% 3.29% 7 6.103 8 6.433 9 6.800 10 1164 1206 1241 BV 5 9593873 1347254372 63.72% 7.225 9.8728 11 7.456 1241 1245 1253 VV 5013635 142994760 6.76% 1.048% 12 7.778 7.934 1271 1300 1306 VV 483242 18060055 852372 28404394 1702681 100769826 3637232 123086463 13 0.85% 0.132% 1306 1326 1332 VV 1.34% 8.317 0.208% 14 1345 1391 1406 VV 1406 1422 1444 VV 4.77% 15 8.502 0.738% 5.82% 0.902% 16 8.677 1444 1452 1459 VV 912714 27869100 1.32% 0.204% 17 8,772 1459 1468 1482 VV 4 673862 1482 1491 1503 VV 2288213 36265238 1.72% 0.266% 18 8.910 2288213 74884318 3.54% 0.549% 19 1503 1509 1516 VV 9.013 830631 24080225 1.14% 0.176% 20 9.172 1516 1536 1542 VV 6 1974541 84815377 4.01% 0.621% 21 9.278 1542 1554 1561 VV 3908931 116177454 5.50% 0.851% 22 9.350 1561 1566 1589 VV 1376671 49532570 960699 24505597 0.363% 23 9.669 2.34% 1612 1620 1626 VV 1.16% 1650 1665 1676 VV 7864105 236729818 1676 1689 1693 VV 5 1683394 58699231 24 9.936 0.180% 11.20% 25 10.075 1.735% 2.78% 0.430% 26 10.145 1693 1701 1709 VV 4339019 103948448 4.92% 27 10.251 0.762% 1709 1718 1738 VV 2 4588389 129009170 1762 1775 1784 VV 2709713 61363472 1784 1793 1800 VV 1221287 33534404 6.10% 0.945% 28 10.583 2.90% 0.450% 29 10.692 1.59% 0.246% 30 10.771 1800 1807 1819 VV 1909805 41427984 1.96% 0.304% 31 10.987 1837 1843 1862 VV 1856110 36423709 
 1837
 1843
 1862
 VV
 1856110
 36423709

 1944
 1960
 1977
 BV
 4
 453094
 27264782

 1977
 1992
 2009
 VV
 5
 516685
 30559188

 2064
 2084
 2105
 VV
 12191544
 595147292

 2105
 2112
 2152
 VV
 3513832
 201140565
 1.72% 0.267% 32 11.674 1.29% 0.200% 13 11.865 1.45% 0.224% 14 12.410 28.15% :5 4.361% 12.575 9.51% 1.474% 6 12.862 2152 2161 2170 VV 2 2112372 84249067 3.98% 0.617% 7 2170 2186 2208 VV 3 12799629 994825840 13.014 47.05% 2208 2217 2252 VV 5 3064288 260781435 12.338 2252 2296 2319 VV 10 9679373 985001730 46.598 7.290% 8 13.193 1.911% g 13.660 0 7.218% 13,901 2319 2337 2347 VV 6 6792051 517666458 24.498 3.793%

f et d

:17

41 14.003 2347 2354 2361 VV 1109193 18798757 0.89% 0.138% 14.170 2361 2382 2396 PV 2396 2414 2424 VV 42 4819990 161290883 7.63% 1.182% 43 14.357 1247827 59395631 2.81% 0.435% 44 14.440 2424 2428 2434 VV 703638 17255458 3166222 74585505 0.82% 0.126% 45 14.532 2434 2444 2449 VV 3166222 3.53% 0.547% 
 2449
 2463
 2469
 VV
 8
 4027554
 173862346
 8.22%

 2469
 2484
 2495
 VV
 5
 4835158
 328705485
 15.55%

 2495
 2522
 2531
 VV
 3
 2981696
 157332454
 7.44%

 2573
 2584
 2591
 VV
 2540694
 50821863
 2.40%
 46 14.644 1.274% 47 14.768 2.409% 48 14.992 15.363 1.153% 49 0.372% 50 2605 2659 2672 VV 10 5914500 437304039 15,806 20.68% 3.204% 
 2672
 2699
 2706
 VV
 4
 1236261
 48276112

 2706
 2714
 2718
 VV
 678053
 15453997

 2718
 2730
 2739
 VV
 6
 2336078
 87197219

 2739
 2768
 2776
 VV
 2
 4378100
 206048974

 2776
 2787
 2795
 FV
 650967
 13722176
 51 16.039 2.28% 0.354% 52 16.125 0.73% 0.113% 53 16.224 4.12% 0.639% 54 16.446 9.75% 1.510% 55 16.561 0.65% 0.101% 

 2795
 2810
 2822
 VV
 4
 659947
 23872517

 2822
 2832
 2840
 VV
 7
 456161
 15038702

 2885
 2900
 2905
 VV
 7
 407051
 16878980

 2917
 2929
 2945
 VV
 2
 1033106
 33355613

 2945
 2964
 2980
 VV
 4
 573901
 34158945

 56 16.693 1.13% 0.175% 57 16.827 0.71% 0.110% 58 17.227 17.398 0.80% 0.124% 59 1.58% 0.244% 60 17.604 1.62% 0.250% 
 2980
 2999
 3011
 VV
 1387701

 3011
 3018
 3029
 VV
 3981190

 3029
 3048
 3056
 VV
 8
 870604

 3056
 3068
 3085
 VV
 6
 730942
 17.811 61 54968907 2.60% 0.403% 62 17.926 1.49% 31427366 0.230% 63 18.097 38260209 1.81% 0.280% 64 18.216 37218328 1.76% 0.273% 65 18.553 3116 3125 3138 VV 1156249 35651180 1.69% 0.261% 
 3246
 3252
 3266
 VV
 572630

 3454
 3483
 3499
 VV
 2
 434645
 66 19.306 572630 12563550 0.59% 0.092% 67 20.670 13604585 0.64% 0.100% 68 20.962 3522 3533 3548 VV 617647 0.57% 12006382 0.088% 3779 3787 3801 PV 5 69 22.462 444839 11421267 0.084% 70 24.687 4151 4164 4170 VV 5 396039 7890206 0.37% 0.058%

1

Sum of corrected areas: 13647220887

SMART.M Fri Sep 11 11:13:07 2015







# Sample No: 15

Area Percent Report Data Path : C:\msdchem\1\data\Commercial 2015.08.07\ Data File : 15.09.01.SAMPLE\_15.D Acq On : 1 Sep 2015 14:15 Operator Sample Misc ALS Vial : 1 Sample Multiplier: 1 Integration Parameters: autointl.e Integrator: ChemStation Method : C:\MSDCHEM\1\METHODS\SMART.M : autoint1.e Signal : TIC: 15.09.01.SAMPLE 15.D\data.ms peak R.T. first max last PK peak # min scan scan scan TY height corr. corr. % of area % max. height area % 720254 20906172 total 193 210 220 RV 720254 ----------1 1.328 210 220 BV 720254 20906172 0.97% 0.167% 243 256 VV 2 10565895 401393614 18.69% 3.205 2 1.523 220 
 220
 243
 250
 VV
 2
 10565855
 401393614
 16.056

 256
 260
 289
 VB
 2594421
 64024826
 2.98%

 633
 665
 692
 BB
 2540280
 114561442
 5.34%

 765
 828
 845
 EV
 5888879
 294562289
 13.72%
 3.205% 0.511% 3 1.622 2.98% 4 4.015 5 0.915% 4.978 2.352% 6 845 860 898 VV 2 18027246 1230151821 57.29% 898 913 955 VB 1706493 79429202 3.70% 0 5.164 
 845
 850
 858
 VV 2
 1802/246
 1230151821
 57.296
 9.0256

 898
 913
 955
 VB
 1706493
 79429202
 3.70%
 0.634%

 970
 1015
 1048
 BV
 4
 15429358
 2147182629
 100.00%
 17.146%

 1048
 1062
 1087
 PB
 4630964
 115267314
 5.37%
 0.920%

 1089
 1108
 1122
 BV
 2
 686758
 22334437
 1.04%
 0.178%
 9.823% 7 5.479 8 6.076 9 6.358 10 6.629 

 6.761
 1122
 1130
 1147
 VV
 2589304
 85901332
 4.00%

 7.170
 1147
 1200
 1236
 VV
 4
 14196725
 1689729894
 78.70'

 7.730
 1282
 1295
 1301
 BV
 600729
 12054797
 0.56%

 7.894
 1317
 1322
 1330
 PV
 945976
 15429899
 0.72%

 8.289
 1355
 1389
 1405
 BV
 1819331
 37857937
 1.76%

 11 0.686% 12 94 78.70% 13.493% 0.56% 0.096% 13 14 0.123% 15 0.302% 

 8.473
 1405
 1420
 1444
 VV
 4383070
 91801770

 8.653
 1444
 1451
 1459
 VV
 712852
 12991032

 8.886
 1482
 1490
 1503
 PV
 2298735
 55857864

 8.996
 1503
 1509
 1517
 VV
 543732
 10507044

 9.146
 1517
 1535
 1540
 VV
 62143703
 66042184

 16 4.28% 0.733% 0.61% 0.104% 17 18 2.60% 0.446% 19 0.084% 0.527% 0.49% 20 3.08% 21 9.253 1540 1553 1560 VV 
 9.253
 1540
 1553
 1560
 VV
 4660241
 110737566

 9.331
 1560
 1566
 1575
 VV
 1262560
 26214507

 9.655
 1613
 1621
 1628
 PV
 2
 712309
 13830203

 9.910
 1650
 1664
 1677
 PV
 9207795
 211060315
 4660241 110737566 5.16% 0.884% 22 1.22% 0.209% 23 0.64% 0.110% 24 9.83% 1.685% 0.389% :5 10.058 1677 1689 1694 VV 4 1604055 48678558 2.27% :6 10.123 1694 1700 1710 VV 4519949 86820652 4.04% 0.693% :7 1710 1718 1737 VV 2 4703924 104661565 10,230 4.87% 8 0.836% 10.568 1766 1775 1784 BV 
 3091144
 57905249

 1220188
 28490599

 2199422
 39738320
 2.70% 10.333 1763 1775 1784 BV 10.679 1784 1794 1802 VV 10.757 1802 1807 1820 VV 0.462% q 0.228% 1.33% 0 1.85% 1 1837 1844 1857 BV 10.974 1899342 32988922 12341190 328565381 1.54% 0.263% 15.30% 2.624% 12.398 2065 2085 2099 PV 2065 2085 2099 FV 1254115 113481835 2099 2113 2140 VV 4 3447447 113481835 2,624% З 12.564 
 5.29%
 0.906%

 0.75%
 0.128%

 2.96%
 0.508%

 12.756
 2140
 2146
 2154
 VV
 916108
 16061672

 12.853
 2154
 2162
 2173
 VV
 2083925
 63610332
 4 5 2,96% 217321892206VV31515894872341000933.69%5.777%220622172249VB3500063623368768810.88%1.866%225122622280BV4099567975513264.54%0.779%228022962336VV413204197135646134263.17%10.832%233623472354PV744983116847150.54%0.093% 6 13.008 7 13.173 8 13.439 9 13.643 D 13.945 2336 2347 2354 PV

41 14.103 2354 2374 2381 VV 4296953 108835025 
 2354
 2354
 2351
 VV
 4299933
 100630423

 2381
 2412
 2421
 VV
 6
 1067266
 51408106

 2421
 2425
 2433
 VV
 2
 603733
 15638875

 2433
 2440
 2446
 VV
 3321772
 64487040

 2446
 2459
 2465
 VV
 7
 3989633
 169429267
 5.07% 0.869% 42 14.323 51408106 2.398 0.738 3.008 0.411% 43 14.404 0.125% 44 14.492 0.515% 45 14.605 7.89% 1.353% 2465 2479 2487 VV 2 7151692 337423451 2487 2515 2526 VV 3 4519822 202211214 46 14.724 15.71% 2.694% 47 14.933 9.42% 1.615% 48 15.058 2526 2536 2545 PV 3 515826 11924737 2545 2556 2570 VV 490676 16931184 0.56% 0.095% 49 15.174 490676 16931184 0.79% 0.135% 50 15.322 2570 2581 2587 VV 2685915 51365809 2.39% 0.410% 15.395 15.755 51 2587 2593 2604 VV 3 496339 14036857 
 2507
 2533
 2604
 VV
 3
 4905339
 14050607

 2604
 2654
 2669
 VV
 2
 8511036
 464212369

 2669
 2684
 2689
 VV
 8
 443614
 22042826

 2689
 2697
 2706
 VV
 1774300
 41198409

 2706
 2712
 2717
 VV
 2
 922223
 18839824
 0.65% 0.112% 52 21.62% 3.707% 53 15.931 1.03% 0.176% 54 16.008 1.92% 0.329% 55 16.100 0.88% 0.150% 56 16.194 2717 2728 2735 VV 3874196 107821643 
 2717
 2725
 2735
 vv
 3671136
 10702131

 2735
 2764
 2774
 vv
 2
 7017461
 278153725

 2774
 2785
 2794
 vv
 1061112
 25993557
 5.02% 0.861% 57 16.405 12.95% 2.221% 58 16.530 1.21% 0.208% 59 16,670 2794 2809 2822 VV 5 1030216 39350251 1.83% 0.314% 60 2822 2833 2841 VV 5 610281 21630592 16.810 1.01% 0.173% 17.381 17.794 61 2918 2929 2954 PB 892117 20861566 0.97% 0.167% 
 2982
 3000
 3012
 PV
 1246466

 3012
 3020
 3031
 VV
 2
 738674

 3031
 3049
 3057
 PV
 7
 615105

 3057
 3069
 3086
 VV
 7
 507412
 62 1246466 34033284 1.59% 0.272% 63 17.914 20388162 0.95% 0.163% 64 18.086 22255146 1.04% 0.178% 65 18.203 20805771 0.97% 0.166% 66 18.543 3121 3126 3150 VV 2 635724 21345698 0.99% 3403 3486 3502 BV 4 3502 3536 3552 VV 0.170% 67 20.667 0.78% 378196 16784002 0.134% 0.123% 68 20.960 485714 15431576 69 24,682 4130 4166 4173 PV 4 372223 0.118% 14764637 0.69%

1.1

Sum of corrected areas: 12523232857

MART.M Fri Sep 11 11:17:11 2015



Instru Sample	ed : 1 ment : Name:	. Sep 2019 UOSJP GCI	5 14:15 4SD	using	AcqMethod	PLANT EXTRACTS.M		
Misc In Vial Nu	nfo : umber: 1					ERR		
Abundanci 1.8e+07	e 86 26 36	96 16 16 16 16 16 16 16 16 16 16 16 16 16	66 23	58856 a	15.09.01 SAMPI හි සී සීසි	E_15.D\data.ms 중 없었다양 중중당 4	27	50 8 43 <u>8</u>
1.7e+07	(3R-trans)- Cubebene Copsene	ieta.,8S*)]- Cubebene nethylene- voohvliene	yophyllene vene- (Z)-	/opnyllene -4-metity ra.,7aS 8a.beta.)- 3a.alpha.)- anyl)-, (S)-	(1S-cis) /ophyllene: /3/8-triene: 3R-trans)-:	ene oxide: b.alpha,)]-: ene oxide: ba.alpha,)-: enpt-2-ene: enpt-2-ene: decatriene: -trimethyl-:	sthy⊦, (E)-:	4-methyl-: xen-1-yl)-: sthyl-, (Z)-: I-hydroxy-:
1.6e+07	lethy!)-, ( .alpha	na.,/a.b .alpha ethyl-9-r Can	Can 3-methy	nacan hexenyl) (a.,7.alph 7.alpha.,8 alpha.,8	ıylethyl)- haCary ila-1(10) tenyl)-∠( alphaC	aryophyll aryophyll arjophyl arjopha (2.2.1)h (2.2.1)h ta. beta., t .E-7-Doc	2,6-trime	idinone, -cyclohe (yloxy)-N
1.5e+07	1-methy	an 3.008 	dimethyl	aip ta, A, bet ta, A, bet ta, A, bet ta, A, bet ta, A, a ta, A, a ta, A, a ta, A, a ta, A, a ta, A, a ta, A, bet ta, A, b, a,	-(1-meth -alp Cada Tethyleth	Ca pha73 Julpha45 Julpha.,45 Julpha.,4 Quino 1,Z-5 taidehyo	ienyl)-2,	5H)-Pter methyl-1 2-one, 6 a. 4-(hex
1.4e+07	enyl)-1-(	2-isopr	ie, 7,11-	-(1,5-dirr 14.,35.be [4aR-(4a 1yl)-, (1.s ethyl-1-r	methyl-1 yl-6-(1-n	pha7.al WI)-, (1.a thyl-2-vit alpha.,4 alpha.,4	,3-butad	6( 4-(6,6-dii ecadien- nidamidi
1.3e+07	nethyleth	ec-1-ene	13.643 te	nzene_1 -(3a.aipt theny!)-, nethyiett /I-4-(5-m	Iro-4,7-di	iR-(1a.al nethylett ,7-Trime ,1S-(1	methyl-1	h-2-one, 5.9-Und ecarboxi
1.20/07	12,598	14.4.0]de	6,10-Doc	/)-, [3aS methylet vyL-1-(1-r 1-methy	hexahyd Iohexene	thy!-, [1a 3e-1-(1-r; 1,7 hylethyl) 3-Cy	1-(2,3-di	3-Buter Benzene
1.28707	<del>  4-meth</del>	Bicyclo		ethylethy ane-7-(1- 7-dimeth	3,5,6,8a- Cyc	-tetrame methylei 4-(1-inet	leptane,	-
1.1e+U/	f-othamy thun, ri			e-4-(1-m -methyle hydro-4 Cycl	sne, 1,2,	-1, 1, 7, 78 nethyl-4	o[4.1.0]ł	
1e+07	<del>16X6116, .</del> methole			aethylen methyl-1 8a-hexa	aphthale	tahydro- ydro-7-n ďro-1,6-c	xabicycl	
9000000				athyl-3-n ydro-4a- 2,4a,5,6	z	15.755 Ba-octal - octal	0-7	
8000000	Za-dimet			dro-7-m e, decah alene, 1,		,3,5,6,7, ,4a,5,6,1 ,a,7,8,8e		andro angelo
7000000	Tvdro-1			Naphtra	9	ne, 1a,2 1,2,3,4 1,2,3,491		an a
6000000	ne. octa		1	benzene Nap		aphthale hthalene hatenol <sub>7</sub>		
5000000	1H-inde	13.173	14 1	[7] [1] [1]	4.933	ropa[a]n Nap A- <del>Naphi</del>		
1000000	detheno.	13.43	3			용 · · · · · · · · · · · · · · · · · · ·		
000000	12.504 4 7			14-432	15 322	<b></b>		
2000000	12.8	3		[H-Cycle	, J.JLL	16.008		
	12.756			14.323		16.580670	17.381	17.794

-



Area Percent Report Data Path : C:\msdchem\1\data\Commercial 2015.08.07\ .Data File : 15.08.25.SAMPLE 16.D Acq On : 25 Aug 2015 13:27 Operator : Sample Misc ALS Vial : 1 Sample Multiplier: 1 Integration Parameters: autoint1.e Integrator: ChemStation Method : C:\MSDCHEM\1\METHODS\SMART.M Title : autointl.e : TIC: 15.08.25.SAMPLE\_16.D\data.ms Signal peak peak R.T. first max last PK corr. corr. % of area % max. total # min scan scan scan TY height \_\_\_\_ \_\_\_\_\_ 283 BB 8268532 300625704 15.34% 2.670% 3696791 240062564 12.25% 2.132% 1 1.507 208 241 
 200
 211
 200
 BV
 3696791
 240062564
 12.15%
 2.132%

 845
 862
 900
 VV
 216648296
 1211719576
 61.81%
 10.760%

 900
 914
 947
 VB
 2
 1104605
 56548605
 2.88%
 0.502%

 970
 1017
 1052
 BV
 3
 13901653
 1960348224
 100.00%
 17.409%
 2 4,980 3 5.173 4 5.483 5 6.090 
 1052
 1064
 1074
 PB
 3040841
 66995316
 3.42%
 0.595%

 1122
 1133
 1160
 VB
 3108450
 111362531
 5.68%
 0.989%

 1160
 1226
 1238
 BV
 5
 10217339
 1312118465
 66.93%
 11.652%

 1160
 1206
 1238
 BV
 5
 10217339
 1312118465
 66.93%
 11.652%
 6 6.371 -6.777 8 7.208 9 7.903 1306 1324 1331 BV 466313 7660003 0.39% 0.068% 8.295 1364 1390 1407 BV 3 858732 22898263 10 0.203% 1.17% 8.484 3315441 73636410 633639 11474536 11 1407 1422 1445 VV 73636410 3.76% 0.654% 
 1445
 1452
 1459
 VV
 633639

 1484
 1492
 1505
 VV
 3
 2029275
 12 8.662 0.59% 0.102% 13 8.897 51990207 2.65% 0.462% 14 9.002 1505 1510 1518 VV 489067 10484375 0.53% 0.093% 15 9.161 1518 1537 1543 VV 4 1685117 56381045 2.88% 0.501% 16 9.266 1543 1555 1562 VV 3642095 94458119 4.82% 0.839% 1562 1568 1588 VV 2 1153342 17 9.342 27662408 1.41% 0.246% 
 1582
 1683
 1686
 VV
 1103342
 27002400
 11110

 1588
 1622
 1628
 VV
 5
 679617
 24725301
 1.26%

 1653
 1667
 1678
 VV
 2
 7874542
 221208928
 11.28%
 18 9,662 0.220% 19 9.929 1.964% 1678 1691 1696 VV 5 1500357 52063477 20 10.070 2.66% 0.462% 10.138 21 1696 1703 1712 VV 3663641 78032642 3.98% 0.693% 22 10.244 1712 1720 1738 VV 2 4557143 112875514 5.76% 1.002% 23 10.580 1767 1777 1787 VV 2803087 55553082 1787 1796 1803 VV 949403 23834503 2.83% 0.493% 24 10.688 949403 23834503 1.22% 1821692 34353493 1.75% 0.212% 25 10.768 1803 1809 1821 VV 0.305% 26 10,984 1839 1846 1857 BV 1963363 34562525 1.76% 0.307% 
 1839
 1846
 1857
 BV
 1963363
 34362323
 1.700

 1857
 1874
 1887
 PV
 5
 489023
 12650944
 0.65%

 1943
 1964
 1980
 BV
 8
 408327
 20720685
 1.06%

 1980
 1996
 2008
 VV
 4
 498710
 17245270
 0.88%

 2069
 2087
 2101
 BV
 13145143
 411705158
 21.00%
 27 11.148 0.112% 28 11.679 0.184% 29 11.869 0.153% 30 12.407 2069 2087 2101 BV 3.656% 2101 2115 2144 VV 3 3634605 160836131 2144 2164 2175 VV 2 2100872 81428808 31 12.572 8 20% 1 4288 12.860 32 4.15% 0.723% 7.245% 
 2144
 2164
 2164
 2164
 2164
 2165
 6.725

 2175
 2189
 2208
 VV
 3
 14735699
 815887653
 41.62%
 7.245%

 2208
 2218
 2249
 VV
 3
 4014769
 226163930
 11.54%
 2.008%

 2249
 2298
 2339
 VV
 5
 11781114
 1354944029
 69.12%
 12.032%
 33 13.011 34 13.18135 13.651 36 13.964 2339 2351 2357 PV 611501 9197203 0.47% 0.082% 
 2357
 2378
 2394
 VV
 4074231
 113380802

 2394
 2414
 2435
 VV
 9
 968916
 55896424

 2435
 2443
 2449
 VV
 2780262
 57007233
 37 14,128 5.78% 1.007% 38 14.337 2.85% 0.496% 39 14.508 2.91% 0.506% 40 14.617 2449 2461 2468 VV 7 3607648 144099226 7.35%

1.280%

-										
41 42	14.748 14.902	2468 2492	2483 2510	2492 2513	vv vv		5587663 2383584	290847004 69719548	14.84%	2.583%
43	14.957	2513	2519	2530	vv	2	3344353	79203656	4 04%	0.0198
44	15.189	2548	2558	2574	vv		454947	14800234	0.75%	0.131%
45	15.344	2574	2584	2591	vv		2795419	53265124	2.72%	0.473%
46	15.419	2591	2597	2607	vv	2	523373	13610842	0.69%	0.121%
4/	15.791	2607	2660	2673	VV	8	6769265	473217162	24.14%	4.202%
48	16.028	2692	2700	2708	VV	2	1389216	31889736	1.63%	0.283%
49	16.114	2708	2715	2720	٧V		586553	12434357	0.63%	0.110%
50	16.213	2720	2732	2739	۷V	5	2372752	78196463	3.99%	0.694%
51	16.346	2739	2754	2759	vv	6	1648094	71426914	3.64%	0.634%
52	16.427	2759	2768	2778	VV		4475437	129143652	6.59%	1,147%
53	16.552	2778	2789	2797	PV		1169649	21451260	1.09%	0.190%
54	16.691	2797	2813	2824	VV		1186665	34170051	1.74%	0.303%
55	16.823	2824	2835	2844	VV	5	443011	13598963	0.69%	0.121%
56	17.393	2919	2932	2948	ΡB	2	726344	15931056	0.81%	0.141%
57	17.806	2983	3001	3014	PV	3	1116567	32514049	1.66%	0.289%
58	17.919	3014	3021	3033	VV	5	746776	22485211	1.15%	0.200%
59	18.095	3033	3050	3059	ΡV	3	614398	21815229	1.11%	0.194%
60	18.213	3059	3070	3088	VV	9	584144	22527900	1.15%	0.200%
61	18.551	3122	3128	3152	vv	3	717188	23764556	1.21%	0.211%

Sum of corrected areas: 11260812283

MART.M Fri Sep 11 11:23:56 2015




#### Sample No: 17

```
Area Percent Report
        Data Path : C:\msdchem\1\data\Commercial 2015.08.07\
       Data File : 15.08.25.SAMPLE 8.D
Acq On : 25 Aug 2015 11:04
        Operator
        Sample
                                 :
        Misc
       ALS Vial : 1 Sample Multiplier: 1
       Integration Parameters: autointl.e
Integrator: ChemStation
                                : C:\MSDCHEM\1\METHODS\SMART.M
       Method
       Title ·
                            : autoint1.e
                            : TIC: 15.08.25.SAMPLE_8.D\data.ms
       Signal
                                                                                                    corr. corr. % of
area % max. total
    peak R.T. first max last PK
                                                                                       peak
         # min scan scan scan TY height

        208
        237
        289
        BB
        12952997
        714576402
        43.85%
        7.607%

        800
        824
        839
        BV
        3669912
        149765020
        9.19%
        1.594%

        839
        857
        896
        VV
        3
        15811472
        836533317
        51.33%
        8.906%

        836
        909
        927
        VB
        1015610
        34337079
        2.11%
        0.366%

        983
        1011
        1045
        VV
        3
        12895591
        1629690064
        100.00%
        17.350%

                                              ---- --- ----
      1
                1.485
      2
                 4.949
       3
                 5.148
       4
                5.451
      5
                6.058

        1045
        1056
        1062
        PB
        4677798
        105222457
        6.46%
        1.120%

        1081
        1104
        1117
        BV
        3
        897832
        37744758
        2.32%
        0.402%

        1117
        1126
        1140
        VV
        1245148
        33589245
        2.06%
        0.358%

        1165
        1199
        1226
        BV
        3
        12690457
        1319151710
        80.94%
        14.044%

      б
                6.319
      7
                6.605
     8
                6.734
      9
                7.164
   10
                                 1226 1253 1279 VV 2 779054 47618031 2.92% 0.507%
                7.484

        1310
        1319
        1328
        BV
        745995
        12712222

        1328
        1387
        1404
        VV
        1207478
        24269924

        1404
        1418
        1443
        VB
        2
        1886596
        40170164

   11
               7.876
                                                                                                                                     0.78%
                                                                                                                                                        0.135%
   12
               8.273
                                                                                                                                    1.49%
                                                                                                                                                         0.258%
   13
               8.459
                                                                                                                                    2.46%
0.74%
                                                                                                                                                        0.428%

        1481
        1489
        1502
        EV
        5
        470054
        12051200
        0.74%

        1538
        1550
        1558
        VV
        2
        823324
        17556451
        1.08%

   14
               8.875
                                                                                                                                                         0.128%
  15
               9.236
                                                                                                                                                      0.187%
  16
              9.892
                                                                               3034933 55733241
1594152 27586756
                                1631 1661 1674 BV
                                                                                                                                    3 42%
                                                                                                                                                        0.593%
                              1693 1698 1708 VV
  17
           10.109
                                                                                                                                    1.69%
                                                                                                                                                        0.294%
  18
           10.216
                                1708 1716 1735 VB
                                                                                974005 23236421
595824 9203226
                                                                                                                                    1.43%
                                                                                                                                                        0.247%
  19
           10.558
                               1749 1774 1782 BV
                                                                                                                                    0.56%
                                                                                                                                                        0.098%
  20
           10.672
                              1782 1793 1801 PV
                                                                                  575204 10734888 0.66%
                                                                                                                                                        0.114%
 21
           10,970
                               1837 1843 1853 BV
                                                                                                          6970348
                                                                                    428134
                                                                                                                                   0.43%
                                                                                                                                                      0.074%
 22
                               2066 2085 2099 BV 13061272 332871882 20.43%
           12.396
                                                                                                                                                       3.544%

        2099
        2113
        2142
        PV
        4
        2681001
        71207589
        4.37%

        2142
        2162
        2172
        VV
        1776092
        47755347
        2.93%

        2172
        2187
        2206
        VV
        2
        15725656
        600076486
        36.82%

 23
           12.560
                                                                                                                                                       0.758%
 24
           12.850
                                                                                                                                                       0.508%
 25
           12.997
                                                                                                                                                        6.388%
 26

      2206
      2215
      2246
      VB
      6
      5299795
      198628278
      12.19%
      2.113%

      2280
      2295
      2331
      VV
      5
      14434826
      1270990754
      77.99%
      13.531%

      2331
      2346
      2352
      FV
      1059234
      17098402
      1.05%
      0.182%

      2352
      2371
      2378
      VV
      5041466
      117512436
      7.21%
      1.251%

      2376
      2410
      2419
      VV
      8
      1008308
      48120605
      2.95%
      0.512%

          13.165
                               2206 2215 2246 VB 6 5299795 198628278 12.19%
 27
           13.636
 28
           13.935
 29
           14.083
 30
          14.311 2378 2410 2419 VV 8 1008308 48120605
31
          14.397
                              2419 2424 2432 VV
                                                                                1110388 23133224
3968254 71957274
                                                                                                                               1.42%
                                                                                                                                                      0.246%
32
                              2432 2438 2445 VV
          14.482
                                                                                                                                   4.42%
                                                                                                                                                      0.766%

        2445
        2459
        2464
        VV
        8
        4088019
        185862436
        11.408

        2464
        2476
        2484
        VV
        8091428
        315400428
        19.358

        2484
        2502
        2506
        VV
        4368089
        111145309
        6.828

33
          14.605

        4088019
        185862436
        11.40%
        1.979%

        8091428
        315400428
        19.35%
        3.358%

        4368089
        11145309
        6.82%
        1.183%

                                                                                                                                                      1.979%
34
          14.703
35
          14.856
36
          14,915
                              2506 2512 2525 VV 2 4869079 103578244
                                                                                                                                  6.36%
                                                                                                                                                      1.103%
                             2525 2534 2542 PV 2 638190 14649828
2566 2578 2585 PV 4 1001316 23113145
37
          15.048
                                                                                                                                0.90%
                                                                                                                                                      0.156%
38
         15.305
                                                                                                                                  1.42%
                                                                                                                                                      0.246%

        2560
        2576
        2567
        267
        267
        2681
        2687
        VV
        3
        469741
        17346179
        1.06%

         15.708
39
                                                                                                                                                      2.354%
40
         15.912
                                                                                                                                                   0.185%
```

đ

41 42 43 44 45	15.992 16.092 16.179 16.390 16.519	2687 2704 2716 2734 2772	2694 2711 2726 2762 2784	2704 2716 2734 2772 2792	VV VV VV VV PV	4 3 4	880641 1298790 4228035 8594298 389304	24332019 22512940 96836753 269698625 10068343	1.49% 1.38% 5.94% 16.55% 0.62%	0.259% 0.240% 1.031% 2.871% 0.107%
46 47 48	17.372 17.787 18.540	2918 2974 3082	2928 2998 3126	2939 3004 3150	PV BV BB	2 2	419472 469254 422455	7475976 7385843 14751677	0.46% 0.45% 0.91%	0.080% 0.079% 0.157%

Sum of corrected areas: 9393083064

SMART.M Fri Sep 11 10:47:06 2015





## Sample No : 18

				Area Percent Report
	Data Pat Data Fil Acq On Operator Sample Misc	h : C:\msd e : 15.08. : 7 Aug :	chem\1\dat 07.SAMPLE 2015 13	ta\Commercial 2015.08.07\ _10.D :24
	ALS Vial	:2 Sa	mple Multi	iplier: 1
	Integrat Integrat	ion Parame or: ChemSt	ters: auto ation	Dintl.e
1	Method Title	: C:\MSDG : autoint	CHEM\1\MET t1.e	IHODS\SMART.M
:	Signal	: TIC: :	15.08.07.S	SAMPLE 10.D\data.ms
pe	eak R.T # min	. first ma scan sca	ax last P an scan T	- PK peak corr. corr. % of PY height area % max. total
1	1.478	3 205 23	21 261 BV	12273964         458763387         23.02%         3.178%           2         5260361         400311743         20.09%         2.773%           2         14948334         1556075360         78.09%         10.779%           3         1030953         72169729         3.62%         0.500%           2         12198358         1992657117         100.00%         13.803%
2	4.96	796 82	22 837 BV	
3	5.155	837 85	4 896 VV	
4	5.488	896 91	0 945 VB	
5	6.085	964 101	2 1050 BV	
6	6.343	1050 105	5 1062 VV	5813790         176470703         8.86%         1.222%           3509396         103690173         5.20%         0.718%           451031         15001215         0.75%         0.104%           1776600         75689856         3.80%         0.524%           5         9301447         1232149457         61.83%         8.535%
7	6.442	1062 107	2 1080 VV	
8	6.669	1080 111	1 1123 PV	
9	6.796	1123 113	2 1155 VB	
10	7.218	1162 120	4 1236 BV	
11	7.472	1236 124	7 1254 VV	5108701         246692269         12.38%         1.709%           452982         7344328         0.37%         0.051%           2         1447807         27190565         1.36%         0.188%           2105024         41930317         2.10%         0.290%           4831494         110733559         5.56%         0.767%
12	7.781	1284 129	9 1306 PV	
13	7.939	1306 132	6 1345 PB	
14	8.316	1382 139	0 1404 PV	
15	8.503	1404 142	1 1442 PV	
16	8.676	1442 1451	L 1458 VV	948156 16834099 0.84% 0.117%
17	8.768	1458 1460	5 1480 VV	4 548987 15899263 0.80% 0.110%
18	8.906	1480 1489	9 1502 VV	3 2285892 60073035 3.01% 0.416%
19	9.009	1502 1507	7 1516 VV	775186 15652746 0.79% 0.108%
20	9.166	1516 1534	1539 VV	6 1931385 71558077 3.59% 0.496%
21	9.273	1539 1552	1558 VV 2	2       3977126       104630296       5.25%       0.725%         5       1140818       29269026       1.47%       0.203%         7       1033464       45693149       2.29%       0.317%         2       7546000       198606599       9.97%       1.376%         4       1880409       62119084       3.12%       0.430%
22	9.344	1558 1564	1574 VV 2	
23	9.665	1610 1618	1649 VV 2	
24	9.924	1649 1662	1673 VV 2	
25	10.070	1673 1687	1691 VV 4	
26	10.139	1691 1698	1707 VV	4763020       105997425       5.32%       0.734%         2       4955507       149683550       7.51%       1.037%         2979307       67833100       3.40%       0.470%         2036598       51857327       2.60%       0.359%         2089993       46062855       2.31%       0.319%
27	10.244	1707 1716	1747 VV 2	
28	10.577	1760 1773	1782 VV	
29	10.690	1782 1792	1799 VV	
30	10.768	1799 1805	1816 VV	
31	10.984	1835 1842	1853 VV	2333012 49929916 2.51% 0.346%
32	11.148	1853 1869	1884 VV 3	3 617099 25898134 1.30% 0.179%
33	11.669	1932 1958	1975 VV 8	3 725946 63028288 3.16% 0.437%
34	11.812	1975 1982	2009 VV 2	2 610232 40213327 2.02% 0.279%
35	12.420	2064 2085	2105 VV	15060210 823868199 41.35% 5.707%
36	12.580	2105 2112	2150 VV	3922667 227159316 11.40% 1.574%
37	12.867	2150 2161	2170 VV	2481573 105046569 5.27% 0.728%
38	13.011	2170 2185	2211 VV 4	13067049 1144318896 57.43% 7.927%
39	13.197	2211 2216	2258 VV 4	2898213 278280250 13.97% 1.928%
40	13.663	2258 2295	2319 VV 6	5 9013088 921874081 46.26% 6.386%

ļ

nen di Angelera

Sum of corrected areas: 14436072433

SMART.M Fri Sep 11 10:51:47 2015







## Sample No : 19

							Area Pe	ercent Re	port
Dat Dat Acq Ope Sam	a Path a File On rator ple	: C:\r : 15.( : 7 ! :	nsdche 09.07 Sep 20	em\1\0 .SAMPI )15 :	data\ LE_F. 14:33	\Commercia .D 3	al 2015.08.	07\	
Mis ALŞ	c Vial	: : 1	Samp:	le Mu	ltip]	lier: 1			
Int Int	egratio egrato:	on Para	ameter nStat:	rs: au ion	ıtoir	ntl.e			
Met Tit	hod le	: C:\1 : auto	ISDCHI	EM\1\1 .e	4ETHC	DDS\SMART.	. M		
Sig	nal	: TIC	<b>:</b> 15	.09.0'	7.SAN	MPLE_F.D\c	lata.ms		
peak #	R.T. min	first scan	max scan	last scan	PK TY	peak height	corr. area	corr. % max.	% of total
1 2 3 4 5	1.500 4.970 5.165 6.087 6.350	230 796 843 968 1045	239 827 860 1016 1061	278 843 876 1045 1070	BB BV VB BV PB	6951457 1438494 3 6462380 2 5636920 2046302	347271456 67687662 372454496 735426029 43679370	47.22% 9.20% 50.64% 100.00% 5.94%	9.577% 1.867% 10.272% 20.282% 1.205%
6 7 8 9 10	7.187 8.285 8.470 9.901 10.116	1127 1365 1404 1600 1675	1203 1389 1420 1662 1699	1234 1404 1444 1675 1708	BB 4 BV VB BV PV	4 5240131 504251 722897 988405 501143	580469344 8980229 13528278 18981065 10043045	78.93% 1.22% 1.84% 2.58% 1.37%	16.009% 0.248% 0.373% 0.523% 0.277%
11 12 13 14 15	12.404 12.562 12.853 13.007 13.170	2053 2099 2143 2173 2206	2086 2113 2162 2189 2216	2099 2142 2173 2206 2246	BV VB 2 BV 4 VV VB 6	4582408 2 697126 4 441911 5560677 6 1479353	115058662 20396366 12768380 210924519 61331133	15.65% 2.77% 1.74% 28.68% 8.34%	3.173% 0.563% 0.352% 5.817% 1.691%
16 17 18 19 20	13.659 14.096 14.489 14.615 14.717	2246 2353 2433 2446 2466	2299 2373 2440 2461 2478	2333 2389 2446 2466 2485	BV VV VV VV VV VV	6 5447363 1544671 955233 5 1224075 3128166	559198979 35345699 17243872 51396910 125584499	76.04% 4.81% 2.34% 6.99% 17.08%	15.422% 0.975% 0.476% 1.417% 3.464%
21 22 23 24 25	14.868 14.926 15.711 16.182 16.397	2485 2508 2598 2717 2733	2504 2514 2647 2726 2763	2508 2526 2667 2733 2773	VV VV 2 BV VV VV	1388347 2 1811349 1534058 917853 2527706	37532912 39158185 45443076 21280197 74734741	5.10% 5.32% 6.18% 2.89% 10.16%	1.035% 1.080% 1.253% 0.587% 2.061%

Sum of corrected areas: 3625919104

SMART.M Fri Sep 11 11:43:18 2015





#### Sample No :20

Area Percent Report Data Path : C:\msdchem\l\data\Commercial 2015.08.07\ .Data File : 15.09.03.SAMPLE A.D Acq On : 3 Sep 2015 12:53 Operator : Sample Misc ALS Vial : 1 Sample Multiplier: 1 Integration Parameters: autointl.e Integrator: ChemStation Method : C:\MSDCHEM\1\METHODS\SMART.M Title : autoint1.e : TIC: 15.09.03.SAMPLE\_A.D\data.ms Signal peak R.T. first max last PK peak # min scan scan scan TY height corr. corr. % of area % max. total 
 1
 1.228
 162
 193
 213
 BB
 2
 7153110
 264252008
 50.05%
 7.212%

 2
 4.935
 805
 821
 843
 BB
 5530559
 184712737
 34
 98%
 5
 6
 4.13
 805 821 843 BB 5530559 184712737 34.98% 5.041% 955 996 1020 BV 4 6642487 495201504 93.78% 13.514% 5.968 3 
 6.146
 1020
 1026
 1052
 VB
 453750
 9039382
 1.71%
 0.247%

 7.056
 1163
 1180
 1207
 VB
 5744841
 258787786
 49.01%
 7.063%
 4 5 

 8.403
 1389
 1409
 1431
 BV
 1211740
 27590079
 5.23%

 8.843
 1472
 1483
 1507
 PV
 7
 695558
 25579089
 4.84%

 9.202
 1533
 1544
 1553
 VV
 5
 1280500
 36935608
 7.00%

 9.914
 1642
 1665
 1673
 BV
 2
 4668306
 169377312
 32.08%

 10.029
 1673
 1684
 1690
 VV
 4
 429913
 15732414
 2.98%

 б 0.753% 7 0.698% 8 1.008% 9 4.622% 10.029 10 0.429% 

 10.127
 1690
 1701
 1708
 VV
 2576568
 72469253
 13.72%

 10.220
 1708
 1716
 1734
 VV
 2
 1509101
 32980608
 6.25%

 10.550
 1759
 1772
 1781
 EV
 2
 622737
 11153929
 2.11%

 10.957
 1819
 1841
 1852
 EV
 534300
 10085401
 1.91%

 12.403
 2064
 2086
 2099
 EV
 5456602
 171201658
 32.42%

 11 1 978% 12 0.900% 13 0.304% 14 0.275% 15 4.672% 

 12.559
 2099
 2113
 2126
 VV
 2
 1259973
 40196182
 7.61%
 1.097%

 12.846
 2141
 2161
 2171
 VV
 3
 721207
 23901103
 4.53%
 0.652%

 13.012
 2171
 2189
 2206
 VV
 2
 5541972
 255322901
 48.35%
 6.968%

 13.173
 2206
 2217
 2246
 VB
 4
 1514028
 66476446
 12.59%
 1.814%

 16 17 18 19 20 2279 2299 2331 VV 5 5158028 528023608 100.00% 14.410% 13.661 
 2361
 2374
 2381
 BV
 1778753
 40912562
 7.75%
 1.117%

 2381
 2412
 2432
 PV
 5
 547527
 27903543
 5.28%
 0.762%

 2432
 2441
 2447
 VV
 1242574
 26416372
 5.00%
 0.721%

 2447
 2461
 2466
 VV
 3
 1672426
 58936664
 11.16%
 1.60%
 14.100 21 22 14.327 23 14.500 24 14.616 2466 2482 2494 VV 4 3136859 170054429 32.21% 25 14.738 4.641% 26 14.946 2494 2517 2530 VV 2 1572196 54447800 10.31% 1.486% 
 15.190
 2530
 2572
 PV
 401499
 13802168
 2.61%

 15.353
 2572
 2586
 2593
 PV
 1812502
 42800977
 8.11%
 27 0.377% 1.168% 9.459% 28 8.11% 29 15.798 2625 2661 2699 BV 9 3459696 346589164 65.64% 16.077 2699 2709 2724 VV 2 639990 15763681 2.99% 30 0.430% 
 16.232
 2724
 2735
 2745
 PV
 2
 833220
 24050405
 4.33%

 16.472
 2745
 2775
 2784
 VV
 2
 1565449
 93284144
 17.67%

 2745
 2707
 2804
 VV
 2
 747424
 15323631
 2.90%
 31 4.55% 0.656% 32 2.546% 33 0.418% 
 2804
 2822
 2834
 PV
 4
 685430
 22498119

 3016
 3024
 3037
 VV
 427373
 12440316
 34 4.26% 0.614% 2.36% 0.340% 16.746 17.941 35

Sum of corrected areas: 3664242982

MART.M Fri Sep 11 11:28:57 2015





# **APPENDIX E**

# Table 1: Mean composition of essential oil derived from black pepper dried at different conditions

NT		Mean composition % (w/w)					
	Component		5	Spouted b	ped dried	-	
Ŭ		Sun -dried	45°C	55°C	65°C	75°C	
1	Cada-1,4-diene	0.026	-	-	-	-	
2	cis β-Famesene	0.016	-	-	-	-	
3	cis-Limonene oxide	_	-	0.101	-	-	
4	cis-β-Terpineol	0.192	-	-	0.017	-	
5	E-Piperitol	0.134	-	-	-	-	
6	L-4-terpineol	2.830	0.381	0.509	0.457	0.662	
7	α-Terpinolene	0.221	-	-	-	-	
8	β-Phorone	-	-	0.153	-	-	
9	β-Sabinyl acetate	-	-	-	0.216	-	
10	(E)-p-2,8-Menthadien-1-ol	-	-	0.337	-	-	
11	3-p-Menthene	-	-	3.398	-	-	
12	cis-Geraniol	-	-	0.052	0.059	-	
13	Cyclene	-	-	-	0.046	-	
14	Cyclosativene	-	-	0.078	-	-	
15	E,Z Ocimene	0.075	-	-	-	-	
16	Geraniol	-	-	0.043	-	-	
17	Limonene-1,2-diol	-	1.246	-	-	-	
18	m-Toluquinaldine	-	-	-	0.054	-	
19	Myrtenal	-	0.975	-	-	-	
20	Ocimene	-	-	0.045	0.046	-	
21	α- Terpinolen	-	-	0.163	-	-	
22	α-Elemene	0.252	-	-	-	-	

		Mean composition % (w/w)						
No	Component		:	Spouted bed dried				
		Sun -dried	45°C	55°C	65°C	75°C		
23	α-Gurjunene	-	-	0.064	-	-		
24	β-Farnesene	-	-	-	0.021	-		
25	β-Selinene	0.103	0.089	0.162	0.256	-		
26	β-Thujene;	-	0.129	-	-	-		
27	δ-Elemene	1.651	1.411	1.978	3.213	4.580		
28	(-)-4-Terpineol	-	-	-	0.504	-		
29	(-)-beta-Pinene	-	-	-	-	-		
30	(-)-Carvone	-	0.149	0.208	-	-		
31	(-)-cis-Sabinol	0.120	-	-	-	-		
32	(-)-Myrtenol	0.647	0.523	1.378	0.643	0.578		
33	(-)-Terpinen-4-ol	-	-	0.931	-	-		
34	(+)-3-Carene	-	1.532	-	0.115	-		
35	(+)-4-Carene	4.712	0.114	0.065	0.272	0.143		
36	(+)-cycloisosativene	-	0.128	-	-	-		
37	(+)-Cyclosativene	-	-	-	-	0.188		
38	(+)epi-Bicyclosesquiphellandrene	0.173	0.099	0.190	0.165	-		
39	(1R)-alpha-Pinene	5.216	4.097	2.507	5.608	6.982		
40	(Cyclopropyl)trivinylsilane	-	-	0.065	-	-		
41	(E)-p-2-Menthen-1-ol	-	-	-	0.247	-		
42	(E)-p-Menth-2,8-dien-1-ol	0.233	0.167	0.608	0.560	0.235		
43	(E)-β-Famesene	-	-	0.033	-	0.050		
44	(R)-α-Pinene	-	-	-	2.541	-		
45	(S)-cis-Verbenol	-	-	-	-	0.052		
46	[1.2.4]Triazolo [1,5-a]pyrimidine-6- carboxylic acid, 7 amino-, ethyl ester	-	0.465	-	-	-		
47	1, 5-cyclooctadiene 1,2 -dimethyl	-	-	0.107	-	-		
48	1,3,5-Hexatriene,3-methyl	0.146	-	-	-	-		

		Mean composition % (w/w)					
No	Component		;	Spouted b	oed dried		
		Sun -dried	45°C	55°C	65°C	75°C	
49	1,5-di(ethenyl)-3-methyl-2- methylidenecyclohexane	-	-	-	0.078	-	
50	1,7,7-Trimethyl-2-vinylbicyclo [2.2.1]hept-2-ene	0.516	0.231	-	0.223	-	
51	10,10-dimethyl-2,6-dimethyl bicyclo [7.2.0]undecane-5.betaol	0.181	0.186	-	-	-	
52	10,10-Dimethyl-2,6-dimethylenebicyclo [7.2.0]undecan-5β-ol	0.324	0.498	-	-	-	
53	1-Allyl-2-hydroxy-6-methyl- cyclohexanecarboxylic acid	-	0.000	-	-	-	
54	1-Chlorosulfonyl-3-methyl-1-azaspiro [3.5]nonan-2-one	-	-	-	0.028	-	
55	1-Ethenyl-1-methyl-2,4-bis-(1- methylethenyl)cyclohexane	-	-	0.028	-	-	
56	1-Methylene-2-vinylcyclopentane	-	-	-	0.276	-	
57	1-Naphthalenol, 1,2,3,4,4a,7,8,8a- octahydro-1,6-dimethyl-4-(1- methylethyl)-, $[1R-(1\alpha,4\beta,4a\beta,8a\beta)]$ -	-	-	-	0.575	-	
58	1S -camphene	0.190	-	-	0.134	0.229	
59	1S-α-Pinene	1.906	0.236	6.349	-	2.436	
60	2,3-Dimethyl-cyclohexa-1,3-diene	0.309	-	-	-	-	
61	2,4,6-Octatriene, 2,6-dimethyl-	-	-	-	-	0.028	
62	2,4-Cycloheptadien-1-one, 2,6,6- trimethyl-	0.212	-	-	-	-	
63	2,4-Dimethylfuran	-	-	-	0.113	-	
65	2,6-Octadien-1-ol, 3,7-dimethyl-, (Z)-	-	-	0.047	-	0.093	
66	2-Acetylcyclopentanone	0.218	-	-	-	-	
67	2-Amino-4-methylpyrimidine	-	-	-	0.027	-	
()	2-Carene	0.318	-	0.056	-	-	
08	2-Cyclonexen-1-oi, 2-methyl-5-(1- methylethenyl)-, (s)	-	-	-	-	0.020	
69	2-Cyclonexen-1-01, 2-methyl-5-(1- methylethenyl)-, cis-	-	-	-	-	0.121	
70	2-cyclohexen-1-one 2-methyl-5-(1- methylethenyl)-(R)-	-	-	-	0.070	-	
71	2-Cyclopenten-1-one, 2-methyl-	-	-	-	-	0.116	
72	2-Oxabicyclo [2.2.2]octan-6-ol, 1,3,3- trimethyl-, acetate	-	-	-	0.031	-	
73	2-Pinen-10-ol	-	-	_	0.116	_	

		Mean composition % (w/w)					
No	Component	Sun -dried		Spouted <b>b</b>	oed dried	dried	
			45°C	55°C	65°C	75°C	
74	3,5-Dimethyl cyclohex-1-ene-4- carboxaldehyde	-	-	_	0.073	_	
75	3-carene	3.889	-	0.840	0.432	0.549	
76	3-Cyclohexen-1-ol, 4-methyl-1-(1- methylethyl)-, (R)-	0.222	-	-	0.436	-	
77	3-methylene-1 6-heptadiene	-	-	0.211	-	-	
78	3-Octyne, 7-methyl-	-	0.075	-	-	-	
79	3-p-Menthene	1.887	3.533	-	-	-	
80	4-ETHYNYL-5-DECENE;	-	-	-	-	-	
81	4-Hydroxy-4-(2,6-dimethyl cyclohex-3- enyl)butan-2-one	-	-	-	0.051	-	
82	4-hydroxy-4-(4 6-dimethyl cyclohex-3- enyl)butan-2-one	-	0.061	-	-	-	
83	5-Hexanal, 4-methylene	-	-	0.067	-	-	
84	5-Hexanal,4-methylene	-	-	0.058	-	-	
85	5-Hexenal, 4-methylene-	-	-	-	-	0.028	
86	5-methylene-1,3a,4,5,6,6a- hexahydropentalen-1-ol	-	-	0.204	-	-	
87	5-Octadecene, (E)-	-	-	0.033	-	-	
88	5-Oxatricyclo [8.2.0.0(4,6)-]dodecane, 4,12,12-trimethyl-9-methylene-, [1R- (1R*,4R*,6R*,10S*)]-;	0.347	-	-	-	-	
89	6,6-Dimethyl-2-methylenebicyclo [2.2.1]heptan-3-one	0.119	-	0.111	0.076	0.082	
90	6-Methyl-6,7 dihydro-9H-5-oxa-9- azabenzocyclohepten-8-one	-	-	-	0.050	-	
91	8a-Methyl-3,4,4a,5,8,8a-hexahydro- 1(2H)-naphthalenone	-	-	-	-	0.719	
92	Alloaromadendrene	0.416	0.069	-	-	_	
93	alpha caryophyllene	0.282	-	-	-	_	
94	Artemiseole	0.030	-	0.135	0.054	0.030	
95	Benzene, (fluoromethyl)-	-	-	0.054	-	-	
96	Benzene, 1-(1,5-dimethyl-4-hexenyl)-4- methyl-		-	0.157	0.106	_	
97	Benzenemethanol, $\alpha, \alpha, 4$ -trimethyl-	-	-	-	-	0.116	
98	Berbenone	_	-	-	0.063	_	

		Mean composition % (w/w)					
No	Component			Spouted bed dried			
	Component	Sun -dried		Γ		Γ	
			45°C	55°C	65°C	75°C	
99	bicyclo 3.1.0 hexane 4-methyl-1-(1-				• • • •		
100	methylethyl)- didehydro deriv	3.190	1.599	1.596	2.041	1.668	
100	bicyclo 4.4.0 dec-1-ene 2-isopropyi-5- methyl-9-methylene-	0 503	_	_	_	_	
101	Bicyclo [3,1,0]hex-2-ene, 4,4,6,6-	0.505	_	_	_	_	
	tetramethyl	-	-	0.047	-	-	
102	Bicyclo [3.1.1]hept-3-en-2-ol, 4,6,6-						
102	trimethyl-;	0.163	-	-	-	-	
103	Bicyclo $[3.1.1]$ nept-3-en-2-one, 4,6,6- trimethyl- $(1S)$ -	_	_	0.031	_	_	
104	Bicyclo [3.1.1]hept-3-ene, 2-		_	0.031	_	_	
	formylmethyl-4,6,6-trimethyl-	-	0.206	-	-	-	
	Bicyclo [3.1.1]heptan-3-ol, 6,6-						
105	dimethyl-2-methylene-, [1S-				0.157		
106	[10, 50, 50]- Bicyclo [3.2.0]heptap_3_ol 2_	-	-	-	0.157	-	
100	methylene-6,6-dimethyl-	-	-	0.067	-	-	
107	Bicyclo [3.3.0]oct-2-en-7-one, 6-						
	methyl-	-	-	0.051	-	-	
108	Bicyclo [4.4.0]dec-1-ene, 2-isopropyl-	0 177	0.400	1 458	0.000	0.821	
109	Bicyclo [6.1.0]nonane. 9-(1-	0.177	0.499	1.450	0.999	0.021	
	methylethylidene)-	-	0.148	-	-	-	
	Bicyclo [7.2.0]undec-4-ene, 4,11,11-						
110	trimethyl-8-methylene-, [1R-			0.520			
111	(1K <sup>+</sup> ,4Z,9S <sup>+</sup> )]-	-	-	0.320	-	-	
	Bornylene	-	-	0.280	0.253	0.444	
112	C. 11. 1/10.2.9 (				0.025		
113	Cadaia-1(10),3,8-triene	-	-	-	0.035	-	
115	Calamenene	-	-	-	-	0.401	
114	_						
115	camphene	0.337	0.356	0.468	0.293	-	
115	Carveol	-	-	0.238	0.162	0.082	
116							
117	Carvotanaceton	-	-	0.197	-	0.074	
11/	Carvotanacetone	_	0.813	-	-	-	
118							
	Caryophylla-4(12),8(13)-dien-5β-ol	-	-	-	0.269	-	
119	Carvonhyllono	0 305	11 301	0.008	12 574	14.79	
120		7.303	11.371	2.220	12.374	2	
	Caryophyllene oxide	3.926	11.531	6.558	5.482	4.173	
121	sis n Mantha 2.8 dist 1 sl	0.150		0.070			
122	cis -p-ivientna-2,8-dien-1-01	0.152	-	0.079	-	-	
	cis-Carveol	0.281	0.749	0.454	0.601	0.109	
123		0.150		0.070	0.017	0.001	
	c1s-Geran10l	0.120	-	0.078	0.065	0.031	

		Mean composition % (w/w)						
No	Component	Sun -dried		Spouted bed dried				
		Sun -uncu	45°C	55°C	65°C	75°C		
124	cis-Limonene oxide	-	-	0.165	-	-		
125	cis-Linalool oxide;	0.059	-	-	-	-		
126	cis-a-Bisabolene	-	-	0.047	-	-		
127	cis-β-Farnesene	-	-	0.074	0.024	-		
128	cis-β-Terpineol	-	-	0.261	0.087	0.085		
129	Crypton	-	0.128	-	-	-		
130	Cryptone	0.199	-	0.278	0.219	0.111		
131	Curcumene	-	0.117	-	-	-		
132	Cyclene	-	-	0.053	-	-		
133	Cyclofenchene	3.335	-	-	-	-		
134	2-methylene-, (1.alpha.,3.alpha.,5.alpha	-	0.346	-	-	-		
135	Cyclohexene, 1-methyl-4-(1- methylethylidene)-	-	-	-	-	0.071		
136	Cyclohexene, 4-ethenyl-4-methyl-3-(1- methylethenyl)-1-(1-methylethyl)-, (3R-trans)-	-	-	0.737	-	-		
137	Cyclohexene, 4-methyl-1-(1- methylethenyl)-	0.174	-	-	-	-		
138	Cycloisosativene	-	-	-	-	0.273		
139	Cyclopentane, 1,3-bis(methylene)-	0.084	-	-	-	-		
140	cyclopropane 1 1-dimethyl-2-(3- methyl-1 3-butadienyl)-	0.757	-	-	-	-		
141	Cyclosativene	0.490	-	0.538	0.294	-		
142	Cymenene	-	-	0.076	-	-		
143	D limonene	8.009	11.051	12.75 0	13.724	12.81 7		
144	d-Camphene	0.193	-	-	-	-		
145	D-Carvone	0.082	-	0.188	0.138	-		
146	Di-epi-a-cedrene	-	-	0.262	0.144	-		
147	dipropyl ester	0.087	-	-	-	-		
148	E Geraniol	-	-	-	-	-		

		Mean composition % (w/w)						
No	Component	Com duisd		Spouted bed drie				
		Sun -ariea	45°C	55°C	65°C	75°C		
149	E-15-Heptadecenal	-	-	-	0.035	-		
150	Elemene	-	_	_	-	0.498		
151	Fucaryone	0.130	_	0.063	_	_		
152	Germacrana D	0.120		0.000	0.033			
153	Globulol			0.048	0.055			
154		-	-	0.048	-	-		
155	Hexanoic anhydride	-	-	0.026	-	-		
156	Humulene epoxide 2	-	0.660	-	0.090	-		
157	Humulene epoxide II	-	-	0.083	-	-		
158	Isogeraniol	-	-	0.047	-	-		
159	Isolimonene	-	-	-	0.204	0.071		
157	Limonene	1.372	-	0.106	-	-		
100	Linalool	1.487	0.338	0.903	0.595	0.621		
161	Longipinene	-	-	-	-	0.177		
162	L-Pinocarveol	-	0.871	-	-	0.459		
163	L-α-Terpineol	0.121	-	-	-	-		
164	L-β-Pinene	2.358	5.715	4.765	5.245	-		
165	methoxy methyl chlorosilane	-	-	0.022	-	-		
166	Methoxy(methyl)chlorosilane	-	_	0.132	0.026	_		
167	Naphthalene, 1,2,3,5,6,8a-hexahydro- 4,7-dimethyl-1-(1-methylethyl)-, (1S- cis)-	_	_	0.452		_		
168	Naphthalene, 1,2,4a,5,6,8a-hexahydro-	0.052	0.262	01102				
169	Naphthalene, 1,2-dihydro-1,1,6- trimethyl-	0.033	-	0.129	0.031	-		
170	Neo-allo-ocimene	_	-	_	0.217	-		
171	Ocimene	_	_	_	0.024	_		
172	o Cumana	0.202			0.027			
173		0.292	-	-	-	-		
174	p-Cymen-7-oi p-Cymen-8-oi	0.222	-	-	0.053	-		
		0.278	-	-	-	-		

		Mean composition % (w/w)					
No	Component	Sun -dried	5	oed dried			
			45°C	55°C	65°C	75°C	
175	Phytol	-	-	0.031	0.032	-	
176	Pinocarveol	0.248	0.089	1.305	0.453	0.051	
177	Pinocarvone	-	-	0.066	0.057	-	
178	p-Mentha-6,8-dien-2-ol	-	0.299	-	-	-	
179	Sabinene	3.431	3.169	3.746	4.076	5.608	
180	Sabinyl acetate	-	0.241	-	-	-	
181	Spiro [2.4]heptane, 1,5-dimethyl-6- methylene-	-	0.112	-	-	-	
182	Spiro [5.6]dodecane	-	0.000	-	-	-	
183	S-Verbenone	0.232	-	0.277	-	-	
184	Sylvestrene	-	-	-	-	0.139	
185	Terephthalic acid, ethyl nonyl ester	-	-	-	0.036	-	
186	Terpinen-4-ol	-	0.217	-	-	1.246	
187	Terpinenol	-	-	-	0.268	-	
188	[6.3.2.0(2,5).0(1,8)]tridecan-9-ol, 4,4- dimethyl-	0.152	-	-	-	-	
189	trans-2,3-Epoxydecane	-	-	0.063	-	-	
190	trans-3-methylpent-3-ene-5-ol	0.063	-	-	-	-	
191	trans-Carveol	-	-	0.114	-	-	
192	trans-Pinocarveol	-	0.613	-	-	-	
193	trans-p-mentha-2 8-dienol	-	-	0.321	-	-	
194	trans-p-Mentha-2 8-dienol	-	0.099	-	-	-	
195	trans-p-Mentha-2,8-dienol	-	-	0.281	-	0.035	
196	1 ricyclo         [7.2.0.0(2,6)]undecan-5-ol,           2,6,10,10-tetramethyl- (isomer 3)	0.053	-	-	-	-	
197	Verbenol	0.184	0.075	0.271	0.150	-	
198	α- Terpinolen	0.134	-	0.073	0.052	-	
199	α-Cadinol	-	-	-	0.163	-	
200	α-Calacorene	0.075	-	0.180	0.063	0.102	

No Component Spouted bed drie	l			
	Spouted bed dried			
<b>Sun - dried</b> <b>45°C 55°C 65°C</b>	75°C			
201         α-Caryophyllene         2.810         0.868         1.093         1.117	1.280			
202         α-Copaene         2.106         8.191         4.371         2.382	7.770			
203 $\alpha$ -Cubebene 4.336 3.743 4.143 6.435	2.813			
204 α-Curcumene 0.136 -	_			
205 α-Elemene	0 139			
206 g-Earnesene 0.130	-			
207 <i>a</i> Gurienene 0.130				
208         -         -         0.009         -	-			
209 0.470 0.700 1.445 1.240	0.132			
a-Muurolene         0.470         0.790         1.445         1.340           210         0.424         0.404         0.404         0.404	1.740			
α-Phellandrene         0.134         0.184         -         -           211             -	-			
α-Pinene         -         6.854         0.127         2.391           212                     391               391	-			
α-Selinene         0.073         0.271         -         0.130           213	0.167			
α-Terpinene         0.296         -         -         -           214   <	-			
α-Terpineol         0.429         0.090         0.880         0.520	0.880			
β-Bisabolene $0.925$ $3.318$ $3.202$ $3.058$	3.733			
β-cis-Caryophyllene         0.781         - <td>_</td>	_			
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	0.048			
218 β-Elemene	0.489			
219 β-Farnesene 0.026 0.036	0.058			
220 β-Gurjenene 0.860	0.579			
221         β-Myrcene         0.919         0.620         0.969         0.618	0.892			
222 β-Phellandrene 0.444 4.479	-			
223 β-Phorone - 0.179 -	_			
224 β-Pinene 1.504 4.207 - 0.153	3.564			
225     β-Selinene     0.117     -     0.150     0.182	0.404			
226     β-Terpinen       -     -	4.746			

		Mean composition % (w/w)						
No	Component			Spouted <b>b</b>	oed dried			
	-	Sun -dried	45°C	55°C	65°C	75°C		
227	ß-Terpinene	_	_	_	4 472	_		
228	ß-Thuiene	_	_	_	-	0.027		
229	γ-Cadinene	0.107	-	-	0.159	-		
230	γ-Muurolene	0.520	-	0.313	0.281	0.468		
231	γ-Terpinene	0.387	-	0.025	0.060	0.123		
232	δ3-p-Menthene	4.827	-	4.102	-	-		
233	δ-Cadinene	1.192	0.882	0.628	1.166	0.895		
234	ε-Cyclogeraniolene	0.021	-	-	-	-		
235	τ-Cadinol	0.134	0.432	-	-	-		
	Trace compounds	5.079	0.583	5.048	2.407	1.911		
	Total	98.84	99.50	100.0	99.92	95.99		

				0	% perce	ntage of	f compo	nents		
Sample No	conditio	Mo	noterpene	è	Oxygen	ated mono	oterpene	sesquit erpene	Oxyge sesquit	enated terpene
	n	C10H16	C10H18	C <sub>10</sub> H	C <sub>10</sub> H <sub>14</sub> O	C <sub>10</sub> H <sub>16</sub> O	C10H18O	C15H24	C <sub>15</sub> H <sub>24</sub> O	C <sub>15</sub> H <sub>26</sub> O
1	Sun dried	41.79	-	-	1.83	6.67	6.62	30.89	5.60	-
2	Sun dried	50.56	-	-	0.13	2.30	4.79	32.69	4.56	-
3	Sun dried	29.56	7.55	-	2.67	0.64	10.71	24.36	8.17	-
4	Sun dried	64.39	7.08	-			1.48	25.15	1.25	0.54
5	45°C	29.09	14.13	-	0.60	4.19	1.69	37.90	5.64	-
6	45°C	66.27		-			1.95	27.78	0.74	0.45
7	45°C	13.33	0.00	-	3.90	12.18	0.00	24.73	39.03	0.00
8	45°C	50.76	-	-		0.82	0.47	39.16	6.09	1.28
9	55°C	24.65	-	0.30	0.38	9.93	5.01	27.51	9.59	-
10	55°C	40.49	-	-	1.26	6.63	2.74	33.38	4.91	-
11	55°C	48.59	-	-	0.19	2.50	3.37	38.84	5.07	0.19
12	55°C	25.73	16.41	-	0.57	6.07	3.05	32.53	7.00	-
13	65°C	49.96	-	-			1.63	34.69	11.13	-
14	65°C	43.82	-	-	0.49	4.70	4.06	37.30	3.74	-
15	65°C	49.64	-	-	0.27	4.64	2.87	32.73	4.18	2.30
16	65°C	46.56	-	-	2.24	1.99	1.73	36.82	4.32	0.65
17	75°C	50.26	-	-	-	0.80	1.71	42.80	2.58	-
18	75°C	42.75	-	-	-	4.23	3.54	39.20	2.59	-
19	75°C	55.81	-	-	-	-	1.30	41.40	1.38	-
20	75°C	27.89	-	-	0.46	2.68	8.68	46.94	10.15	-

# Table 2: Percentage of Constituents of essential oil extracted from black pepper dried at different drying conditions

Component	Concentration	Formula and structure	Molecular
	(% w/w)		weight
Caryophyllene	14.792	C <sub>15</sub> H <sub>24</sub>	204.3511
Caryophyllene	11.531	C <sub>15</sub> H <sub>24</sub> O	220.3505
oxide		H H	
α-Copaene	8.191	C <sub>15</sub> H <sub>24</sub>	204.3511
α-Cubebene	6.435	$C_{15}H_{24}$	204.3511

Table 3: Information of five most abundant sesquiterpene in black pepperessential oil under investigation [120]

Component	Percentage	Formula and structure	Molecular
			weight
D limonene	13.724		136.2340
(1R)- α -Pinene	6.982		136.2340
α-Pinene	6.854	C <sub>10</sub> H <sub>16</sub>	136.2340
L-β-Pinene	5.715	C <sub>10</sub> H <sub>16</sub>	136.2340
Sabinene	5.608	C <sub>10</sub> H <sub>16</sub>	136.2340
δ3-p-Menthene	4.827		138.2499

# Table 4: Information of six most abundant monoterpene in black pepper essential oil under investigation [120]

		A	NOVA Table				
			Sum of	df	Mean	F	Sig.
C10H16 * Dry_tempe rature	Between Groups	(Combined)	436.641	4	Square 109.160	.519	.723
	Within Group	s	3153.769	15	210.251		
	Total		3590.409	19			
C15H24 *	Between Groups	(Combined)	446.632	4	111.658	5.282	.007
Dry_tempe rature oxides *	Within Group	S	317.070	15	21.138		
	Total		763.702	19			
	Between Groups (Combined)		186.821	4	46.705	1.102	.394
Dry_tempe	Within Group	s	593.603	14	42.400		
rature	Total		780.424	18			

Table 5: ANOVA table for comparative study of sesquiterpenes, monoterpenes and oxygenated terpenes with drying temperature

Table 6: ANOVA table for Caryophyllene with drying temperature

ANOVA Table											
			Sum of Squares	df	Mean Square	F	Sig.				
Caryophylle	Between Groups	(Combi ned)	107.132	4	26.783	1.722	.201				
dry temp	Within G	roups	217.711	14	15.551						
J_J_J_	Total		324.842	18							

	ANOVA Table											
			Sum of Squares	df	Mean Square	F	Sig.					
oil_yield *	Between Groups	(Comb ined)	1.558	4	.390	3.148	.046					
Dry_conditio	Within Groups		1.856	15	.124							
115	Total		3.414	19								

## Table 7: ANOVA table for oil yield vs. drying temperature

### **APPENDIX F**

 Table 1: Operating conditions, heat transfer coefficient, important non-dimensional numbers of heat transfer process of black pepper drying in conventional spouted bed.

Dum	Operat	ing con	ditions	Properties of air			h <sub>p</sub>	Dimensionless numbers			
Kull	H (cm)	T <sub>gi</sub> (°C)	U (m/s)	ρ (kg/m <sup>3</sup> )	μ x 10 <sup>-5</sup> (Ns/m <sup>2</sup> )	k <sub>g</sub> (W/mK)	(W/m <sup>2°</sup> C)	Nu	Re	Gu	H/dp
0	14	45	2.37	1.1095	1.9307	0.0276	49.00	9.25	708.07	0.3111	26.90
1	14	55	2.37	1.0757	1.9763	0.0283	45.00	8.27	670.65	0.4273	26.90
2	14	65	2.37	1.0439	2.0212	0.0291	54.67	9.79	636.35	0.4846	26.90
3	14	75	2.37	1.0139	2.0655	0.0298	57.24	9.99	604.82	0.5667	26.90
4	14	65	1.60	1.0439	2.0212	0.0291	51.69	9.25	429.71	0.4846	26.90
5	14	65	1.97	1.0439	2.0212	0.0291	59.09	10.58	530.73	0.4846	26.90
6	7	65	2.37	1.0439	2.0212	0.0291	65.93	11.80	636.35	0.5000	13.45
7	18	65	2.37	1.0439	2.0212	0.0291	51.09	9.15	636.35	0.4877	34.59
8	20	65	2.37	1.0439	2.0212	0.0291	46.96	8.41	636.35	0.5538	38.43

#### Table 1. Continued

Run	H (cm)	T <sub>gi</sub> (°C)	U (m/s)	ρ (kg/m <sup>3</sup> )	μ x 10 <sup>-5</sup> (Ns/m <sup>2</sup> )	k <sub>g</sub> (W/mK)	h <sub>p</sub> (W/m <sup>2°</sup> C)	Nu	Re	Gu	H/dp
10	20	55	2.37	1.0139	2.0655	0.0298	47.16	8.67	670.65	0.4091	38.43
11	22	75	2.37	1.0757	1.9763	0.0283	45.64	7.97	604.82	0.4923	42.28
12	20	45	2.37	1.0139	2.0655	0.0298	40.80	7.70	708.07	0.3000	38.43
13	14	45	2.37	1.1095	1.9307	0.0276	51.09	9.64	708.07	0.3111	26.90
14	18	45	2.37	1.1095	1.9307	0.0276	53.82	10.16	708.07	0.3444	34.59
15	18	55	2.37	1.1095	1.9307	0.0276	43.50	7.99	670.65	0.5455	34.59
16	18	55	2.37	1.0757	1.9763	0.0283	53.68	9.86	670.65	0.4545	34.59
17	14	45	1.60	1.0757	1.9763	0.0283	47.09	8.89	478.14	0.3111	26.90
18	14	60	2.37	1.1095	1.9307	0.0276	57.50	10.43	653.13	0.5417	26.90
19	18	75	2.37	1.0595	1.9988	0.0287	56.29	9.83	604.82	0.6000	34.59
20	18	75	1.60	1.0139	2.0655	0.0298	53.51	9.34	408.42	0.5933	34.59
21	22	55	2.37	1.0139	2.0655	0.0298	45.28	8.32	670.65	0.4273	42.28
22	22	45	2.37	1.0757	1.9763	0.0283	40.86	7.71	708.07	0.3444	42.28

#### Table 1. Continued

Run	H (cm)	T <sub>gi</sub> (°C)	U (m/s)	ρ (kg/m <sup>3</sup> )	μ x 10 <sup>-5</sup> (Ns/m <sup>2</sup> )	k <sub>g</sub> (W/mK)	h <sub>p</sub> (W/m <sup>2°</sup> C)	Nu	Re	Gu	H/dp
23	22	60	1.60	1.1095	1.9307	0.0298	40.18	7.29	441.04	0.5500	42.28
24	14	35.3	2.37	1.0595	1.9988	0.0283	34.50	6.69	747.74	0.0935	26.90
25	24	75	2.37	1.1444	1.8857	0.0298	52.40	9.15	604.82	0.5840	46.12
26	24	65	2.37	1.0139	2.0655	0.0276	48.37	8.66	636.35	0.5200	46.12
27	22	65	1.60	1.0439	2.0212	0.0276	44.86	8.03	429.71	0.5200	42.28
28	24	65	2.57	1.0439	2.0212	0.0276	48.95	8.76	691.39	0.5200	46.12
29	24	55	2.57	1.0439	2.0212	0.0283	47.27	8.69	728.66	0.4327	46.12
30	24	55	2.37	1.0757	1.9763	0.0283	47.91	8.80	670.65	0.4327	46.12
31	7	65	1.60	1.0757	1.9763	0.0276	65.10	11.65	429.71	0.5200	13.45
32	7	75	2.37	1.0439	2.0212	0.0287	67.18	11.73	604.82	0.6000	13.45
33	7	45	2.06	1.0139	2.0655	0.0298	58.14	10.98	616.10	0.3067	13.45
34	7	55	2.33	1.1095	1.9307	0.0298	63.93	11.75	660.97	0.4327	13.45
35	7	55	1.72	1.0757	1.9763	0.0283	61.80	11.36	486.74	0.4364	13.45

Tabl	le 1.	Continu	ed

Run	H (cm)	T <sub>gi</sub> (°C)	U (m/s)	ρ (kg/m <sup>3</sup> )	μ x 10 <sup>-5</sup> (Ns/m <sup>2</sup> )	k <sub>g</sub> (W/mK)	h <sub>p</sub> (W/m <sup>2°</sup> C)	Nu	Re	Gu	H/dp
36	7	65	1.72	1.0757	1.9763	0.0291	63.14	11.31	461.85	0.5200	13.45
37	7	65	2.54	1.0439	2.0212	0.0291	67.60	12.10	682.27	0.5200	13.45
38	7	75	2.49	1.0439	2.0212	0.0298	69.47	12.13	635.37	0.5840	13.45
39	7	65	2.06	1.0139	2.0655	0.0291	67.75	12.13	553.69	0.5200	13.45
40	14	60	1.60	1.0439	2.0212	0.0287	51.49	9.34	441.04	0.4800	26.90
41	14	60	2.54	1.0595	1.9988	0.0287	54.99	9.97	700.26	0.4800	26.90
42	14	65	2.59	1.0595	2.0212	0.0291	56.80	10.17	696.04	0.4923	26.90
43	7	36	1.68	1.0439	1.8890	0.0269	49.64	9.61	529.79	0.1333	13.45
44	7	40	2.15	1.1418	1.9076	0.0272	57.13	10.93	659.78	0.2200	13.45
45	14	40	2.20	1.1272	1.9076	0.0272	45.60	8.73	675.54	0.2200	26.90
46	18	65	2.49	1.1272	2.0212	0.0291	57.65	10.32	668.49	0.5200	34.59
47	18	65	2.06	1.0439	2.0212	0.0291	48.91	8.76	553.69	0.5200	34.59

# **APPENDIX G**

### Table 1: Summary of model for heat transfer

Model Summary <sup>b</sup>							
Model	R	R Square	Adjusted R Std. Error of		Durbin-		
			Square	the Estimate	Watson		
1	.889 <sup>a</sup>	.791	.777	.0299600	1.981		
a. Predictors: (Constant), LogGu, LogHbydp, LogRe							
b. Dependent Variable: logNu							

#### Table 2: ANOVA Table for model for heat transfer

ANOVA <sup>a</sup>								
Model		Sum of	df	Mean	F	Sig.		
		Squares		Square				
	Regression	.150	3	.050	55.524	.000 <sup>b</sup>		
1	Residual	.039	44	.001				
	Total	.189	47					
a. Dependent Variable: logNu								
b. Predictors: (Constant), LogGu, LogHbydp, LogRe								

### Table 3: Coefficients for model for heat transfer

Coefficients <sup>a</sup>									
Model	Unstandardized Coefficients		Standardized Coefficients	t	Sig.	Collinearity Statistics			
	В	Std. Error	Beta			Tolerance	VIF		
(Constant)	1.081	.170		6.363	.000				
LogRe	.137	.064	.158	2.154	.037	.886	1.12		
LogHbydp	293	.024	881	-12.274	.000	.922	1.08 5		
LogGu	.180	.028	.463	6.329	.000	.886	1.12 9		
a. Dependent Variable: logNu									


Figure 1 : Model for heat transfer



Figure 2: Model for heat transfer

### **APPENDIX H**

#### 1. Terminal velocity of a particle

• Terminal falling velocity corresponding to a value of Re' lies between 0.2 to 500 [12]:

$$u_t = \frac{g{d_p}^2}{18\mu}(\rho_s - \rho_f)$$

• Terminal falling velocity corresponds to a value of Re' greater than about 500 [12]:

$$u_t^2 = \frac{3gd_p}{\rho_f}(\rho_s - \rho_f)$$

- $u_t terminal velocity (m/s)$
- $g-gravity \ accelaration \ (m/s^2)$
- d<sub>p</sub> particle diameter (m)
- $\mu-viscosity$  of medium (Ns/m²)
- $\rho_{\text{s}}$  density of solid particle (kg/m³)
- $\rho_f$  density of fluid (kg/m<sup>3</sup>)
- Re'- particle Reynolds number

#### 2. Calculation of dimensionless numbers [104]

#### 1 Reynolds (Re) number

$$\operatorname{Re} = \frac{d_{p}U\rho}{\mu}$$

- d<sub>p</sub> particle diameter (m)
- U- Superficial air velocity (m/s)

 $\rho-density$  of medium at dryng temperature (kg/m<sup>3</sup>)

 $\mu$  – viscosity of medium (Ns/m<sup>2</sup>)

#### 2 Nusselt (Nu) number

$$Nu = \frac{h_p d_p}{k_g}$$

 $h_p -$  heat transfer coefficient of air to particle (W/m<sup>2</sup>K)

d<sub>p</sub> – particle diameter (m)

 $k_g$  – conductivity of medium (W/mK)

#### 3 Gukhman Number

$$Gu = \frac{T_{gi} - T_{gi,w}}{T_{gi}}$$

 $T_{gi}$  –Air inlet temperature (K)

 $T_{gi,w}$  – gas inlet wet bulb temperature (K)

# **3.** Air thermo physical properties were calculated as function of the temperature [132]

• Kinematic viscosity of air (v)

$$\upsilon = -1.155 \times 10^{-14} \mathrm{T}^3 + 9.572 \times 10^{-11} \mathrm{T}^2 + 3.760 \times 10^{-8} \mathrm{T} - 3.448 \times 10^{-6}$$

- Thermal conductivity of air (k<sub>g</sub>)  $k_g = -1.5207 \times 10^{-11} \text{T}^3 + 4.8574 \times 10^{-8} \text{T}^2 + 1.0184 \times 10^{-4} \text{T} - 3.933 \times 10^{-4}$
- Humid volume is the volume of the unit mass of dry air and its associated water vapour at that temperature and pressure [12].

$$V_h = \frac{22.4}{29} \left(\frac{T}{273}\right) + \frac{22.4H}{18} \left(\frac{T}{273}\right)$$

#### $V_h = (0.00283 + 0.00456H)T$

• Humid heat is the heat required to raise the temperature of unit mass of dry air and its associated vapour one degree K at constant pressure [12]

$$C_s = C_{air} + HC_{vapour}$$
$$C_s = 1.005 + 1.88 H$$

- Average particle diameter (d<sub>p</sub>) of black pepper were measured and was taken as 0.005204 m or 5.204 mm.
- True density of raw black pepper was measured using a densometer and the true density is 747.696 kg/m3. True density value was not used in any calculation of this study.

Where

 $k_{\rm g}$  - Thermal conductivity of air (W/mK

 $V_h$  - humid volume (m3/kg) of dry air)

T- Temperature (K)

H - Humidity (kg of moisture/kg of dry air)

Cs - humid heat (kJ/kg K)

C<sub>vapour</sub> - heat capacity of water vapour

 $\upsilon$ - kinematic viscosity (m<sup>2</sup>/s)

## List of Publications

No	Description	Category
1	G. K. Jayatunga and B. M. W. P. K. Amarasinghe, Drying	Index
	Kinetics, Quality and Moisture Diffusivity of Spouted Bed	Journal
	Dried Sri Lankan Black Pepper. Journal of Food Engineering.,	
	vol. 263, pp. 38–45, Dec. 2019.	SCI/Scopus
2	Jayatunga G.K, Amarasinghe B.M.W.P.K., Drying Kinetics of	
	Black Pepper Dried in A Spouted Bed Dryer with or without	Refereed
	Draft Tubes. International Journal of Manufacturing &	Iournal
	Industrial Engineering – IJMIE, Volume 1: Issue 2, pp. 6–10,	Journal
	Publication Date : 25 June 2014	
3	G.K. Jayatunga, S.M.N.D Martino and B.M.W. P.K.	
	Amarasinghe, Thin Layer Drying Models for Drying of Black	
	Pepper in Spouted Bed Dryer with Internal Devices.	
	Transaction of IESL annual session. October 2016. Sri Lanka	
4	Jayatunga G.K, Amarasinghe B.M.W.P.K., Mathematical	
	Modeling of Drying Kinetics of Black Pepper in a Spouted Bed	Refereed
	Dryer with and without Non Porous Draft Tubes. International	Conference
	Conference on "Trends in Multidisciplinary Business and	
	Economic Research (TMBER 25-26 March 2015). © 2015	
_	Global Illuminators, Bangkok, Thailand	
5	Jayatunga G.K, Amarasinghe B.M.W.P.K., Effect of	
	Configuration of Draft Tubes on Drying Kinetics of Black	
	Pepper Dried in a Spouled Bed Dryer. Transaction of IESL	
6	annual session. October 2014. Sri Lanka	
6	Jayatunga G.K And Amarasinghe B.M.W.P.K, Effect of drying	
	TELIM Symposium 2018. Sei Lonko	
7	In UM Symposium 2018, Sh Lanka	<b>F</b> ( 1 1
/	of maisture diffusivity of black papper drying in conventional	Extended
	on moisture unfusivity of black pepper drying in conventional	Abstract -
8	Spouled ded, 11 OW Symposium 2017, SH Lanka	conterence
0	Internal Device on Drving Rate of Black Denner In Spouted	proceeding
	Red ITLIM Symposium 2016 Sri Lanka	
0	Drying Kinetics of Black Denner in Spouted Red Dryon	
7	Proceeding of the HETC Symposium July 2014 Colombo Sri	
	I anka	